

PURDUE UNIVERSITY and PURDUE RESEARCH FOUNDATION Lafayette, Indiana

Report No. TM-69-1

I

I

E

:

:

「キャンマション」となっていたので、

The second states and

A FUNDAMENTAL INVESTIGATION OF THE PHENOMENA THAT CHARACTERIZE LIQUID-FILM COOLING

> by R. A. Gater M. R. L'Ecuyer

Technical Memorandum

Contract Nonr 1100(21) and NASA Grant NsG 592

Jet Propulsion Center Purdue University January 1969

# RECEDING PAGE BLANK BOT FRIED.

#### ACKNOWLEDGMENTS

The investigation reported herein was initiated under the sponsorship of the Office of Naval Research, Power Branch, under Contract Nonr 1100(21). The intermediate phases of the investigation were sponsored by the National Aeronautics and Space Administration under Grant NsG 592. The authors wish to express their appreciation for the support provided by the aforementioned agencies.

The authors are also indebted to Dr. B. A. Reese, Director, Jet Propulsion Center, Purdue University, for his unwavering support of the research program and for his many helpful suggestions.

The help of Messrs. W. Timmons, G. Hurst, C. Merkel, T. Miller and T. Virgin in the fabrication of the research apparatus is gratefully acknowledged.

The authors also wish to thank Mrs. Joyce Gater for her typing and repeated retyping of the original manuscript, and Mrs. Virginia Hawks for her expert typing and invaluable assistance in the preparation of the final manuscript.

Finally, the aucnors wish to express their appreciation to Dr. R. B. Gaither, Chairman, Department of Mechanical Engineering, University of Florida, for his cooperation and assistance in the publication of this report.

iii

## TABLE OF CONTENTS

																										Page
LIST	r of	TAE	3LE	ES .	•	•	•	•	•		•	•	٠	•	•	•	•	•		,	•	•	•	•	•	ix
LIST	r of	FIG	JUF	RES	•	•	•	•	•		•	•	•	•	•	•	•	•		•	•	•	•	•	•	x
ABST	FRACT	• •	•	•	•	•	•	•	•		•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	xvi
1.	INTE	וססו	ICI	101	ι.	•	•	•	•	•	•	•	•	•	•	•	•		4	,	•	•	•	•	•	1
	1.1	Ge	ene	eral	D	is	cu	s s	ic	n		•	•	•	•	•	•	•	4	•	•	•	•	•	•	1
	1.2	A	St	tate	emei	nt	0	f	tł	) e	Ρ	ro	ь1	ler	n	•	•	•		•	•	•	•	•	•	2
		1. 1.	.2.	. 1 . 2	The The	e	Li Ga	qu s-	id Va	-    p	Fi or	1 m C	n ( ;oc		olo ed	ed R€	Re eg	eg io	ic n	o n	•	•	•	•	•	4 6
	1.3	A	Ge	ener	ral	R	ev	ie	W	0	f	th	e	L.	it	era	iti	ur	е		•	•	•	•	•	8
	1.4	Tł	ıe	Sco	pe	0	f	th	е	I	n v	es	t	iga	at	ior	۱			•	•	•	•	•	•	15
2.	THE	EXF	PER	RIME	ENT	4L	I	NV	ES	5T	IG	AT	.1(	) N	•	•	•	•		•	•	•	•	•	•	18
	2.1	Tł	ıe	Bas	sic	E	хр	er	in	iei	nt	a 1	ļ	\p	pa	rat	tus	s		•	•	•	•	•	•	18
	2.2	Tł	ıe	Exp	ber	i m	en	ta	1	M	o d	e l	•	•	•	•	•	•		•	•	•	•	•	•	25
	2.3	Tł	ıe	Exp	ber	i m	en	ta	1	Pa	a r	an	ne t	tei	rs	•	•	٠		•	•	•	•	•	•	30
	2.4	Tł	ie Cł	Pho nara	oto acto	gr: er	ap is	hi ti	C C S	Da	at	a	f(	or •	ti •	he •	F •	i]	m	s ·	ur •	•fa	ice	•	•	33
		2.	. 4 .	1	The	e 9 n	In +	f1	ue	en E 4	ce		)f	tl	he	Li	i qı	u i	d	P	ar	an is	iet	er	s	37
		2.	. 4 .	2	The	e Pai	In rai	f1 me	ue te	en er:	ce s	or	)f   1	tl th	he e	Ga Fil	as Im	S S	ti	re rf	an ac	1 :e		C 3	•	57
		2.	, 4 .	. 3	( Sui	C h nm	ar ar	ac y	te	er	is •	ti	•	s •	•	•	•	•		•	•	•	•	•	•	43 48
	2.5	Tł	ne Ve	Dat eloc	ta :it:	fo y	r Ga	Ma s	s s S 1	s tr	Tr ea	ar m	1s1 •	fei	r ·	for •	•	a	C	on •	st •	ar •	it .	•	•	49

-

-

1

į.

;•

E

ſ

ĺ

I

.

Page

and the second state of th

가지 가지 가지 않는 것은 아이들에 있는 것을 수가 하는 것 같다.

ā,

	2.	5.	1	T	he a	] nd	in:	f1 th	u e	en L	c i	e qu	o1 ic	F	t! V	he is	e i C	Su os	r	fa ty	C	e on	Te	en th	s e	io	n			
	2.	5.	2	T	D he P	at 1 ar	a n ai	f fl me	oi ue te	en er	Ma C ( S	as e o	s of n	T F t	ri ti ho	ar he e	ns P D	fe Ga at	r s a	S f	t	re r	ar Ma	n as	s	•	•	•		57
	2.	5.	3	Si	T um	ra ma	n: r	s f y	eı	•	•	•	•	•	•	•		•	•	•		•	•	•		•	•	•		61 63
2.6	Τh	e Ac	Dai cel	ta lei	f ra	or ti	n i	Ma g	s s Ga	s a s	Tı	ra St	ns re	s f e a	e: m	r.	f	or •	•	a n	!	•	•	•		•	•	•		64
2.7	Τh	e Te	Dat	ta era	f at	or ur	e	th	e	M	a : •	xi •	mı	ı m		Li.	q	ui •	d •	F •	i	] m •	•	•		•	•	•		67
CORRE	ELA	ΤI	ON	01	F	T۲	E	E	XF	ΡĘ	R	ΙM	E	١T	AI	_	D.	AT	A	•		•	•	•		•	•	•		71
3.1	Co	rr a	ela Cor	nt' Ist	io ta	n nt	0	f Ve	ti 10	ne oc	l i 1	Da ty	ta (	a Ga	f ( s	or S	St	Ma re	s	s m	T	ra •	ns •	sf	e	r •	fo	r •		72
	3.	1.	1 2	G e T I	en he	er E	a v	1 a 1	D <sup>2</sup> uz	is at	cu id	us on	si	io of	n	tt	ne	R	a	te	•	of		Si	m	n]	e	•		72
	3.	1.	3	T	M	a s C	S	Trr	ra el	in Ia	s† t	f e i o	r n	0	f	t	:h	e	Ė	nt	r	a i	nr	ne	n	t	•	•		79
	۲	1	4	т	P	ar o	ar	ne	te i	er - a	) ו	e o	i	, , ,	•	• Fi		• = n	•	•	0	f	•	•		•	•	•		90
		<u> </u>	т -		C	or	r	e 1	at	ti	01	n	fo	br	1	9 0	)	•	•	с •	Ű	•	•	•		•	•	•		<b>9</b> 8
	3. 3.	1. 1.	5 6	TI TI	he P he t	C ar C he	o ar or	rr ne np Me	e te ai	ia er ri su	t I S( r(	io r on ed	n c F	0 , of Ra	f ・ t	t th es	:h ie	e · P of	R r	ou ed Ma	g i s	hn ・ ct s	es • • Ti	ss 1 ra	a n	nd sf	er	•	1	02 08
3.2	Th	e fo	Moc r N	lii 1a:	Fi SS	ca T	t	io an	ri S 1	0 fe	f r	t	he o	e A	Ba	a s c c	;i )u	c nt	C	or fo	r	el a	a i n	ti	0	n			-	
	2	81 2	ווס ו	ura M/	ir.	у ~1	r	1 I 2 D	ш Ч	د ۸	0 ( n :	וכ 1 פ	e	1	L (	er	ıg	τn		•		•	•	•		•	•	•	ב ו	10
	3.	2.	2	A	n A	En	ים: 1 י	ir vt	i i i		1 1 1	A	pr Pr	sı or su	5 02 1	xi t	m	at	i	• • •	)	to	• 1	th	e	•	•	•	1	24
3.3	Co	rr an	ela Ac	at cce	io el	n er	0 'a	f ti	tl	ne J	ا Ga	Da a s	ta	a St	f( r	or ea	m	Ma	s •	د ا	T	ra •	.n: •	sf	e	r •	fo	or •	1	29
	3.	3.	1	TI	he M	E	V	al T	ua rz	at an	10	on fo	r	)f f	0	th r	ie a	S	i A	mp c c		e ï e	Ri	at at	e	0 na	f			
	3.	3.	2	Ţ	G	a s M	10	Fi	01	v a	n	d	Ai	na	1	y s	- i 1	s_	•	•		•	•	•		•	•	•	1 1	29 33
	5.	5.	5	11	ne t	( h€	01 	mp Me	a 1 a 2	r 1 5 u	5 <u>(</u> r(	g n e d		of Ra	t	t n e s	ie i	۲ of	r	ed Ma	1	CT S	e Ti	ı ra	a n	nd sf	er	•	1	39

.

•

ŀ

• • •

-

// \_\_\_\_\_

E

**Weight** 

I

3.

	3.4	Correlation of the Experimental Data for the Maximum Liquid Temperature	142
		3.4.1 A Method for Predicting the Maximum	142
		3.4.2 Comparison of the Experimental and the Predicted Results for the	- • •
		Maximum Film Temperature	145
4.	CORRE REF	ELATION OF THE DATA OF THE PRIMARY FERENCES	150
	4.1	Introduction	150
	4.2	An Approximate Method for Evaluating the Surface Tension	157
	4.3	The Data Due to Kinney, Abramson, and Sloop	160
		•	
		4.3.1 Introductory Remarks	160
		4.3.2 The Basic Apparatus	160
		4.3.4 Comparison of the Present Theory With	102
		the Experimental Data	163
		4.3.5 A Comment on the Presentation of Data	
		for Liquid-Film Cooling in Terms of	
		Dimensionless Parameters	1/1
	4.4	The Data Due to Emmons	176
		4.4.1 Introductory Remarks	176
		4.4.2 The Basic Apparatus	177
		4.4.3 The Assumptions	177
		4.4.4 Comparison of the Present Ineory with	102
		A A F An Additional Commont on the Persanch	102
		Due to Emmons	184
	4.5	The Data Due to Hermann, Leitinger, and	
•			189
		4.5.1 Introductory Remarks	101
		4.5.2 INC BASIC Apparatus	103 TAT
		4.5.4 Comparison of the Present Theory With	1 J J
		the Experimental Data	194
		4.5.5 An Illustrative Example for	
		$T^0 = 4000 \text{ p and } n^0 = 500 \text{ nota}$	106
		g = 4000 κατα ρ = 500 ρετα	T 30

I

I

I

E

I

Page

-

5. CONCL	LUSIONS	202
BIBLIOGRA	APHY	206
NOTATION		212
ADDENDTY	A AN ANALYSTS FOR THE GAS-VAPOR COOLED	
REGION	· · · · · · · · · · · · · · · · · · ·	217
A.1	Introduction	217
A.2	The Case Where the Wall is Adiabatic	218
	A 2 ] The Flow Model	218
	A.2.2 The Energy Balance	220
	A.2.3 The Evaluation of $m^{"}$	225
	A.2.4 A Special Case	229
	A.2.5 Experimental Data for the Adiabatic-	
	Wall Temperature From Reference	
	(19)	232
<b>Δ</b> 3	The Case Where the Wall is Cooled	
	Externally	234
A.4	An Observation on the Data of Emmons for	997
	Gas-Vapor Cooling	231
APPENDIX	BDESCRIPTION OF APPARATUS	241
B.1	General Description	241
B.2	The Gas Generator	247
6 9	The Test Costion and Test Diato	260
D.J	The rest Section and rest Flate	200
<b>B.4</b>	The Separator	266
APPENDIX	CEXPERIMENTAL PROCEDURE	270
C.1	Pre-Run Preparations	270
C.2	In-Run Procedure	272
C.3	Post-Run Procedure	273
APPENDIX	DPHYSICAL PROPERTIES OF FILM COOLANTS	274
D.1	Subject Investigation	274
D.2	Investigation Due to Emmons	279

.

.

The second secon

T

[

E

Į

Page

		Page
APPENDIX LIQUID	EA SAMPLE CALCULATION FOR THE MAXIMUM TEMPERATURE	283
APPENDIX	FDATA REDUCTION	286
F.1	General Data Reduction	286
F.2	Gas Stream Parameters	288
F.3	Mass Transfer Data	291
APPENDIX	GEXPERIMENTAL DATA	292

.

And a second second

2° -----

3

.

yiii

# LIST OF TABLES

lable		Page
1.	Nominal Flow ParametersConstant Velocity Gas Stream	32
2.	Nominal Flow ParametersAccelerating Gas Stream	34
3.	Comparison of the Nominal Flow Parameters for the Primary References to Those for the Present InvestigationConstant Velocity Gas Stream	152
4 .	Comparison of the Nominal Flow Parameters for the Primary Reference (24) to Those for the Present Investigation Accelerating Gas Stream	154
5.	Estimation of the Surface Tension for the Analysis of the Data Due to Emmons	181
Appendix Table		
B.1.	Components of the Gas Generator	250
B.2.	Output Characteristics of the Gas Generator	251
G.1.	Average Values of the Pertinent Flow Parameters During the Experimental Tests With a Constant Velocity Gas Stream	294
G.2.	Average Values of the Pertinent Flow Parameters During the Experimental Tests With an Accelerating Gas Stream	297

# LIST OF FIGURES

Figure		Page
1.	Illustration of the Liquid-Film Cooling Process	3
2.	Illustration of Gas-Vapor Cooling	7
3.	Comparison of Liquid-Film Cooling and Simple- Mass-Transfer Cooling	12
4.	Schematic Diagram of Experimental Apparatus .	19
5.	Schematic Diagram of Experimental Test Section	21
6.	The Experimental Model	26
7.	Influence of m <sub>l</sub> on Film Surface CharacteristicsLow Gas Stream Velocity	38
8.	Influence of m <sub>l</sub> on Film Surface CharacteristicsHigh Gas Stream Velocity .	39
9.	Influence of Surface Tension on Film Surface Characteristics	41
10.	Influence of G, u <sub>g</sub> , and M <sub>o</sub> on Film Surface CharacteristicsG Constant	44
11.	Influence of G, u <sub>g</sub> , and M <sub>o</sub> on Film Surface Characteristicsu <sub>g</sub> Constant	46
12.	Influence of G, $u_g$ , and $M_o$ on Film Surface Characteristics $M_o$ Constant	47
13.	Rate of Liquid Withdrawal as a Function of Vent Flow Rate for Separator	53

;-•

н н **н** 

7

[

l

х

Figure		Page
14.	Example of Experimental Data for Mass Transfer. Constant u <sub>g</sub> . Methanol Film Coolant	55
15.	Influence of Surface Tension and Liquid Viscosity on Experimental Data for Mass Transfer	58
16.	Influence of Gas Stream Parameters on Experimental Data for Mass Transfer	62
17.	Experimental Data for Mass Transfer. Accelerating Gas Stream. Methanol and Water Film Coolant	65
18.	Comparison of Maximum Liquid Temperature and Boiling Temperature	70
19.	Schematic Illustration of Separate Contributions to Net Rate of Mass Transfer	. 78
20.	Control Volume for Interfacial Energy Balance	81
21.	Entrainment Parameter $e_0$ as a Function of the Entrainment Group $X_e$	93
22.	Roughness Parameter r as a Function of the Roughness Group X <sub>r</sub>	105
23.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer. Constant Velocity Hot Gas Stream. Methanol Film Coolant	110
24.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer. Constant Velocity Hot Gas Stream. Butanol Film Coolant	111
25.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer. Constant Velocity Hot Gas Stream. Water Film Coolant	112
26.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer. Constant Velocity Hot Gas Stream. RP-1 Film Coolance	113

ſ

ſ

ſ

ľ

1

ľ.

ſ

The C

C .....

[

Ø

Figure		Page
21.	Comparison of Present Theory With Data for Mass Transfer. Constant Velocity Cold Gas Stream	115
28.	Model for Analysis of Arbitrary Film- Cooled Length	119
29.	Matching of Empirical Equation to Analytical Result. $m_s^{"} = cx^{-1/5}$	125
30.	Matching of Empirical Equation to Analytical Result. m <sub>s</sub> = cx	127
31.	Pertinent Gas Stream Parameters as a Function of the Distance From the Point of Liquid Injection. Accelerating Gas Stream	134
32.	Model for Analysis of Experimental Data for Mass Transfer. Accelerating Gas Stream	135
33.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer. Accelerating Gas Stream	140
34.	Computed Results for Maximum Liquid Temperature as a Function of T <sub>g</sub> and p for an Air-Water System	146
35.	Comparison of Present Theory With Data for Maximum Liquid Temperature	148
36.	Schematic Diagram of Apparatus of Kinney, Abramson, and Sloop	161
37.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer Due to Kinney, Abramson, and Sloop. 4-inch Diameter Smooth-Surface Tube. Water Film Coolant	165
38.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer Dui to Kinney, Abramson, and Sloop. 2-inch Diameter Smooth-Surface Tube - Water Film	
	Coolant	166

. . .

xii

Į

ĺ

Ŀ

Ē

Satility

La antes a

I

I

コーマー いまい時の いろいしてい

--2

· · · · · · ·

xiii

Figure		Page
39.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer Due to Kinney, Abramson, and Sloop. 4-inch Diameter Rough-Surface Tube. Water Film Coolant	169
40.	Nusselt-Reynolds Number Plot of Data Due to Kinney, Abramson, and Sloop. Smooth- Surface Tubes	172
41.	Nusselt-Reynolds Number Plot of Data Due to Kinney, Abramson, and Sloop. Rough- Surface Tube	173
4?.	Schematic Diagram of Apparatus of Emmons	178
43.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer Due to Emmons. Various Gas Stream Reynolds Numbers	183
44.	Comparison of Simple Theory and Present Theory With Data for Mass Transfer Due to Emmons. Various Film Coolants	185
45.	Schematic Diagram of Apparatus of Hermann, Leitinger, and Melnik	192
46.	Coolant Flow Rate as a Function of the Film- Cooled Length and the Surface Tension	197
47.	Local Mass Flow Rate as a Function of the Axial Position in the Nozzle and the Coolant Flow Rate	200
Appendi: Figure	х	
A.1.	Flow Model for Analysis of Gas-Vapor Cooling. Adiabatic Wall	221
A.2.	Comparison of the Present Theory With the Data for Adiabatic-Wall Temperature Due to Kinney, Abramson, and Sloop	233
B.1.	Photograph of Experimental Apparatus. Configuration for Tests With Constant Velocity Gas Stream	242

7.4 S

1

2

I

F

I

I

I

Append Figure	ix	Page
B.2.	Photograph of Experimental Apparatus. Configuration for Tests With	940
		243
B.3.	Photograph of Control Room	240
B.4.	Schematic Diagram of Gas Generator	249
B.5.	Schematic Flow Diagram	256
B.6.	Sectional Drawing of Test Section	261
B.7.	Cross Sectional Drawing of Test Section	262
B.8.	Photographs of Test Section	265
B.9.	Sectional Drawing of Separator	267
B.10.	Photograph of Separator	268
D.1.	Surface Tension as a Function of Liquid Temperature for Liquids of Present Investigation	275
D.2.	Liquid Viscosity as a Function of Liquid Temperature for Liquids of Present Investigation	276
D.3.	Heat of Vaporization as a Function of Liquid Temperature for Liquids of Present Investigation	277
D.4.	Vapor Pressure as a Function of Liquid Temperature for Liquids of Present Investigation	278
D.5.	Surface Tension as a Function of Liquid Temperature for the Liquids Employed by Emmons	280
D.6.	Heat of Vaporization as a Function of Liquid Temperature for the Liquids Employed by Emmons	281
D.7.	Vapor Pressure as a function of Liquid Temperature for the Liquids Employed by Emmons	282

二三百

.

xiv

Appendi Figure	ix Pag	e
G.1.	Data for Mass Transfer. T <sub>g</sub> =600F, p=75 psia. Water Film Coolant 29	8
G.2.	Data for Mass Transfer. T <sub>g</sub> =600F, p=150 psia. Water Film Coolant 29	9
G.3.	Data for Mass Transfer. T <sub>g</sub> =400F, p=75 psia. Methanol Film Coolant	0
G.4.	Data for Mass Transfer. T <sub>g</sub> =600F, p=75 psia. Methanol Film Coolant	)1
G.5.	Data for Mass Transfer. T <sub>g</sub> =400F, p=150 psia. Methanol Film Coolant	)2
G.6.	Data for Mass Transfer. T <sub>g</sub> =400F, p=75 psia. Butanol Film Coolant	)3
G.7.	Data for Mass Transfer. T <sub>g</sub> =400F, p=150 psia. Butanol Film Coolant	)4
G.8.	Data for Mass Transfer. Tg=600F, p=75 psia. Butanol Film Coolant	)5
G.9.	Data for Mass Transfer. $T_g$ =400F, p=75 psia. RP-1 Film Coolant	96
G.10.	Data for Mass Transfer. T <sub>g</sub> =400F, p=150 psia. RP-1 Film Coolant	)7
G.11.	Data for Mass Transfer. $T_g=600F$ , p=75 psia. RP-1 Film Coolant	8
<b>G.12.</b>	Data for Mass Transfer. Cold Flow, p=75 psia. Methanol Film Coolant	)9
G.13.	Data for Mass Transfer. Cold Flow, p=75 psia. Water and Butanol Film Coolant 31	.0
G.14.	Data for Mass Transfer. Accelerating Gas Stream. Water and Methanol Film Coolant 31	1

ころうち ないない ちょうちょう ちょうちょう ちょうしょう しょうしょう ちょうしょう しょうしょう

**Margaret** 

一日 あんたち あっとない なたい ちょういい

1

хv

#### ABSTRACT

Gater, Roger Allen. Ph.D., Purdue University, January 1969. A Fundamental Investigation of the Phenomena That Characterize Liquid-Film Cooling. Major Professor: Dr. Mel R. L'Ecuyer.

An experimental investigation was conducted to determine the net rate of mass transfer (evaporation plus entrainment) from a thin liquid film to a proximate gas stream. Three different flow situations were investigated: (a) constant velocity, cold gas flow; (b) constant velocity, heated gas flow; and (c) strongly accelerated, heated gas flow. The experiments were conducted in a horizontal test section by passing an air stream over a liquid film formed on the lower surface. The liquid film was 2-in. wide and was terminated at a length of 10 in. by a knife-edge capture slot. The flow parameters for the experiments with a constant velocity gas stream were: gas stream temperatures of 40F, 400F, and 600F; gas stream static pressures of 75 and 150 psia; gas stream velocities of 40 to 400 fps; and liquid coolants of methanol, butanol, water and RP-1 (a hydrocalbon fuel). The flow parameters for the experiments with an accelerating gas stream were: stagnation temperature of 400F; stagnation pressures of 50, 75, 100, 125 and 150 psia;

「あい、ころい、ころい、「夏夜夜

variation in gas stream velocity of 200 to 660 fps over the liquid film; and liquid coolants of methanol and water.

The measured rate of mass transfer with a heated gas stream was found to be as much as 10 times greater than the rate of mass transfer predicted from the simple (or classical) theory for mass transfer. In most cases, the mass transfer due to the entrainment of bulk liquid into the gas stream was the most significant contribution to the overall mass transfer. In those cases where evaporative mass transfer was important, the "effective" roughness of the surface of the liquid film was found to increase the rate of evaporation by as much as 80 percent over that predicted by the simple theory.

All of the data for mass transfer (approximately 600 datum points) was correlated in terms of relatively simple dimensional parameters, with one parameter characterizing the phenomenon of entrainment and a second parameter characterizing the phenomenon of film surface roughness. Of particular significance is the fact that a fifteenfold variation in the liquid viscosity was not found to have any noticeable effect on the rate of entrainment.

こまちまで、行成しま、肥を見てた

An analysis was made to extend the results of the present investigation to the more general case of an arbitrary film cooled length. The utility of the analysis was investigated by analyzing the data reported in three primary references. The agreement between the reported data and the

#### xvii

results predicted from the present theory was good, with the agreement being to within the accuracy of the experimental data in most cases.

Photographs were taken of the surface of the liquid film so that the gross characteristics of the film surface could be studied as a function of the pertinent flow variables for the liquid and gas phases.

Measurements were made of the maximum temperature that a liquid film obtains for the subject flow situation. A theory was developed for the prediction of the maximum liquid temperature that was in good agreement with the experimental data.

An analysis was made of the heat transfer characteristics for the gas-vapor cooled region immediately downstream of the liquid film. The analysis allowed for compressible gas flow in the presence of a favorable pressure gradient. Consideration of the limited experimental data that are reported in the literature for the subject problem indicated that the analysis was promising.

#### 1. INTRODUCTION

#### 1.1 General Discussion

Liquid-film cooling refers to the introduction of a thin continuous liquid film onto a given surface for the purpose of protecting that wetted surface from thermal damage by a proximate hot gas stream. This technique provides a means of protecting, for example, the internal surfaces of a rocket motor from the injurious effects of the hot propellant gases. The motivation for developing liquid-film cooling technology for rocket motors is due to the use of higher energy chemical propellants, the trend toward higher combustion chamber pressures, and the development of nuclear rocket engines. These newer developments are characterized by a large increase in the heat flux from the working fluid to the walls of the combustion chamber and exhaust nozzle, such that the regenerative cooling of those walls becomes marginal (1, 2, 3)\*.

In the design of a liquid-film cooled system for a rocket motor, the two principal problems to be considered are (a) that of determining the rate of coolant consumption that will be required to wet, and thus thermally protect, a \*Numbers in parentheses refer to REFERENCES.

and the second second

į.

specified internal surface area, and (b) that of determining the effect that the cooling process has on the performance of the system. In the subject investigation, attention was concentrated on the first problem only; specifically, with the investigation of the phenomena that characterize the convective exchange of heat and mass between a hot gas stream and a thin liquid film. Although the principal motivation for this research was the interest in liquid-film cooling, the investigation is of a much broader interest because it relates to the basic problem of the interaction of a high velocity gas stream and a thin liquid film, a situation that is frequently encountered in engineering.

#### 1.2 <u>A Statement of the Problem</u>

Figure 1 illustrates the process termed liquid-film cooling. Presented in the figure is a hot gas stream flowing along a surface A-B. A liquid coolant is introduced at station  $x = x_1$  and is swept downstream over the surface by the gas, thereby forming a continuous protective liquid film. As a result of the evaporation of liquid in the film, together with the possible entrainment of liquid droplets into the gas stream, the liquid film will terminate a. some point, say,  $x = x_2$ . The wall region wetted by the liquid will herei... be termed the <u>liquid-film cooled region</u>.

Since the gas and vapor mixture which is convected downstream is relatively cool, the wall region immediately downstream of the liquid film will be insulated to some degree



hot gas stream --



2 :

3

.

7 i

\*\*\* \*

ľ

{

F.

F

[]

**[**: |.

(` |

[ \

ſ

A

from the hot gas stream. It is convenient to define the <u>gas-vapor cooled region</u> as that wall region over which the temperature of an adiabatic surface would increase from that for the liquid film at  $x = x_2$  to the recovery temperature for the mainstream at, say,  $x = x_3$ . The extent of the gas-vapor cooled region so defined is thus a measure of the effectiveness of the gas-vapor cooling process.

The basic analytical and design problem can be stated as that of determining, for given liquid and gas flow parameters, (1) the rate of coolant injection required to establish a desired wetted area, and (2) the degree of the insulating effect of the gas-vapor layer downstream of the liquid film. The remaining portion of this discussion is directed toward a more detailed consideration of each of these two problems.

### 1.2.1 The Liquid-Film Cooled Region

<u>The basic problem</u> for the liquid-film cooled region is that of being able to predict the wetted surface area as a function of the rate of liquid injection and as a function of the pertinent flow parameters for both the liquid and gas phases. The solution to this problem is difficult to obtain because of the complex phenomena that characterize the flow of a high velocity gas over a liquid film.

The subject problem is outlined more specifically as follows: For the case where a high velocity, hot gas stream is flowing over a thin (of the order of a few thousandths of

an inch thick) liquid film, the mass transfer from that film is made up of two fundamental contributions: (a) the mass transfer that results from the transfer of energy to the liquid film and the resultant evaporation of the liquid; and (b) the mass transfer that results from a physical breakdown of the interfacial structure such that liquid droplets are entrained into the gas stream.

The evaporative contribution to the total mass transfer rate (item (a) above) can be divided into yet two more distinct contributions: (a-1) that contribution which would result if the liquid film surface presented a hydrodynamically smooth, zero velocity boundary to the gas stream; and (a-2) the remaining portion of the evaporative mass transfer which results because the surface of the liquid film is moving at a finite velocity and does not, in general, present a "smooth" surface to the gas stream. (The surface of the liquid film generally exhibits a wavy or pebbled appearance which should serve to intensify the turbulent transfer of heat and mass near the gas-liquid interface.)

14.14

ł

The second second

li

Contribution (a-1) is the same as that prevailing in the case of simple mass transfer\*. That part of the total evaporative contribution is relatively amenable to analysis. Reference (4), for example, presents a recent review of the extensive analytical and experimental work performed on that problem of simple mass transfer. Although there still

\*The process termed herein "simple-mass-transfer cooling" is equivalent to the process that is frequently termed "transpiration cooling" in the published literature.

remain some problem areas in the thorough treatment of simplemass-transfer cooling, the effect of mass addition to a boundary layer upon the rate of heat transfer to the surface has been adequately described for the purposes of the present investigation.

No systematic attempt has previously been made to experimentally determine the individual contribution of each of the three aforementioned mass transfer mechanisms (a-1), (a-2), and (b) to the overall mass transfer rate. Although the fundamental problem is that of being able to predict the total mass transfer rate for a particular flow situation, it is evident that in order to develop a satisfactory correlation for that total mass transfer rate, a basic understanding has to be developed of the nature of each of the three contributions discussed above.

#### 1.2.2 The Gas-Vapor Cooled Region

Figure 2 serves to introduce the analytical problem for the gas-vapor cooled region. Illustrated in Fig. 2 are the pertinent wall temperature profiles immediately downstream of the terminus of the liquid film. The upper solid curve presented represents the wall temperature profile for the case where the wall is adiabatic\*. The lower solid curve represents a characteristic  $\mu$ , of the for the case where the wall is cooled externally. As indicated in the figure, it is convenient to base the heat transfer coefficient for gas-vapor

\*The term "adiabatic" means that the rate of heat transfer through the wall to the surroundings is negligible.







Fig. 2 mustration of Gas-Vapor Couling

cooling, denoted by  $h_{gv}$ , on the temperature difference  $(T_{aw}-T_w)$ . When that is done, the rate of heat transfer to an externally cooled wall is given by

$$q_{w} = h_{qv} \left( T_{aw} - T_{w} \right)$$
 (1-1)

It follows, therefore, that the basic problem is to determine, for given flow parameters, (1) the adiabatic wall temperature profile, and (2) the local heat transfer coefficient  $h_{av}$ .

1.3 <u>A General Review of the Literature</u>

Research that has been performed that is pertinent to the subject problem can be conveniently grouped into two principal areas:

(a) research concerned primarily with the investigation of the detailed nature of the flow of a gas over a thin liquid film; and

(b) research concerned with an actual application of liquidfilm cooling.

The first area of research includes the investigation of such fundamental phenomena as interfacial structure and film instability\* (5, 6, 7, 8, 9), entrainment (10, 11, 12), weiting characteristics (13, 14, 15), and pressure crop in

\*Film instability is defined as the occurrence of large scale disturbances on the surface of the liquid film. These disturbances are often characterized as roll waves. annular, two-phase flow (16, 17). In general, those investigations were characterized by ambient pressure and temperature, essentially zero heat transfer rates, and relatively low interfacial shear forces. Such conditions are significantly different from those that characterize the typical application of liquid-film cooling. Furthermore, few data exist for the case where the liquid phase was not water. Although the experimental advantages of using water as the liquid phase are obvious, the fact remains that few other liquids have physical properties (particularly surface tension) that are similar to those of water. Considerable doubt thus exists regarding the application of results that were obtained with water as the liquid phase to other liquids. A detailed review of the literature in this first area of research is not presented here because of the question as to the relevance of the results--particularly in a quantitative sense--to the present investigation. The interested reader is referred to the excellent reviews presented in references (5, 10, 18). However, several specific references are made to those experimental results throughout this report when that reference gives some insight into the interpretation of the experimental results of the present investigation.

The research of the second area (namely, applications of liquid-film cooling) is characterized by relatively high rates of heat and mass transfer between the gas stream and

the liquid film (19, 20, 21, 22, 23, 24). The objectives of those research programs were basically (a) to establish the feasibility of liquid-film cooling the thrust chamber and/or nozzle of a rocket motor; and (b) to determine the relationship between the rate of coolant consumption and the resultant wetted surface area as a function of the gas stream Reynolds number, the gas stream pressure and temperature, the properties of the liquid coolant, etc. Attempts to develop empirical correlations for the experimental data have met with only limited success. The apparent reason for the lack of success has been the failure to incorporate into those correlations the influence of the interfacial phenomena\* on the rate of mass transfer from the liquid film to the gas stream (25, 26). Moreover, theoretical analyses developed for liquid-film cooling (e.g., 20, 21) have either lacked agreement with the experimental data or have resulted in correlations that were not justified physically; the previous analyses for liquid-film cooling have been critically reviewed in Ref. (26). The failure of the theoretical analyses is apparently due to the fact that they have assumed a direct correspondence between the problem of liquidfilm cooling and that of simple-mass-transfer cooling. Analytical models which had resulted in satistactory analyses for the case of simple-mass transfer were also assumed to describe the heat and mass transfer characteristics for the

\*That is, the effect of entrainment and of film surface roughness.

1

case of liquid-film cooling. Figure 3 aids in an explanation of the motivation for such an approach. Illustrated in the figure is a control surface "S" placed at the surface of an evaporating liquid film and at the surface of a smooth porous wall through which mass is being transferred into the gas stream. Assume that for each case the flow parameters, the boundary conditions, and the rate of mass transfer across the control surface into the gas stream are the same. It is then reasonable to argue that the following procedure can be employed to analyze the problem of liquid-film cooling. Thus

(a) The simple theory for mass transfer is employed to calculate the rate at which energy is transferred across the control surface "S" to the liquid film. The simple theory suggests that the rate of heat transfer to the liquid film decreases as the rate of mass transfer from the liquid film into the gas stream increases. That phenomenon is sometimes referred to as thermal "blockage".

(b) The rate of mass transfer from the liquid film into the gas stream is determined by writing an appropriate energy balance across the gas-liquid interface\*. This step is coupled to step (a) through the "blockage" phenomenon so that there is only one value for both the rate of heat transfer to the ity id film and the rate of mass transfer into the gas phase that is predicted for any particular

\*See Subsection 3.1.2 for a discussion on the interfacial energy balance.





situation. The following paragraph discusses how the experimental data of the primary references agree with the foregoing argument.

Emmons (20) concluded from his experimental investigation that the analogy between simple-mass-transfer cooling and liquid-film cooling was valid. Emmons suggested that the rate of heat transfer to an evaporating liquid film was always less than the rate of heat transfer to a dry wall, all other conditions being the same. He attributed this, of course, to thermal blockage. However, that conclusion drawn by Emmons is contradicted by the experimental data obtained by other investigators\*. Kinney, Abramson, and Sloop (19) have reported experimental data showing that the rate of heat transfer to a liquid film can be substantially greater (by a factor of 3 or more) than the rate of heat transfer to a dry wall\*\*. Luikov (27) obtained similar results from well controlled experiments in which extreme care was taken to ensure identical gas flow conditions and surface temperatures for (a) dry surface and (b) liquid surface test sections. The coefficient of heat transfer to the liquid film was consistently 20 to 60 percent greater than the coefficient of heat transfer to the dry wall. Hermann, Leitinger and Melnik (24) considered the problem of film-

4 4 L 6 V.

1

ł

.

i

1

 <sup>\*</sup> Evidence is presented in Section 4.4 of this report that the data of Emmons do not, in fact, support the analogy between simple-mass-transfer and liquid-film cooling.
 \*\*The data of Ref. (19) are analyzed in Section 4.3.

cooling a convergent-divergent nozzle with water\*. They found that the amount of coolant required to completely wet the convergent portion of the nozzle was roughly 2 times that predicted by the theory for simple mass transfer. They also estimated that the amount of coolant required for the divergent portion of the nozzle was 3.3 times that predicted by the simple theory\*\*.

The foregoing discussion raises a serious question regarding the adequacy of a simple-mass-transfer model for the analysis of liquid-film cooling. The difficulty of obtaining an adequate empirical correlation or developing a satisfactory theoretical analysis for liquid-film cooling on the basis of the previously reported data for the rate of mass transfer from a thin liquid film to a high velocity gas stream is due primarily to:

 (a) the limited amount of data that has been obtained for conditions similar to those encountered in an application of liquid-film cooling;

(b) the contradictory trends exhibited by the data of the primary references;

(c) the possibility of appreciable experimental error due to the severity of the flow parameters investigated in many cases; and

\* The data of Ref. (24) are analyzed in Section 4.5.
\*\*The discussion of Section 4.5 (in particular, that of Subsection 4.5.5) suggests that the estimates by Hermann, et al. (24) of the mass transfer rates for both the subsonic and supersonic portion of a nozzle cannot be generally applied.

ţ

(d) the scarcity of data for film coolants other than water.

The present research program was initiated because of the incomplete understanding of the phenomena that characterize liquid-film cooling, and because of the limited information that could be obtained from the existing literature.

#### 1.4 The Scope of the Investigation

The subject investigation had the primary objective of experimentally determining the net rate of mass transfer from a thin liquid film to a high velocity gas stream under conditions of relatively high pressure, temperature, and interfacial shear forces with significant variations in the physical properties of the liquid phase. Two cases were investigated: (a) the case where the velocity of the gas stream was essentially constant over the length of the film-cooled region, and (b) the case where the gas stream was strongly accelerated over the wetted surface. The latter case was investigated because of the interest in the application of liquid-film cooling to cool the exhaust nozzles of high-energy propulsion devices.

F

E

E

[i

(

A limited number of experiments were conducted for Case (a) in which the gas stream was not heated. These experiments served to (1) isolate the mass transfer that was due to entrainment from the mass transfer that was due to the evaporation of the liquid in the film, and to (2) significantly extend the range of values for the pertinent flow parameters that was investigated in the present

investigation.

A correlation was developed for the mass transfer data\*, isolating the contributions to the total mass transfer rate that were due to both (a) the simple mass transfer, and (b) the interfacial phenomena. The data correlation gives an explicit expression for the net rate of mass transfer from a thin liquid film to a high velocity gas stream in terms of the pertinent variables for both the gas and the liquid phases.

The maximum temperature that the liquid film obtains when it is introduced onto the surface of an adiabatic wall was measured for each experimental test that was conducted. A simple theory was developed for the prediction of the maximum film temperature that was in good agreement with the experimental data. The theory can be employed to estimate the maximum liquid-film temperature for the general case where the film temperature is not a known parameter.

A number of still photographs were obtained in the investigation for the purpose of making some qualitative observations about how the pertinent liquid and gas stream parameters influence the character of the interfacial structure. Those observations were utilized in the development of the aforementioned correlation for the experimental mass transfer data.

\*Approximately 600 datum points are reported herein for the rate of mass transfer from a thin liquid film to a high velocity gas stream.

16

1.84

The utility of the design formula (i.e., the correlation for mass transfer) that was developed for liquid-film cooling was investigated by attempting to correlate the published data of the primary references (19, 20, 24). In many cases, the values of the pertinent flow parameters (e.g., gas stream temperature, pressure, velocity and filmcooled length) investigated by those references differed substantially from those of the subject investigation. Analysis of those data, therefore, gave considerable insight into the validity of the design formula proposed herein for liquid-film cooling.

A subordinate objective of this investigation was to present an analysis for the gas-vapor cooled region immediately downstream of the terminus of the liquid film. The analysis presented is essentially the same as that previously reported by Gater, L'Ecuyer, and Warner (26). Since that original presentation, however, the analysis has been further evaluated and additional comments regarding its usefulness can be made.

Í.

#### 2. THE EXPERIMENTAL INVESTIGATION

2.1 The Basic Experimental Apparatus\*

Figure 4 illustrates schematically the experimental apparatus that was employed in the investigation. The basic components of the apparatus were a hot gas generator and a two-dimensional mixing section, an adiabatic approach section, a film-cooled test section, and a variable area exhaust nozzle.

The hot gas generator employed methyl alcohol and air as the reactants. Because so much diluent air was added to the combustion gases before they entered the experimental tunnel, the properties of the gas stream were assumed to be the same as those for dry air.

Prior to entering the test section, the gas stream passed through an assortment of fine mesh screens in the mixing section, thereby appreciably reducing the turbulence level and the non-homogenities of the flow. The gas then passed through a 36-inch long adiabatic approach section so that the remaining irregularities in the gas stream had a chance to dissipate and the velocity boundary layer had an

\*The experimental apparatus is discussed in greater detail in Appendix B.

18

traffer 1

R


I

I

I

Ĩ



្ន ដុំ ភិ

).9

opportunity to develop before the liquid film was introduced. (The thickness of the velocity boundary layer at the point of liquid injection was on the order of 0.2 to 0.4 inches.) The inside surface of the bottom wall of the approach section was surface ground so that a hydrodynamically smooth boundary was presented to the developing boundary layer.

Figure 5 shows schematically the test section employed in the investigation. The nominal internal dimensions of the test section are 2 in. by 5.5. in. cross section by 18 in. long. Mounted in each side wall of the test section were 6 circular Pyrex windows of 1 in. O.D. A single oblong Pyrex window was mounted in the top wall of the test section. These windows permitted visual and photographic observation of the liquid film. The photographic technique that was employed is illustrated in Fig. 5. The test plate to which the liquid film was confined was recessed into the bottom wall of the test section. The liquid coolant was introduced onto the test plate through an 1/8-in. thick piece of "regimesh" that was 1/2 in. long and 2 in. wide. This injector design provided a uniform distribution of coolant and also kept the normal component of velocity for the liquid at the injector face to something less than one ft per sec. The liquid film was confined to the central portion of the test plate by means of a trough that was machined into the plate. The trough was 2 in. wide and 0.030 in. deep, and it extended from the point of liquid injection to a point

20

- 2007 and - 4

\* algeorith

10.0

is



I

4

. .



At that point the liquid film was 10 inches downstream. mechanically terminated by means of a knife-edge capture slot. The projected area for the capture of the liquid film was 2.25 in. wide\* by 0.060 in. high\*\*. In preliminary experimentation with the effectiveness of the capture slot technique, it was found that it was necessary to allow some of the gas stream to bleed through the slot with the liquid. Attempts to withdraw only the liquid film resulted in spillage of the liquid film over the capture slot. The significance of this bleed gas flow is discussed in Section 2.5. In addition, it was visually observed that a small amount of liquid frequently spilled around the outside corners of the capture slot. This was particularly true if the rate of liquid injection was very high. To eliminate that spillage, the streamlines of the gas flow were made to converge slightly in the immediate neighborhood of the capture slot. Then, because the flow behavior of the liquid film was found to be strongly affected by the flow pattern of the gas stream, the liquid film was made to converge slightly toward the center of the capture slot thereby preventing spillage of liquid around the corners of the slot. For most of the experimental tests in which the gas stream velocity was essentially constant over the length of the wetted test

\* 0.25 inch wider than the liquid film \*\*approximately an order of magnitude greater than the maximum thickness that was investigated for the liquid film

plate, the convergence of the streamlines for the gas flow was accomplished by replacing the last circular Pyrex window in each side wall of the test section with an 1/2-in. thick by 1-in. O.D. stainless steel plug to which was welded an 1/8-in. thick by 1-in. wide by 1-1/2-in. long stainless steel plate. The degree to which these plates deflected the gas stream was regulated by rotating the plates, thereby permitting the projected area of each of the plates to be varied from 1/8 in. by 1-1/2 in. to 1 in. by 1-1/2 in. These plates were normally adjusted so that the liquid film would converge from a width of 2 inches at a point approximately 1/2 inch upstream of the leading edge of the capture slot to a width of 1-1/2 inches at the capture slot. The behavior of the liquid film for the remainder of the tests that employed a gas stream of constant velocity and for all of those tests where the gas flow was accelerated over the test plate was controlled by mounting two pieces of 1/32-in. thick stainless steel sheet directly on the test plate on each side of the capture slot. These sheets were formed so that they protruded 1/4 in. into the gas stream and caused the gas flow (and, therefore, the liquid film) to converge toward the center of the capture slot in the manner as previously described. Both of the aforementioned control devices were equally effective in the control of the behavior of the liquid film in the immediate neighborhood of the capture slot.

The gas and liquid mixture that was withdrawn from the test plate was passed through a cyclone separator. The gas phase was vented off and the liquid phase was collected in a beaker. Thus, by measuring the rate of liquid injection and the rate of liquid withdrawal, the net rate of mass transfer from the liquid film to the gas stream was determined. Also determined during the course of each mass transfer measurement were the temperatures for the liquid at the point of injection and at the point of withdrawal. The liquid temperatures were measured with thermocouples and they were employed to accurately specify the energy balance for the liquid film and to determine the values for the pertinent physical properties for the liquid phase.

For those experiments where the gas stream was accelerated over the length of the wetted test plate, the acceleration of the gas stream was accomplished by inserting an aluminum wedge into the top of the test section (see the dashed line in Fig. 5). Five static pressure taps were installed in the aluminum wedge so that the static pressure (and, therefore, the gas stream Mach number, velocity, and static termperature) could be determined as a function of the coordinate x. In those experiments the nozzle was removed from the test section and the flow was choked at the minimum area formed by the wedge.

f

In the subject investigation, the experimental apparatus was designed so that extraneous heat transfer and fluid

dynamic effects could be safely neglected, rather than trying to correct for them analytically. The liquid film was injected so as to wet only the center 2 inches of the 5.5-in. wide lower surface of the test section, thereby permitting the influence of the complex flow phenomena for the corners of the rectangular duct on the rate of convective heat transfer to the liquid film to be neglected. In establishing an energy balance for the liquid film, it is difficult to determine the amount of energy transferred through the wetted test plate to the liquid film in a lateral direction. Therefore, in order to substantially reduce the lateral conduction of heat through the test plate to the liquid film, two insulating grooves (0.060 in. wide) were cut into the underside of the test plate. The grooves extended to within 0.035 in. of the surface that was wetted by the hot gas, and were placed just outside the region that was wetted by the liquid film. The insulation grooves, extending from the point of liquid injection to the point of liquid withdrawal, increased the resistance of the plate to lateral heat conduction by roughly a factor of 50.

#### 2.2 The Experimental Model

Figure 6 illustrates schematically the experimental model for the subject investigation. The figure presents a high velocity, hot gas stream flowing over a horizontally mounted flat plate. The film coolant was injected at a



Gas Stream ----



÷

100-Carlos

the transfer of the second

3

Ì

l

]

ļ

]

1

2

rate  $m_1$  (1b/sec-ft)\* at  $x_1 = 40$  inches and was withdrawn at a rate  $m_2$  (1b/sec-ft) at  $x_2 = 50$  inches. The film-cooled length, denoted by  $l_0^{**}$ , was equal to 10 inches throughout the investigation.

The gas stream flow conditions that were investigated were such that the velocity boundary layer developing on the flat plate could be treated as being fully turbulent from the leading edge of the plate. The wall surface upstream of the point of liquid injection was essentially adiabatic so  $x_{+} = x_{1} = 40$  inches was that a thermal entrance lengti established. The thickness of the velocity boundary layer at the point of liquid injection was on the order of 0.2 in. to 0.4 in. Due to the fact that a velocity boundary layer developed on each wall of the experimental tunnel, a correction had to be made in the calculation of the gas stream velocity to account for displacement effects. The calculations showed that the gas stream velocity generally increased by approximately 5 percent in going from  $x = \hat{v}$  to  $x = x_{+}$ . For those investigations in which the gas stream was accelerated, the acceleration began at a point 6 inches upstream of the point of liquid injection and continued to a

\* The units (lb/sec-ft) follow from dividing the rate of liquid injection (lb/sec) by the width of the injector (ft).

<sup>\*\*</sup>In general, the film-cooled length will be denoted by
"1". The term 1 refers to the special case of the
present investigation where the film-cooled length was
fixed at 10 inches.

point 2 inches downstream of the end of the film cooled plate.

As indicated in Section 2.1, the liquid film was terminated by means of a knife-edge capture slot. Thus, unlike previous investigations of liquid-film cooling, the filmcooled length for the present investigation was a fixed parameter. That characteristic of the experimental model allowed the influence of the rate of liquid injection on the cooling process to be investigated independently of the other flow parameters. Moreover, it enabled the wetted surface area to be accurately determined so that the experimental data could be properly evaluated. In previous experiments (e.g., Refs. 20-24), the wetted surface area was determined by injecting a film of known width and then estimating the average film cooled length by means of thermocouples that vere embedded into the test section wall. However, in such experiments it has been found that it is very difficult to obtain a uniform termination of the liquid film (19, 20, 28). That is, when the film becomes progressively thinner as a result of the transfer of mass from the film to the gas stream, it reaches a point where surface tension effects and/or flow non-uniformities in the gas stream cause the liquid film to break down locally. It is, therefore, generally very difficult to properly interpret such external wall temperature measurements.

An experimental technique that was introduced into the subject investigation was that of preheating the liquid

R

before injection to the maximum temperature that the film assumes when it is in equilibrium with the gas stream. (When the wetted surface is diabatic, that maximum film temperature is essentially the wet-bulb temperature for the liquid.) The advanatages of this experimental technique are essentially twofold: First, because the physical properties for the liquid phase are rather strong functions of the liquid temperature, it is much easier to accurately specify the values for the physical properties of the liquid if the liquid film is isothermal. The second advantage to preheating the liquid before injection is that the energy balance at the gas-liquid interface can then be accurately specified. In previous experimental investigations only an "average" energy balance for the entire film could be prescribed because the liquid was introduced at a relatively low temperature, thereby making it impossible to ascertain what portion of the local heat transfer resulted in an increase in the sensible enthalpy of the liquid and what portion resulted in the evaporation of the liquid. To determine the local rate of heat transfer to the liquid film, however, it is necessary that the local energy balance be prescribed more accurately. The problem of specifying that energy balance is simplified considerably by injecting the liquid at the maximum temperature it will assume when it is in thermodynamic equilibrium with the hot gas stream.

Due to limitations in the heating capacity of the heat

exchanger that was employed, the liquid could not always be preheated completely to the maximum film temperature. But the liquid was always injected at a temperature substantially greater than the ambient temperature so that the aforementioned advantages to the preheating of the liquid coolant werc largely realized.

### 2.3 The Experimental Parameters

In order to minimize the experimental error and to obtain experimental data for mass transfer of sufficient accuracy to elucidate the phenomena and the parameters of importance, some sacrifice had to be made in the severity of the flow parameters that were investigated, especially the gas stream temperature. A number of datum points exist from previous investigations for the case of relatively high gas stream temperatures (to 4100R), but the accuracy of those datum points has necessarily suffered because of the measurement problems involved. Moreover, such severe gas stream temperatures place a physical limitation on the number of experimental datum points that can be obtained with a reasonable degree of effort. For those reasons, therefore, since both accuracy and quantity were badly needed for experimental data on liquid-film cooling, the gas stream temperature for the subject research program was limited to 600F. However, this temperature still resulted in an appreciable rate of evaporation of the liquids investigated because of the relatively high gas stream Reynolds numbers

) (

that were employed.

As mentioned previously, experiments were conducted for both the case where the gas stream velocity was essentially constant over the length of the film-cooled test section and for the case where the gas stream was accelerated over the length of the wetted test section. Moreover, a limited number of experiments were conducted for the first case for which the gas stream was not heated. Those experimental tests are termed herein the cold-flow experiments.

The liquids employed in the investigation were water, methanol, butanol, and RP-1 (a hydrocarbon fuel that is essentially kerosene). These liquids resulted in a rather wide variation in the pertinent physical properties\*. For example, the liquid viscosity was varied by roughly a factor of 15:1, the surface tension by 3:1, and the heat of vaporization by 4:1. The significant variation in the physical properties for the liquid phase enabled an accurate determination of the influence that each property exerted on the heat and mass transfer process.

Table 1 summarizes the nominal flow parameters for those experiments in which the velocity for the gas stream was essentially constant over the length of the wetted test plate. The table also serves to introduce some pertinent

<sup>\*</sup>Appendix D presents the heat of vaporization, the liquid viscosity, the surface tension, and the vapor pressure as a function of the liquid temperature for the liquids that were investigated.

T	a	b	1	е	1

1

U

Nominal\* Flow Parameters--Constant Velocity Gas Stream T<sub>a</sub>, Gas Stream Static Temperature Hot Flow 400 F, 600 F Cold Flow 40 F p, Gas Stream Static Pressure 75 psia, 150 psia u, Gas Stream Velocity Hot Flow 50-400 fps Cold Flow 40-200 fps  $10^{6}$ -10<sup>7</sup> per ft Re, Gas Stream Reynolds Number 1, Film Cooled Length 10 inches Liquid Film Coolants Water Methanol **Butanol** RP-1 (Hydrocarbon Fuel)

\*See Table G.1 of Appendix G for a Tabulation of the Specific Flow Parameters for Each Experimental Test With a Constant Velocity Gas Stream. notation that is employed throughout the remainder of this report.

Table 2 summarizes the nominal values of the pertinent flow parameters for the experimental tests in which the gas flow was accelerated over the wetted test plate.

#### 2.4 Photographic Data for the Film Surface Characteristics

A number of photographs of the surface characteristics of the liquid film were obtained for a significant range of values of the pertinent liquid and gas flow parameters. A 35 mm camera with a close-copy lens, together with a 3microsecond strobe unit for the light source, was employed to photograph the surface of the liquid film. The camera was focused on the surface of the liquid film through the Pyrex window that was mounted in the top wall of the test section (refer to Fig. 5). The strobe unit was focused on the surface of the liquid film through the side windows in the test section. All of the photographs presented herein were taken at a point 6 inches downstream of the point of liquid injection.

The surface of a thin liquid film over which flows a high velocity gas stream is generally characterized by a large number of small scale disturbances (or waves) which are essentially three dimensional in character. These small scale disturbances give the surface of the liquid film a "pebbled" or "cross-hatched" appearance. If large scale disturbances (instability waves) develop, they travel over and a state of the state of the

#### Table 2

Nominal\* Flow Parameters--Accelerating Gas Stream  $T^0_{\sigma}$ , Gas Stream Stagnation Temperature 400 F p<sup>0</sup>, Gas Stream Stagnation Pressure 50, 75, 100, 125, 150 psia u<sub>q</sub>, Gas Stream Velocity\*\* 200-660 fps p, Gas Stream Static Pressure\*\* 50 - 40 psia 75 - 60 psia 100 - 82 psia 125 - 105 psia 150 - 123 psia M, Gas Stream Mach Number\*\* 0.14-0.51 1, Film Cooled Length 10 inches Liquid Film Coolants Water Methanol

\* See Table G.2 of Appendix G for a Tabulation of the Specific Flow Parameters for Each Experimental Test With an Accelerating Gas Flow.

\*\*The Numbers Indicate the Variation in the Indicated Parameter Over the Length of the Film Cooled Plate.

this pebbled substructure at a velocity which is greater than the velocity of the smaller scale disturbances (10). A number of investigators (e.g., 19, 29-31) have proposed a criterion for the onset of these large scale disturbance waves on the surface of the liquid film. They have proposed that if the Reynolds number for the liquid film exceeds a certain critical value, then the large scale disturbance waves will develop on the surface of the liquid film. Furthermore, it has been suggested (e.g., 9, 20, 21, 22, 29, 30) that appreciable entrainment of liquid from a liquid film into a high velocity gas stream should occur only if the liquid-film Reynolds number exceeds that critical value. First it should be commented that the photographic data of the present investigation did not allow the suggested criterion for film instability to be evaluated. It should also be noted, however, that the data for mass transfer (discussed later in Section 2.5) did not confirm the existence of a critical liquid-film Reynolds number with reference to the phenomenon of entrainment. In fact, it was observed that the liquid viscosity  $\mu_1$  (which enters into the definition of the liquidfilm Reynolds number) had no measurable influence on the rale of mass transfer from a thin liquid film to a high velocity gas stream. Indeed, it is significant that the experimental data for mass transfer could be readily correlated without making any direct reference to the phenomenon of film instability.

The remainder of this section discusses how the pertinent flow parameters for the liquid and gas phases influence the gross structural characteristics of the surface of the liquid film. A limited number of photographs of the surface of the liquid film are presented to illustrate several important points. It should be noted, however, that the observations that are made with reference to the influence of the pertinent flow parameters on the film surface characteristics are the result of ... analysis of nearly 200 such photographs, and not just the analysis of those photographs that are presented herein. These observations will be pertinent to the correlation that is developed in Section 3.1 for the expe imental data on mass transfer.

Subsection 2.4.1, which follows, discusses how the surface characteristics of a liquid film are influenced by the parameters for the liquid phase; namely, the rate of liquid injection  $m_{\perp}$ , the liquid viscosity  $\mu_{\parallel}$ , and the surface tension  $\sigma$ . Subsection 2.4.2 discusses how the surface characteristics are dependent on the gas stream velocity  $u_{\rm g}$ , the gas stream mass velocity G, and the momentum parameter  $M_{\rm o}$ , where the parameters G and M<sub>o</sub> are defined by

 $G = \rho_g u_g (lb/ft^2 - sec)$  (2-1)

R

and

$$M_{o} = \rho_{g} u_{g}^{2} (1b_{f}/ft^{2}) \qquad (2-2)$$

respectively.

# 2.4.1 The Influence of the Liquid Parameters on the Film Surface Characteristics

The purpose of the present section is to discuss how the following parameters influence the characteristics of the liquid film surface: (a) the rate of liquid injection  $m_1$ , (b) the liquid surface tension  $\sigma$ , and (c) the liquid viscosity  $\mu_1$ .

Rate of Liquid Injection m<sub>1</sub>. Figures 7 and 8 present photographs of a 1-inch-square surface element of the liquid film. The pertinent flow parameters are listed in the figures for each photograph. The photographs shown in Fig. 7 were obtained for a case where the gas stream velocity had the relatively low value of 35 ft/sec. Figure 7 (a) illustrates that for the low liquid flow rate of 0.0144 lb/sec-ft, the film surface is characterized by a uniform distribution of essentially two dimensional waves. Figure 7 (b) demonstrates that as the liquid flow rate is increased to 0.0822 lb/sec-ft, with the other parameters remaining constant, the interfacial structure becomes less ordered. However, that particular change in the character of the liquid film surface with increasing rate of liquid injection was observed only

flow direction ----

38



Ĩ

I

Í

I

I

I

$$T_g = 42$$
 F  
p = 82 psia  
 $u_g = 35$  fps  
G = 15.5 lb/ft<sup>2</sup>-sec  
 $M_o = 17$  lb<sub>f</sub>/ft<sup>2</sup>

Methanol Film Coolant

(a) m<sub>1</sub>= 0.0144 lb/sec-ft



 $T_g = 42 F$  p = 82 psia  $u_g = 35 fps$   $G = 15.5 lb/ft^2$ -sec  $M_o = 17 lb_f/ft^2$ 

Methanol Film Coolant

(b) m<sub>1</sub>=0.0822 lb/sec-ft

Fig. 7 Influence of m<sub>1</sub> on Film Surface Characteristics--Low Gas Stream Velocity



 $T_g = 38 F$  p = 77 psia  $u_g = 115 fps$   $G = 50 1b/ft^2 - sec$  $M_o = 178 1b_f/ft^2$ 

Methanol Film Coolant

(a) m<sub>1</sub>=0.0144 lb/sec-ft



T<sub>g</sub>= 38 F p = 77 psia u<sub>g</sub>= 115 fps G = 50 lb/ft<sup>2</sup>-sec M<sub>o</sub>= 178 lb<sub>f</sub>/ft<sup>2</sup>

Methanol Film Coolant

(b)  $m_1 = 0.0816 \ lb/sec-ft$ 

Fig. 8 Influence of m<sub>1</sub> on Film Surface Characteristics--Higk Gas Stream Velocity

for low gas stream velocities, less than, say, 50 ft/sec. Figure 8 presents two photographs for the more important case where the gas stream velocity is relatively high (115 ft/sec in this case). Comparing these photographs with those shown in Fig. 7, it is seen that the size and the wave length of the disturbance waves on the surface of the liquid film has decreased significantly with the increase in the gas stream velocity. Moreover, comparison of Fig. 8 (a) and Fig. 8 (b) shows that the scale of the disturbance waves is not noticeably altered by increasing the rate of liquid injection  $m_1$ from 0.0144 to 0.0816 lb/sec-ft. The study of a large number of such photographs indicated that, with the exception of those few cases that were investigated where the gas stream velocity was very low, the conclusion could be drawn that the basic scale of the small disturbance waves on the surface of the liquid film was essentially independent of the rate of liquid injection.

Surface Tension  $\sigma$ . Figure 9 presents two photographs that were obtained under essentially the same flow conditions with two different liquids. The liquid shown in Fig. 9 (a) is RP-1 and the liquid shown in Fig. 9 (b) is water. The values of surface tension at the corresponding liquid temperatures in these photographs are approximately 19 dynes/cm for the RP-1 and 61 dynes/cm for the water. These photographs demonstrate that the scale of the interfacial disturbances is generally more ordered for the case where RP-1 is

flow direction ----

41

RP-i film coolant  $T_g = 431$  ,F p = 152 psia  $u_g = 89$  fps G = 41 1b/ft<sup>2</sup>-sec  $M_o = 112 1b_f/ft^2$   $m_1 = 0.036 1b/sec-ft$  $\mu_1 = 2.4 \times 10^{-4} 1b/sec-ft$ 

(a)  $\sigma = 19$  dynes/cm

C

I

I



water film coolant  $T_g = 495 F$  p = 148 psia  $u_g = 95 fps$   $G = 42 lb/ft^2$ -sec  $M_o = 125 lb_f/ft^2$   $m_1 = 0.058 lb/sec-ft$  $\mu_1 = 2.1 \times 10^{-4} lb/sec-ft$ 

(b)  $\sigma = 61 \text{ dynes/cm}$ 

Fig. 9 Influence of Surface Tension on Film Surface Characteristics the liquid. The surface of the water film is characterized by two types of disturbance waves, one having a larger scale and being more two-dimensional in appearance than the other. In fact, water was the only liquid that clearly showed two types of disturbance waves. The other liquids that were photographed, namely, methanol, butanol, and RP-1, did not exhibit that characteristic. Since the surface tension for water was significantly greater in all cases than the surface tension for the other liquids investigated\*, it is suggested that this phenomenon of the existence of two distinct types of disturbance waves for the water films was the result of the relatively high values of surface tension for that liquid.

Liquid Viscosity  $\mu_1$ . The photographs of the subject investigation did not permit a direct observation of the influence of the liquid viscosity on the surface characteristics of the liquid film. For any two photographic experiments for which the liquid viscosity differed substantially, there also occurred a substantial variation in one or more of the remaining pertinent flow parameters. However, by crudely interpolating between the results of several photographs, it was apparent that if the liquid viscosity had an influence on the surface characteristics for the liquid film, it was not pronounced. Relative to the surface tension for the liquid and to the pertinent gas stream parameters (which \*See Fig. D.1 in Appendix D.

E

E

Arran .

are discussed in the following subsection), the liquid viscosity appeared to be of secondary importance in determining the structural characteristics of the gas-liquid interface.

# 2.4.2 The Influence of the Gas Stream Parameters on the Film Surface Characteristics

The present subsection discusses how the following parameters for the gas stream alter the surface characteristics of the liquid film: (a) the gas stream velocity  $u_g$ , (b) the gas mass velocity G (defined by Eq. (2-1)), and (c) the momentum parameter  $M_0$  (defined by Eq. (2-2)). Because these three parameters are interrelated, it was not possible to conduct a series of experiments wherein any two of these parameters were held constant while the remaining parameter was varied. It was possible, however, to obtain photographic data where one of these parameters was held constant while the remaining two parameters were varied. It is thus possible, in an indirect way, to arrive at some conclusions about how the subject gas stream parameters influenced the film surface characteristics.

1 / r m 1 m 2 / r

Constant G, Variable  $u_g$  and  $M_o$ . Figure 10 presents two photographs of a liquid film where water is the liquid phase. The parameters listed in the figure show that the gas mass velocity G is essentially the same for both photographs while the momentum parameter  $M_o$  and the gas stream velocity  $u_g$  vary substantially. Figure 10 shows that the scale of the interfacial disturbances decreases as the velocity  $u_g$  (or the momentum parameter  $M_o$ ) increases. It follows, therefore, flow direction  $\rightarrow$ 

44

L.

Transformation of

a de la constitue d

an salata contra da

a shaked a

I

T<sub>g</sub>= 20 F p = 78 psia u<sub>g</sub>= 64 fps M<sub>o</sub>= 65 lb<sub>f</sub>/ft<sup>2</sup> m<sub>l</sub>= 0.0390 lb/sec-ft

Water Film Coolant





T<sub>g</sub>= 20 F p = 150 psia u<sub>g</sub>= 31 fps M<sub>o</sub>= 26 1b<sub>f</sub>/ft<sup>2</sup> m<sub>l</sub>= 0.0162 1b/sec-ft

Water Film Coolant

(b)  $G = 28 \ lb/sec-ft^2$ 

Fig. 10. Influence of G, u<sub>g</sub>, and M<sub>o</sub> on Film Surface Characteristics--G Constant that the value for the gas mass velocity G (or, the gas stream Reynolds number) does not give sufficient information to characterize the scale of the interfacial waves.

<u>Constant  $u_g$ , Variable  $M_0$  and G.</u> Figure 11 presents two photographs of a liquid film for which methanol is the liquid phase. The parameters listed in the figure show that the gas stream velocity  $u_g$  is essentially the same for both photographs while the parameters  $M_0$  and G vary significantly. Figure 11 shows that the liquid film for which the gas mass velocity G (and the momentum parameter  $M_0$ ) has the largest value exhibits the smaller interfacial scale. It follows, therefore, that the value of the gas stream velocity  $u_g$  does not give sufficient information to characterize the interfacial film structure.

Ł

Constant  $M_0$ , Variable  $u_g$  and G. Figure 12 presents two photographs for a methanol liquid film for which the momentum parameter  $M_0$  has essentially the same value while the gas stream velocity  $u_g$  and gas mass velocity G vary substantially. The scale of the interfacial wave pattern in each photograph is virtually the same. The photographic data suggest, therefore, that the momentum parameter  $M_0$  is the variable for the gas stream that most governs the resultant film surface characteristics. This observation has not been made previously in the published literature, due in part to the fact that very few (if any) detailed investigations into film surface characteristics have been made for



ł.

F

Contract of

ſ

personal and

1

R

F

I

I

1

1

 $T_g = 30 F$  p = 79 psia  $G = 25 1b/ft^2$ -sec  $M_o = 22 1b_f/ft^2$  $m_1 = 0.0138 1b/sec-ft$ 

Methanol Film Coolant

(a) u<sub>g</sub>= 58 fps



T<sub>g</sub>= 30 F p = 155 psia G = 56 1b/ft<sup>2</sup>-sec M<sub>o</sub>= 112 1b<sub>f</sub>/ft<sup>2</sup> m<sub>1</sub>= 0.0138 1b/sec-ft

Methanol Film Coolant

(b) u\_≃66 fps

Fig. 11. Influence of G, u<sub>g</sub>, and M<sub>o</sub> on Film Surface Characteristics--u<sub>g</sub> Constant



47

1

Normal Social

F

Į.

J

T<sub>g</sub>= 40 F p = 80 psia u<sub>g</sub>= 90 fps G = 39 1b/ft<sup>2</sup>-sec m<sub>1</sub>= 0.0144 1b/sec-ft

Methanol Film Coolant





T<sub>g</sub>= 30 F p = 155 psia u<sub>g</sub>= 66 fps G = 56 lb/sec-ft<sup>2</sup> m<sub>1</sub>= 0.0138 lb/sec-ft

Methanol Film Coolant

(b)  $M_0 = 112 \ 1b_f/ft^2$ 

Fig. 12. Influence of G, u<sub>g</sub>, and M<sub>o</sub> on Film Surface Characteristics--M<sub>o</sub> Constant

gas stream pressures and temperatures significantly different from ambient. That fact has precluded the independent investigation of the influence of each of the parameters  $u_g$ ,  $M_o$ , and G on the characteristics of the surface of the liquid film. (That is, for a constant value of the gas stream density, it is not possible to hold any two of the parameters  $u_g$ , G and  $M_o$  constant while the remaining parameter is varied.) The foregoing observation, however, is related in a fundamental way to the correlation for the mass transfer data that is developed in Section 3.1 of this report.

### 2.4.3 Summary

The observations made in Subsection 2.4.1 and in Subsection 2.4.2 regarding the influence of the pertinent liquid and gas stream variables on the character of the surface of a liquid film can be summarized as follows:

Į.

E

Ę

いいがきまえない ちもちち いちない

1

(a) The interfacial structure is not noticeably altered by substantial variations in the rate of liquid injection m<sub>1</sub>, except for those cases where the gas stream velocity is very low (less than approximately 50 ft/sec).

(b) The liquid viscosity  $\mu_1$  apparently does not influence the surface characteristics to any significant degree. (c) The surface tension  $\sigma$  apparently alters the surface characteristics in that two distinct wave forms were gener ally observed when water was the liquid phase (relatively high values for  $\sigma$ ) that were not observed for either butanol, methanol, or RP-1 as the liquid phase (relatively low values for  $\sigma$ ).

ş

きいき

ţ

. 19.76

ភ្នំ

(d) The scale of the small interfacial disturbances is well characterized by the value for the momentum parameter  $M_o$  (and not G or  $u_g$ ) for the gas stream; the scale of the interfacial disturbances decreases as the value for the momentum parameter  $M_o$  increases.

# 2.5 <u>The Data for Mass Transfer</u> for a Constant Velocity Gas Stream

The following procedure was employed to obtain experimental data on the rate of mass transfer from the liquid film to the gas stream. The hot gas generator was started and preselected values for the gas stream temperature, pressure, and air flow rate (gas stream Reynolds number or gas mass velocity) were established. The liquid film flow rate was established and the electric heat exchanger for preheating the liquid was turned on. Sufficient time was allowed to attain thermal equilibrium (normally 10 minutes) before the mass transfer measurements were made.

Prior to making quantitative mass transfer measurements it was necessary to insure that the liquid film was properly wetting the test surface and that the capture slot was functioning so as to remove all the liquid remaining on the test surface. First a visual observation was made through the Pyrex window in the roof of the test section to determine the wetting characteristics of the liquid film. During the course of the investigation it was found that if any

non-uniformities existed in the flow field, the liquid film would not remain in the trough that was machined into the test plate. Frequently one of the mixing screens would fail (due to the high gas stream temperature and the large pressure drop across the screen), resulting in an irregular gas flow field and a corresponding poorly wetted test plate The only experimental data that are reported herein are those for which the liquid film remained in the trough and completely wetted the 2-in. by 10-in. test surface.

Since the rate of mass transfer from the liquid film was determined as the difference between the rate of liquid injection and the rate of liquid withdrawal, the necessity of an accurate determination of these flow rates is apparent. The rate of liquid injection was determined by means of a turbine flowmeter. The flowmeter was normally calibrated before each experimental test by the time-weight technique. During the course of an experimental test, the output of the turbine flowmeter was continuously recorded on a stripchart recorder. A cyclone separator was employed to measure the rate at which liquid was withdrawn from the test plate at the capture slot\*. As mentioned previously, the effective operation of the capture slot required that a certain amount of the gas flow be withdrawn with the liquid, thus producing a two-phase withdrawal flow. The gas and liquid phases were separated by means of the cyclone separator, thereby

\*See Appendix B.4 for the design of the separator.

IJ

enabling the measurement of the liquid withdrawal rate. The cyclone separator was found to provide effective separation of the gas and liquid phases. In preliminary experimentation, it was found that under conditions of negligible mass transfer from the liquid film (cold, low velocity gas flow) the rates of liquid injection and withdrawal generally agreed to within less than 5 percent. Even with the effective performance of the cyclone separator, it was recognized that the direct contact of the gas and liquid phases during the flow through the separator could result in some error in the measurement of the rate of liquid withdrawal due to saturation of the gas flow with liquid vapor. Thus, it was desired to keep the rate of gas flow through the separator as low as possible while maintaining effective operation of the capture slot. The procedure employed is described below.

Once it was ascertained that the wetting characteristics of the film were correct, the liquid flow rate was set at the maximum value to be investigated. The valve on the separator which controlled the rate at which the gas phase was vented to the surroundings (and thus controlled the rate at which gas was withdrawn at the capture slot) was opened a small amount. A measurement was made of the rate at which the liquid was being captured at the withdrawal slot. The vent valve was then opened further and a second measurement of the rate of liquid withdrawal was made. This procedure was repeated until a further opening of the vent

Contra a la contra de la contra

THE COMPL

valve did not increase the rate of liquid withdrawal. Figure 13 presents a typical plot of the rate of liquid withdrawal as a function of the rate at which air was vented from the cyclone separator. The figure illustrates that the rate of liquid withdrawal increases rapidly at first with an increasing vent flow rate and then reaches a maximum, or plateau, such that a further increase in the vent flow rate does not appreciably affect the rate of liquid withdrawal. The increase in  $m_{2}$  with vent gas flow rate reflects an increase in effectiveness of the capture slot in withdrawing the liquid from the test surface. When the vented gas flowrate is too low, some of the liquid spills over the capture slot and then m<sub>2</sub> is not the total film flowrate at the point of withdrawal. As the vented gas flowrate is increased, the spillage is reduced until with sufficient vent flow the capture slot collects and withdraws all of the liquid on the test surface. In all of the experimental tests, the vent flow rate was always set equal to or greater than the value required to realize no change in the rate of liquid withdrawal with an increase in the vent flow rate. For the magnitude of the vent flow rate normally required, rough calculations showed that even if the gas phase that entered the separator was completely dry air, and the gas phase that was vented from the separator was completely saturated with vapor at the corresponding pressure within the separator, the amount of mass transfer that would have occurred



Fig. 13. Rate of Liquid Withdrawal as a Function of Vent Flow Rate for Separator

53

i

balla de la consect

within the separator would be negligible in comparison to the total rate of mass transfer that was normally realized from the liquid film to the gas stream (not more than 5 percent).

After the vent flow rate was set, the quantitative mass transfer data were obtained by decreasing the rate of liquid injection in incremental stages and making the corresponding measurements of the rate of liquid withdrawal. Thus, during any one experimental test, the only parameter that was varied was the rate of liquid injection.

Figure 14 illustrates the typical nature of the experimental data for mass transfer that were obtained in the present investigation. The data shown in Fig. 14 are for methanol as the liquid phase and they are for a nominal gas stream temperature and pressure of 400F and 75 psia, respectively. Plotted in the figure is the rate of liquid withdrawal, m<sub>2</sub>, as a function of the rate of liquid injection,  $m_1$ , with the gas stream .elocity,  $u_q$ , as a parameter. The solid curve shown at an inclination of 45° represents the condition of no mass transfer; at any point on that curve, the rate of liquid injection is equal to the rate of liquid withdrawal. The vertical distance between the solid curve and any one datum point represents the total rate of mass transfer from the liquid film for that particular combination of flow conditions. Figure 14 illustrates the important fact that the rate of mass transfer from the

H

Π




liquid film to the gas stream was found to be <u>a function of</u> <u>the rate at which liquid was injected onto the test plate</u>; the theory for simple mass transfer, however, does not predict any such dependence on the rate of liquid injection. Figure 14 shows further that, to within the accuracy of the experimental data, the total rate of mass transfer from the liquid film to the gas stream increased linearly with the rate of liquid injection. This fact was not foreseen before the experimental investigation was begun, but it greatly simplified the correlation of the experimental data, as is discussed in detail in Section 3.1 of this report.

The remainder of this section discusses how the pertinent flow parameters for both the liquid and gas phases influenced the net rate of mass transfer from the liquid film to the gas stream. The mass transfer data of some of the experimental tests is considered in detail. Although it was not feasible to discuss the data of each experimental test of the present investigation, it should be noted that the illustrative cases that are considered are representative of all of the tests of the present investigation. That includes not only those tests for which a constant velocity, hot gas stream was employed, but also those tests where the hot gas stream was accelerated over the wetted test plate and those tests for which the gas flow was not heated (cold flow, constant velocity gas stream). An inclusive graphical presentation of the data for mass transfer is given in

Appendix G.

۲۰٫۵۰ مارد. مارد الارد. ۲۰٫۵۰ مارد مارد الارد الارد Subsection 2.5.1, which follows below, discusses the influence of the liquid surface tension and the liquid viscosity on the total rate of mass transfer. Subsection 2.5.2 discusses the influence of the gas stream flow parameters on the measured rate of mass transfer from the liquid film to the gas stream.

2.5.1 The Influence of the Surface Tension and the Liquid Viscosity on the Data for Mass Transfer

Figure 15 serves to illustrate the influence of both the surface tension  $\sigma$  and the liquid viscosity  $\mu$  upon the experimental data for mass transfer. Presented in the figure are the experimental data for mass transfer that were obtained for the experimental tests 113, 205, and 37. The liquids employed in these three tests were methanol, butanol, and water, respectively. The pertinent flow parameters for each test are listed in the figure. The dashed curves presented in Fig. 15 represent the condition of simple mass transfer as computed for the pertinent gas flow conditions\*. If there had been no entrainment of liquid into the gas stream and if the liquid film had presented a smooth boundary to the developing boundary layer, the experimental data would lie along the dashed curves. Thus, the vertical distance between any one of the dashed curves and the datum

\*Subsection 3.1.2 discusses how the simple mass transfer rates were evaluated. Table G.1 of Appendix G lists the average value for the simple mass transfer rate for each experimental test that is reported.



6



:

Π

Î

IJ

points for that curve represents the contribution to the total rate of mass transfer that is due to the interfacial phenomena (film surface roughness and entrainment).

The three cases presented in Fig. 15 were chosen for purposes of illustration because the pertinent flow parameters for the gas stream, with the exception of the gas stream temperature,  $T_g$ , are essentially the same while the viscosity and surface tension for the liquid phase are significantly different\*.

Compare first the data that were obtained for water (test 37) and the data that were obtained for methanol (test 113). Figure 15 shows that the mass transfer due to the interfacial phenomena is much greater for the case where. methanol is the liquid phase. Referring to the values for the pertinent parameters listed in Fig. 15, it is seen that only the surface tension  $\sigma$  is significantly different for these two cases. These data suggest, therefore, that the surface tension influences to a considerable degree the rate of mass transfer from a thin liquid film to a high velocity gas stream.

Comparing the data for methanol (test 113) and the data for butanol (test 205) in Fig. 15, we note that the measured rates of mass transfer are essentially the same for both cases. Moreover, the simple mass transfer contribution is

\*The temperature employed to evaluate the liquid viscosity  $\mu_1$  and the surface tension  $\sigma$  was the average of the liquid temperature at the point of liquid injection and at the point of liquid withdrawal.

virtually the same for each case so that the contribution due to interfacial phenomena is the same. In comparing these two cases, it is important to note that all of the parameters listed in the figure are nearly the same, except for the liquid viscosity  $\mu_1$ . The liquid viscosity for butanol for test 205 is roughly 2.5 times that for methanol for test 113. These data suggest, therefore, that the liquid viscosity has no appreciable influence on the rate of mass transfer from a liquid film to a high velocity gas stream. The significance of that result was previously commented on in Section 2.4. It was pointed out that a number of investigators have proposed that a liquid-film Reynolds number (which is inversely proportional to the liquid viscosity  $\mu_1$ ) should be of primary importance in determining the rate at which liquid is entrained from the surface of a thin liquid film by a high velocity gas stream. The results of the present investigation, however, do not support that proposal.

It should be emphasized that the aforementioned dependence of the net mass transfer rate on the surface tension  $\sigma$  was exhibited by all of the experimental data of the present investigation, including those obtained under cold-flow conditions, and those realized for the case where the gas stream was accelerated over the wetted test plate.

2.5.2 The Influence of the Gas Stream Parameters on the Data for Mass Transfer

Figure 16 serves to illustrate how the pertinent parameters for the gas stream ( $u_g$ , G, and  $M_o$ ) influenced the experimental data for mass transfer. The figure presents the data for mass transfer that were obtained for the experimental tests 210, 205, and 213. The flow conditions for each test are indicated in the figure. The liquid phase employed in each test was butanol. The dashed lines in Fig. 16 indicate the condition of simple mass transfer.

Compare first the experimental tests 210 and 205. Figure 16 shows that the total rate of mass transfer for test 205 was significantly greater than for test 210. The contribution due to simple mass transfer, however, was essentially the same for both cases. It follows, therefore, that the mass transfer due to the interfacial phenomena was greater for test 205 than it was for test 210. The parameters listed in Fig. 16 indicate that the gas stream velocity,  $u_{o}$ , and the momentum parameter for the gas stream,  $M_{o}$ , differed substantially for the two cases, while the gas mass velocity, G, was essentially the same for each case. Consider next the data shown in Fig. 16 for test 210 and for test 213. The parameters listed in the figure indicate that the momentum parameter  $M_{o}$  was the same for both cases, while the remaining parameters show a varying degree of difference. The realized rates of mass transfer for the

4 IU - 1

いた ちんまい 「王の書を見たい」



(11-202/dl) lewerbdie biupil to eser .

S<sup>M</sup>



62

l

1

1

two cases are essentially the same. The data of Fig. 16 suggest, therefore, that the momentum parameter  $M_0$  for the gas stream characterizes, to a large degree, the extent of the contribution to the net rate of mass transfer that is due to the interfacial phenomena. Again it should be emphasized that this observation was supported by all of the experimental data of the present investigation, and not by just those that are presented in Fig. 16.

## 2.5.3 Summary

The foregoing discussion suggests that the contribution to the net rate of mass transfer that is due to the interfacial phenomena:

(a) increases in a linear manner with the rate of liquid injection, m<sub>1</sub>;

(b) increases with decreasing values for the surface tension  $\sigma$ ;

(c) is apparently not a function of the viscosity for the liquid,  $\mu_{\rm l};$  and

(d) is characterized by the momentum parameter for the gas stream,  $M_0$ , with the rate of mass transfer increasing with increasing values for  $M_0$ .

These observations form the basis for the mass transfer correlation that is developed in Section 3.1 of this report.

### 2.6 <u>The Data for Mass Transfer for an</u> <u>Accelerating Gas Stream</u>

The stagnation temperature for the gas stream was maintained constant at 400F for that phase of the investigation where the gas stream was accelerated over the wetted test surface. Moreover, the Mach number profile was fixed, such that the Mach number increased from 0.14 at the point of liquid injection to 0.51 at the point of liquid withdrawal\*. The stagnation pressure was varied from 50 to 150 psia in increments of 25 psia, and both water and methanol were employed as the liquid phase. The experimental procedure employed in these investigations was basically the same as that outlined in the previous section.

Figure 17 presents all of the experimental data that was obtained in this portion of the investigation. Plotted in the figure is the rate of liquid withdrawal as a function of the rate of liquid injection with the stagnation pressure,  $p^{0}$ , as a parameter. The experimental data obtained for both water and methanol as the liquid phase are shown in Fig. 17. Again, like the data that was presented in Figs. 14, 15, and 16, the total rate of mass transfer (the vertical distance between the line for no mass transfer and any one datum point) increases linearly with the rate of liquid injection. The data in Fig. 17 demonstrate that for the same flow conditions (same total pressure, total temperature, and degree

\*i.e., only one area-contraction ratio (one aluminum wedge in Fig. 5) was employed in the investigation



(31-292/df) [swendativ biupil to ester . Sm

65

ł

E

land and a

E

PT BARRENN'S

ľ

. . .

of acceleration of the gas stream) the rate of mass transfer was much greater for the case where methanol was the liquid phase than for the case where water was the liquid phase. That is due, in part, to the fact that the heat of vaporization for water is roughly 3 times that for methanol, so that for a given rate of heat transfer to the surface of the liquid film, the rate of vaporization of methanol is greater than is the rate of vaporization of water. But, more importantly, the rate of entrainment of methanol into the gas stream was much greater than was the rate of entrainment of water because the surface tension for methanol is roughly one-third that of water. The rather strong dependence of the rate of mass transfer from the liquid film into the gas stream on the surface tension was discussed in Subsection 2.5.1 of the present report.

A correlation is developed in Section 3.3 for the mass transfer data presented in Fig. 17. It will be shown that the data correlation is a natural extension of the correlation that is developed in Section 3.1 for the experimental data for mass transfer for the case where the gas stream velocity was essentially constant over the length of the wetted test plate.

....

1

### 2.7 The Data for the Maximum Liquid Film Temperature

In general,  $T_{l}$ , the temperature of the liquid at any point within the liquid film, is less than  $T_{s}$ , the temperature at the surface of the liquid film, and is greater than  $T_{w}$ , the temperature at the surface of the wetted wall. However, for the special case where the wetted wall is adiabatic (as was the case for the present investigation), it is possible to obtain the condition

 $T_s = T_1 = T_w$ 

That condition occurs when all of the energy transfer to the liquid film goes into the vaporization of the liquid in the film\*. Moreover, at the point downstream from the point of liquid injection where that condition is obtained, the temperature  $T_{1}$  (= $T_{w}$  =  $T_{s}$ ) is a maximum. That maximum liquid temperature is denoted herein by  $T_{1.m}$ .

The maximum liquid temperature,  $T_{1,m}$ , was determined experimentally for the present investigation by measuring  $T_{1,2}$ , the temperature for the liquid coolant at the point of liquid withdrawal. Recall that the attempt was made in the subject investigation to preheat the liquid before injection so that the sensible enthalpy rise that would occur for the liquid after it was introduced onto the test plate would

\*That is, none of the energy transfer to the surface of the liquid film goes to increase the sensible enthalpy for the liquid in the film.

be minimized. Therefore, for those tests where it was possible to completely preheat the liquid before injection, it follows that the temperatures  $T_{1,m}$  and  $T_{1,2}$  were equivalent. Moreover, for those experimental tests wherein the liquid could not be completely preheated before injection (due to limitations in the heating capacity of the heat exchanger that was employed), it was found that the measured withdrawal temperature was not influenced significantly by the rate at which liquid was injected onto the test plate, thereby suggesting that the rate of energy transfer to the liquid film was sufficiently rapid to cause the allowable increase in the sensible enthalpy for the liquid to occur before the liquid film reached the capture slot. It follows, therefore, that the temperaturess  $T_{1,2}$  and  $T_{1,m}$  could be reasonably equated for all of the experimental tests of the subject investigation.

The liquid temperature  $T_{1,2}$  was measured with a chromelalumel thermocouple\*, and the indicated temperature was continuously recorded on a stripchart recorder. Experimental values for  $T_{1,2}$  were obtained for each of the four liquids that was employed, and for the entire range of values for the gas stream parameters that was investigated\*\*. Table G.1 of Appendix G lists the average value of  $T_{1,2}$  for each of the experimental tests that are reported herein. Also,

\* See Fig. B.6 of Appendix B for the physical location of the thermocouple.

\*\*Refer to Tabl⊳ 1 in Section 2.3.

.

there is a graphical way in which the data can be presented that is of interest. In most previous investigations of liquid-film cooling (e.g., 20, 22, 23), the maximum temperature for the liquid film was taken to be the boiling temperature for the liquid at the prevailing pressure. However, it is demonstrated theoretically in Refs. (21) and (26), and in Section 3.4 of this report, that the temperature of the liquid film can never be equal to that boiling temperature. Figure 18 graphically illustrates this point for the data obtained in the subject investigation. Presented in Fig. 18 is a plot of the boiling temperature for the liquid at the prevailing pressure versus  $T_{1,2}$ , the measured liquid-film Data are presented for water, methanol, butanol, temperature. and RP-1. Figure 18 shows that the maximum film temperature can be substantially less than the boiling temperature at the prevailing pressure. The maximum difference between these two temperatures encountered in the present investigation is approximately 355F, and it was realized for RP-1 at a pressure of 150 psia and a gas stream temperature of 400F.

м`,

ż





70

\$\*\*\*\*\*

to set the set

Π

IJ

 $\square$ 

### 3. CORRELATION OF THE EXPERIMENTAL DATA

The present section is concerned with the correlation of the experimental data that were discussed previously in Section 2. It should be stated at the outset that it was the intention of the present investigation to develop formulae that could be employed to accurately design a liquidfilm cooled system but which would not involve excessive computations. Therefore, the present investigation sought to consider the details of the complex phenomena that characterize the interaction of a high velocity gas stream and a thin liquid film only to the extent necessary to develop those design formulae. These interfacial phenomena have been the object of a large number of experimental and/or analytical investigations reported in the literature. To illustrate, references (5, 32) were concerned with the amplitude and wave length spectra that the gas-liquid interface exhibits for various liquid and gas stream flow conditions; references (33, 34) were concerned, in part, with applying stability theory to analyze the phenomenon of film instability; and references (28, 35) investigated the structure of the boundary layer which develops over the surface of a liquid film. It is virtually impossible; however, to

È

----

utilize these very interesting data--particularly in a quantitative sense--to answer the basic engineering question: what is the net rate of mass transfer from a thin liquid film to a high velocity, hot gas stream? The relatively simple formulae developed herein answer that question explicitly.

Section 3.1 presents a correlation of the data for mass transfer with a constant velocity gas stream. Correlation of the data for mass transfer where the gas stream was accelerating over the wetted test surface is discussed in Section 3.3 Analysis of the latter data requires first the consideration of the problem of how the net rate of mass transfer varies with the liquid-film cooled lenght 1; Section 3.2 considers that problem. Section 3.4 presents an analysis of the data for the maximum liquid temperature.

## 3.1 <u>Correlation of the Data for Mass Transfer</u> for a Constant Velocity Gas Stream

### 3.1.1. General Discussion

The experimental data for mass transfer for the subject investigation showed that for fixed gas stream flow conditions, the rate of liquid withdrawal, m<sub>2</sub>, could be approximately expressed as a linear function of the rate of liquid

e'i

injection, m<sub>1</sub>\*. Thus

生きななのい えて よりまる 妻一説 「王二」「副語のいを成

「「「「」

E

T H H

Line.

$$m_2 = -A + B m_1 \tag{3-1}$$

If  $m_1$  is subtracted from both sides of Eq. (3-1), the resultant expression can be written in the form

$$m_1 - m_2 = A + (1 - B) m_1$$
 (3-2)

Noting that the quantity  $(m_1 - m_2)$  is simply m', the net rate of mass transfer from the liquid film to the gas stream, we obtain from Eq. (3-2) the expression

$$m' = A + (1-B) m_1$$
 (3-3)

Equation (3-3) expresses the net rate of mass transfer from the liquid film as a linear function of the rate of liquid injection.

In the original statement of the problem in Section 1.2, it was indicated that m' was made up of two fundamental contributions: (a) the contribution due to the evaporation of the liquid in the liquid film, and (b) the contribution due to the entrainment of bulk liquid into the gas stream. The problem, therefore, is to determine how each of these two contributions can be evaluated from Eq. (3-3). With

\*Refer to Figs. 14, 15, and 16 and to Appendix 6.

reference to that problem, consider the following two points: (a) The phenomen of liquid entrainment generally has been associated with the shearing off of liquid droplets from large scale disturbance waves (instability waves) on the surface of the liquid film. Therefore, it is reasonable to assume that the rate at which liquid is entrained fro the surface of these large scale disturbance waves into the gas stream would increase with the frequency of occurrence of these waves. It has been demonstrated (10) that the frequency of occurrence of these waves increases with  $m_1$ , the rate at which liquid is introduced onto the wetted wall surface. Therefore, the rate of entrainment of liquid into the gas stream should be proportional to  $m_1$ .

(b) It can be argued that the evaporation contribution to the total rate of mass transfer from the liquid film to the gas stream should be largely independent of the flow parameters for the liquid phase. The extent to which the turbulent transport of heat to the surface of the liquid film is intensified because of the effective roughness of the surface of the liquid film should be primarily a function of the physical scale of the substructure on the surface of the liquid film. The photographic data discussed in Section 2.4 suggested, however, that while the scale of that substructure is influenced to a considerable degree by the flow parameters for the gas stream (e.g., u<sub>g</sub>, G, and M<sub>o</sub>), it is not noticeably altered either by significant variations in the rate

at which liquid is injected onto the test plate or by significant variations in the physical properties for the liquid phase. It thus seems reasonable to assume, at least as a first approximation, that the evaporative contribution is not dependent on the parameters for the liquid phase.

The foregoing two points and the form of the Eq. (3-3) suggest the simple hypothesis that (a) the first term on the right-hand side of Eq. (3-3) represents the contribution to the total mass transfer rate that is due to the evaporation of the liquid in the film, and (b) the second term on the right-hand side of Eq. (3-3) represents the contribution that is due to the entrainment of bulk liquid from the surface of the liquid film into the gas stream. Thus, it is hypothesized herein that

A = the evaporation contribution (1-B) m<sub>1</sub> = the entrainment contribution The ultimate justification for this simple hypothesis is the fact that it resulted in a satisfactory correlation of the experimental data. This will be demonstrated in Subsection 3.1.5 of the present report.

Referring again to the statement of the subject problem presented in Section 1.2, it was indicated that the evaporative contribution to the total rate of mass transfer m' could be subdivided into (a) that due to the simple mass transfer and (b) that due to the interfacial phenomena (the effective surface roughness). If the term A in Eq. (3-3)

1 4.4

gives the total contribution to m' that is due to evaporation, and if  $m'_s$  denotes the rate of simple mass transfer, then a parameter r, termed the <u>roughness parameter</u>, can be

defined such that

 $A = (1 + r) m'_{s} (lb/sec-ft)$  (3-4)

The parameter r is a measure of the degree to which the interfacial structure intensifies the turbulent transport of heat from the gas stream to the surface of the liquid film. For the limiting case where the surface of the liquid film becomes hydrodynamically smooth, the parameter r goes to zero, and the problem reduces to the classical problem of simple mass transfer.

In addition to the roughness parameter r, it is convenient to define an <u>entrainment parameter</u> e such that

 $e_0 = 1 - B$  (3-5)

where B is defined by Eq. (3-1). The discussion of Section 2.5 suggests that  $e_0$  is primarily a function of the surface tension  $\sigma$  for the liquid phase and of the momentum parameter  $M_{c}$  for the gas stream. The subscript "o" on  $e_{c}$  indicates that  $e_{0}$  was determined from experimental data for which the liquid-film cooled length was maintained constant at 10 inches (1 = 1<sub>0</sub> = 10 inches). Section 3.2 of the subject report shows how the experimental results obtained for that fixed film cooled length can be utilized to analyze the general case of liquid-film cooling where the liquid-film cooled length 1 is arbitrary.

Substitution of Eqs. (3-4) and 3-5) into Eq. (3-3) results in the following expression. Thus

$$m' = (1 + r) m_{s}' + e_{0} m_{1}$$
 (3-6)

Equation (3-6) is the primary equation that was utilized in the present investigation to correlate the net rate of mass transfer from the liquid film to the gas stream. The significance of each of the terms in Eq. (3-6) is shown graphically in Fig. 19. The lower solid curve in the figure represents a least-squares fit of the data for mass transfer that were realized for a typical experimental test. The upper solid curve in Fig. 19 represents the condition of no mass transfer, and the upper dashed curve is the curve that is predicted by the simple theory for mass transfer. The lower dashed curve represents the curve that would be predicted by the present theory, accounting for the roughness of the film, if there was no entrainment of liquid into the gas stream. Figure 19 shows how each of the terms in Eq. (3-6),  $m'_s$ ,  $rm'_s$ , and  $e_{0}m_1$ , can be conveniently interpreted graphically.

Figure 19 and Eq. (3-6) show that the subject problem



78

a suggestion of

of correlating the experimental data for m' is reduced to that of calculating the simple mass transfer rate  $m'_s$  and to the correlation of the parameters r and  $e_0$  on the basis of the experimental data. Subsection 3.1.2 discusses the method that was employed to calculate  $m'_s$  in the present investigation. Subsection 3.1.4 and 3.1.5 discuss the techniques for correlating the parameters  $e_0$  and r, respectively, in terms of pertinent flow parameters. Subsection 3.1.6 then illustrates the agreement of the resultant data correlation with the experimental mass transfer data of the present investigation.

# 3.1.2 The Evaluation of the Rate of Simple Mass Transfer m's\*

The net rate of simple mass transfer from the liquid film to the gas stream is related to the local rate of mass transfer by the following expression. Thus

$$m'_{s} = \int_{x_{1}}^{x_{2}} m''_{s} dx \qquad (3-7)$$
  
=  $(x_{2} - x_{1})\overline{m}''_{s} = 1_{0} \overline{m}''_{s} \qquad (3-8)$ 

\*The primary reference for this section is the textbook by Kays (36) on convective heat and mass transfer.

79

ľ

F

E

where

m's = the net rate of simple mass transfer
m's = the local rate of simple mass transfer
lo = the film cooled length ( = 10 inches)
m's = the average value for m's over the film cooled
length lo

A method is outlined in the present section for the evaluation of the local rate of simple mass transfer  $m_s^{"}$ . The method consists of two basic steps: (a) writing an energy balance across the gas-liquid interface so that  $m_s^{"}$ can be directly related to  $q_s$ , the local rate of heat transfer from the hot gas stream to the surface of the liquid film, and (b) determining  $q_s$  from the comprehensive theory that is presented by Kays (36) for that problem. The solution for  $m_s^{"}$  is then completed by developing a closedform expression for the average rate of simple mass transfer,  $\overline{m}_s^{"}$ , that appears in Eq. (3-8).

The Energy Balance at the Gas-Liquid Interface. Figure 20 illustrates a general interfacial element placed at the surface of a stable, non-reactive liquid film with the dotted lines representing the surfaces considered in writing an energy balance for the element. The different energy fluxes involved are defined as follows:

(1)	۹s	= (i	<sup>k</sup> ∂t/∂y)s	=	the rate at which ferred across the conduction.	energy is S-surface	trans- by
(2)	۹L	= ()	< at/ay)	=	the rate at which ferred across the conduction.	energy is L-surface	<b>trans</b> - by



~

1

ſ



(3)  $m_s^{"}h_{v,s} =$  the rate at which energy is convected across the S-surface by evaporated vapor.

- (4)  $m_s^{"} h_L$  = the rate at which energy is convected across the L-surface by the evaporating liquid.
- (5) q<sub>r,S</sub> = the rate at which energy is transferred across the S-surface by radiation.
- (6) q<sub>r,L</sub> = the rate at which energy is transferred across the L-surface by radiation.

The energy balance for the interface element illustrated in Fig. 20 can be written as follows:

$$q_{S} = q_{L} + m_{S}^{"} (h_{v,S} - h_{L}) + (q_{r,L} - q_{r,S})^{T}$$
 (3-9)  
I II III III IV

Term III in Eq. (3-9) can be rewritten in the form

m" H

ž

こう しっている 山田田田田 、、大田田田田田田田

where  $H_v$  is the latent heat of vaporization for the liquid coolant which corresponds to the saturation condition that is prescribed by the film surface temperature  $T_s$ . Furthermore, for the temperature levels encountered in the subject investigation, term IV in Eq. (3-9) can be considered negligible in comparison to a term such as term III. Thus, neglecting that term, Eq. (3-9) can be rewritten to yield

Į

2. R. Hell.

X

 $q_{s} = q_{L} + m_{s}^{"} H_{v}$  (3-10)

The term  $q_{L}$  in Eq. (3-10) is that portion of the heat transfer to the surface of the liquid film which crosses the

interface and goes into raising the sensible enthalpy of the liquid and/or is transferred through the wetted plate to the surroundings. In the subject investigation, the test plate was essentially adiabatic so that the heat transfer across the test plate to the surroundings was negligible. Moreover, an attempt was made in the investigation to preheat the liquid before injection so that no rise in sensible enthalpy would occur for the liquid once it was introduced onto the test plate. The elimination of the rise in sensible enthalpy for the liquid would make the quantity q, in Eq. (3-10) equal to zero and thus simplify the energy balance However, due to limitations in as mentioned in Section 2.2. the heating capacity of the heat exchanger employed to preheat the liquid, some rise in the sensible enthalpy for the liquid usually occurred after the liquid was injected. То correct for that fact, the term  $q_1$  in Eq. (3-10) was approximated by the expression\*

$$q_{L} = m_{s}^{"} (h_{1,2} - h_{1,1})$$
 (3-11)

where

 $h_{1,2}$  = sensible enthalpy for the liquid at the point of withdrawal, x =  $x_2$  $h_{1,1}$  = sensible enthalpy for the liquid at the point of injection, x =  $x_1$ 

\*This approximation is the same as that employed in Refs. (19, 29).

Substitution of Eq. (3-11) into Eq. (3-10) results in the following basic expression for the interfacial energy balance. Thus

$$\frac{q_{S}}{m_{S}^{*}} = H_{v} + h_{1,2} - h_{1,1}$$
(3-12)

It is convenient to introduce a parameter  $\phi$  that is termed herein the <u>energy balance parameter</u>, where  $\phi$  is defined as the group of terms on the right-hand side of Eq. (3-12). Thus

$$\phi = H_v + h_{1,2} - h_{1,1}$$
 (3-13)

or  $\phi = \frac{q_S}{m_S^{\prime\prime}}$  (3-14)

Since the energy balance parameter  $\phi$  is expressed solely in terms of quantities that can readily be evaluated, the problem of determining m<sup>\*</sup><sub>S</sub> thus reduces to that of evaluating  $q_S$ .

The Evaluation of  $q_S$ . It is convenient to express  $q_S$  in terms of St<sub>s</sub>, the local Stanton number for simple heat transfer, where St<sub>s</sub> is defined by

 $St_{s} = \frac{q_{S}}{G c_{pg} (T_{g} - T_{s})}$ (3-15)

where

- G = the mass velocity for the gas stream
- c = the specific heat at constant pressure for the gas stream
- $T_n$  = the temperature for the gas stream

 $T_s$  = the temperature at the surface of the liquid film

The complete evaluation of the local Stanton number St<sub>s</sub> requires that the effect of the following on the transport phenomena be accounted for: (a) the thermal entrance length  $x_t = x_1$ , (see Fig. 6); (b) the dependence of the physical properties for the gas phase on the temperature T and on the local concentration of the injected vapor C; (c) the rate of transfer of vapor from the surface of the liquid film into the gas phase; and (d) non-unity values for the Prandtl number Pr for the gas phase. If St<sub>so</sub> denotes the local value for the Stanton number for the case where none of these effects are present, then the Stanton number St<sub>s</sub> can be related to St<sub>so</sub> through a number of correction factors that are denoted by F. Thus

$$St_{s} = F_{x_{\star}} F_{T}F_{C} F_{m} F_{Pr} St_{so}$$
(3-16)

where

- the solution

4

and the second s

A Drown and

- $F_x$  = the correction factor for thermal entrance t length effects  $(x_t = x_1)$
- $F_T$  = the correction factor for the dependence of the physical properties of the gas on the temperature

- F<sub>C</sub> = the correction factor for the dependence of the physical properties for the gas on the concentration of the injected vapor C
- F<sub>m</sub> = the correction factor for the injection of the mass (i.e., vapor) into the developing boundary layer\*
- F<sub>Pr</sub> = the correction factor for a non-unity Prandtl number for the gas stream

Kays (36) suggests that for the case where the gas stream velocity is constant\*\*, the correction factors in Eq. (3-16) can be evaluated by the following empirical equations. Thus

$$F_{x_{t}} = (1 - \frac{x_{t}}{x})^{-0.12}$$
(3-17)

$$F_{T} = \left(\frac{T_{g}}{T_{s}}\right)$$
 (3-18)

$$F_{\rm C} = \left(\frac{M_{\rm s}}{M_{\rm g}}\right)^{0.4}$$
 (3-19)

$$F_{m''} = \frac{\ln(1+B_{h})}{B_{h}}$$
(3-20)

$$F_{Pr} = (Pr)^{-0.4}$$
 (3-21)

where

- Mg = the molecular weight for the gas stream Ms = the molecular weight for the gas-vapor mixture at the surface of the liquid film
- \* F<sub>m</sub><sup>+</sup> accounts for the thermal "blockage" effect that was discussed in Section 1.3.

\*\*This restriction actually applies to only  $F_{x_{+}}$ .

B<sub>h</sub> = a dimensionless heat-mass transfer parameter

$$= \frac{c_{pg}(T_g - T_s)}{\phi}$$
 (3-22)\*

To complete the solution to  $q_S$ , both the term  $St_{so}$  in Eq. (3-16) and the term  $M_s$  in Eq. (3-19) must be evaluated. The following classical expression for turbulent heat transfer to a flat plate was employed to evaluate  $St_{so}$  (36). Thus

$$St_{so} = 0.0295 \text{ Rey}^{-0.2}$$
 (3-23)

where

Rex = the Reynolds number for the gas stream based on the characteristic length x

$$\equiv \frac{Gx}{\mu_g}$$
 (3-24)

μ<sub>g</sub> = the dynamic viscosity for the gas stream evaluated at T<sub>g</sub>

The molecular weight  $M_s$  is related to  $M_v$ , the molecular weight for the vapor, and  $M_g$ , the molecular weight for the free-stream gas, by

$$\frac{1}{M_{s}} = \frac{C_{s}}{M_{v}} + \frac{(1 - C_{s})}{M_{g}}$$
(3-25)

\*If compressibility effects are important (Mach number greater than, say, 0.5), then the temperature difference  $(T_g - T_s)$  should be replaced by  $(T_{g,aw} - T_s)$ , where  $T_{g,aw}$  is the adiabatic-wall recovery temperature for the gas stream.

where

C<sub>s</sub> = the concentration for the vapor at the surface of the liquid film The vapor concentration C<sub>s</sub> can be determined by the follow-

ing approximate expression (36).

$$C_s = \frac{B_h}{1 + B_h}$$
 (3-26)\*

Thus

The Evaluation of  $m'_s$ . Equations (3-14), (3-15), (3-16) and (3-22) can be combined to obtain the following basic expression for  $m''_s$  as a function of the coordinate x. Thus

 $m_{s}^{"} = G B_{h} F_{x_{t}} F_{T} F_{C} F_{m}^{"} F_{Pr} St_{so} \qquad (3-27)$ 

For the subject case where the gas stream velocity is constant over the film cooled length  $l_0$ , it can be reasonably assumed that only the terms  $F_{x_t}$  and  $St_{so}$  in Eq. (3-27) are a function of the coordinate x. Therefore, for that special case, the result of the substitution of Eq. (3-27) into Eq. (3-7) can be written as

$$m'_{s} = G B_{h} F_{\bar{i}} F_{C} F_{m''} F_{Pr} \int_{x_{1}}^{x_{2}} F_{x} St_{so} dx$$
 (3-28)

-

1

ł

 $\Pi$ 

IL IS NOT PUSSIBLE tO EVALUATE the integral in Eq. (3-28) in

\*See the footnote following Eq. (3-87a) of the present report for an explanation of the origin of this equation.

closed form if Eq. (3-17) is substituted for  $F_{x_t}$ . However, Reynolds, Kays, and Kline (37) have developed the following approximate expression for that integral term:

$$\int_{x_{1}}^{x_{2}} F_{x_{t}} \operatorname{St}_{so} dx = \left\{ \frac{5}{4} \frac{\left(\frac{x_{2}}{x_{1}}\right)^{0.2} \left[ \left(\frac{x_{2}}{x_{1}}\right)^{0.9} - 1 \right]^{\frac{8}{9}}}{\left[ \left(\frac{x_{2}}{x_{1}}\right)^{-1} \right]} \right\} \right\} = 1_{0} \operatorname{St}_{so,2}$$
(3-29)

where  $St_{so,2}$  is the value for  $St_{so}$  that is obtained from Eq. (3-23) with  $x = x_2$ . For a value of  $x_1 = 40$  inches and a value of  $x_2 = 50$  inches, the non-dimensional term inside the brackets { } in Eq. (3-29) has a value of 1.374. Substituting that value into Eq. (3-29) and the resultant expression into Eq. (3-28), one obtains the basic expression that was employed in the present investigation to evaluate the <u>rate of simple mass transfer</u> from the liquid film into the <u>constant velocity gas stream</u>. Thus

Ŀ

F

 $m'_{s} = 1.374 \text{ G B}_{h} F_{T} F_{C} F_{m''} F_{Pr} l_{o} St_{so,2}$  (3-30)

Table G.1 of Appendix G lists the average value for the simple rate of mass transfer,  $m'_s$ , that was calculated for each experimental test. The range of values that was computed for each of the dimensionless terms in Eq. (3-30) and also for the interfacial concentration C<sub>s</sub> and the heat-mass.

transfer parameter B<sub>h</sub> is summarized as follows:

 $F_{T} = 1.015 \text{ to } 1.150$   $F_{C} = 0.976 \text{ to } 1.096$   $F_{m''} = 0.86 \text{ to } 0.96$   $F_{Pr} = 1.168 \text{ (for all cases)}$   $St_{so,2} = 0.0021 \text{ to } 0.0030$   $C_{s} = 0.081 \text{ to } 0.247$   $B_{h} = 0.083 \text{ to } 0.328$ 

## 3.1.3 The Correlation of the Entrainment $Parameter e_0$

The present section is concerned with the correlation of the entrainment parameter e, that occurs in the primary equation (3-6). The procedure that was employed herein to develop the correlation was purely inductive. That is, the procedure that was employed was to first note how each of the pertinent flow parameters influenced the rate of entrainment of liquid from the liquid film into the gas stream, and to then develop a general correlation that would reflect each of those observations. Thus an analytical model is not postulated in this report for the approximate analysis of the entrainment phenomenon. Moreover, the method of dimensional analysis was not directly employed to develop the subject correlation for e. The ultimate justification for the simple procedure that was employed is the fact that it resulted in a satisfactory correlation of the parameter e in terms of the pertinent variables for both
the gas and liquid phases. The next subsection of this report, Subsection 3.1.4, does, however, consider in some detail the possible physical significance of the resultant correlation that was developed for  $e_0$ . The discussion in Subsection 3.1.4 suggests a simple procedure for correlating the roughness parameter "r" that was defined by Eqs. (3-4) and (3-6). Subsection 3.1.5 presents the correlation for the parameter r.

The discussion of the experimental data for mass transfer that was presented in Section 2.5, and summarized in Subsection 2.5.3, showed that the two parameters which most influenced the net rate of entrainment of liquid from the liquid film into the gas stream were the momentum parameter  $M_0$  for the gas stream and the surface tension  $\sigma$  for the liquid phase. The rate of entrainment of liquid increased with increasing values for  $M_0$  and with decreasing values for  $\sigma$ . In view of those observations, the attempt was made to correlate the parameter  $e_0$  in terms of a dimensional entrainment group "X<sub>e</sub>" which is defined as\*

とうできょうというも、ことにしていたのであるので、いたまでもないで、いいい

21 F 41

$$X_{e} = \frac{M_{o}^{a}}{\sigma} \left(\frac{T_{g}}{T_{s}}\right)^{\frac{a}{2}}$$
 (3-31)

where "a" is a constant that was determined experimentally. The term  $(T_g/T_s)^{a/2}$  in Eq. (3-31) arises from the substitution for the gas stream density  $\rho_g$  in the parameter M<sub>0</sub> of the

\*Evidence is presented in Subsection 3.1.4 that  $X_e$  is equivalent to a dimensionless Weber number.

corrected density  $\rho_g (T_g/T_s)^{\frac{1}{2}}$ . This is a technique that is frequently employed in the correlation of heat transfer data to account for the fact that the density for the gas within the boundary layer region is inversely proportional to the temperature for the gas (36)\*. It should be noted, however, that because of the low gas stream temperatures that were employed in the subject investigation, the term  $(T_g/T_s)^{a/2}$ was not crucial to the correlation of the experimental data for the entrainment parameter  $e_o$  (discussed immediately below). It is included in the definition of the entrainment group  $X_e$  primarily to be consistent with the accepted practice for accounting for variable property effects.

A lengthy trial-and-error procedure employing all of the experimental data presented herein showed that the form of the entrainment group presented in Eq. (3-31) resulted in the best correlation of the data over the complete range of parameters investigated, and that the parameter  $e_0$  could be rorrelated satisfactorily if the constant "a" in Eq. (3-31) was set equal to  $\frac{1}{2}$ . Figure 21 presents the resultant correlation of the data for a =  $\frac{1}{2}$ . Presented in the figure is  $(1-e_0)$  as a function of  $X_e$ , where  $(1-e_0)$  was determined for each experimental test from the slope of the least-squares fit curve for the experimental data of that test (e.g., as shown in Fig. 14). The nominal values for the gas stream temperature and pressure are indicated in Fig. 21, together

\*This technique was employed, in part, to derive the expression for the aforementioned correction factor  $F_T$  (see Eq. (3-18)).



.......

こう ションティー・ション あいまたい

あいまたい ひまい

I

Į

Fig. 21. Entrainment Parameter  ${f e}_{f o}$  as a Function of the Entrainment Group  ${f X}_{f e}$ 

with the four liquids that are represented by the data. A1though there is admittedly some scatter in the experimental results, some of the scatter can be attributed to the fact that  $(1-e_0)$  was determined from a least-squares fit of the "raw" experimental data for mass transfer. Indeed, because the data presented in Fig. 21 do represent such a substantial variation in both the pertinent parameters for the liquid phase and for the gas phase, it is somewhat surprising that e can be correlated so well in terms of the relatively simple parameter  $X_p$ . Perhaps most significant is the fact that  $e_{o}$  was not found to show any dependence on the liquid viscosity  $\mu_1.$  The data presented in Fig. 21 represent a variation in  $\mu_1$  from 1.7x10<sup>-4</sup> lb/ft-sec for water at a temperature of 265F to  $30 \times 10^{-4}$  lb/ft-sec for butanol at a temperature of 40F. The results of the subject investigation are thus contrary to the suggestion made by some investigators that a liquid-film Reynolds number (which is inversely proportional to  $\mu_{1}$  ) is the parameter that most characterizes the entrainment phenomenon\*.

The curve shown in Fig. 21 is given by the following expression. Thus

$$e_0 = 1.0 - \exp[-5 \times 10^{-5}(X_e - 1000)]$$
 (3-32)

1

where X<sub>e</sub> is substituted with the dimensions lb<sub>f</sub><sup>-1/2</sup>. Equation

\*Refer to the previous discussion on this point in Subsection 2.4.1.

(3-32) suggests that a critical value for  $X_e$  of 1000 lb<sub>f</sub><sup>-1/2</sup> exists, below which no entrainment of liquid occurs. The critical or limiting value for  $X_e$  of 1000 lb<sub>f</sub><sup>-1/2</sup> was determined partly from the experimental data of the present investigation, but also from the experimental data due to Steen and Wallis (37, 38). They considered in some detail the problem of determining the critical flow conditions at which entrainment from a liquid film first occurs. As a result of their experiments, it was suggested that a critical gas stream velocity could be defined by the condition that

$$\pi_{c} = 2.5 \times 10^{-4}$$
 (3-33)

where

$$r = \frac{u_{g} \mu_{g}}{\sigma} \left(\frac{\rho_{g}}{\rho_{1}}\right)$$
 (3-34)

ρ<sub>l</sub> = liquid density c - denotes critical value Recognizing that

$$\rho_g^{0.5} u_g = M_0^{0.5}$$
 (3-35)

it follows that the parameters  $X_e$  and  $\pi$  are related by the euqation

$$X_{e} = \frac{\rho_{1}^{0.5}}{\mu_{g}} \pi$$
 (3-36)

The data reported by Steen et al. were for cold air flow, with most of the data obtained for water as the liquid phase. Assuming that the temperature of both the gas stream and the liquid film was approximately 60°F, and substituting the corresponding values for  $\mu_g$  for air and  $\rho_1$  for water at that temperature, Eq. (3-26) gives the following relationship between X<sub>e</sub> and  $\pi$  for the data of Steen et al. Thus

. ....

$$X_e = 3.8 \times 10^6 \pi (1b_f^{-\frac{1}{2}})$$
 (3-37)

Substitution of the critical value for  $\pi$  that was proposed by Steen and Wallis into Eq. (3-37) results in the corresponding critical value for  $X_e$  of 950  $1b_f^{-\frac{1}{2}}$ . The critical value of 1000  $1b_f^{-\frac{1}{2}}$  for  $X_e$  in Eq. (3-32) was selected as a convenient compromise between the data of the present investigation and those for the investigation of Steen et al.

The experimental data of Steen and Wallis (37, 38) for the onset of entrainment, though obtained under much less severe flow conditions than those that characterized the present investigation, are in good agreement with the data of the subject investigation. It is important to note, however, that it was found that it was not possible to correlate the experimental data of the present investigation for  $e_0$  in terms of  $\pi$  instead of in terms of  $X_e$ . Unlike the investigation due to Steen et al., the gas stream viscosity  $\mu_g$  for the present investigation varied substantially. Although the parameter  $\pi$  does have the advantage over the parameter  $X_e$  of being dimensionless, the successful correlation of the data of the subject investigation suggests that the parameter  $X_e$  better characterizes the phenomena. If will be argued in the following subsection that although the parameter  $X_e$  is not itself a dimensionless variable, the correlation of  $e_0$  in terms of that parameter is equivalent to the correlation of  $e_0$  in terms of a dimensionless Weber number.

It is interesting on note from Fig. 21 that for the present investigation the entrainment parameter e varied from essentially zero to a maximum value of approximately The maximum value for  $e_0$  that is shown in Fig. 21 0.6. corresponds to the experimental test 117 where methanol was employed as the film coolant. For that test, the computed simple rate of mass transfer was 0.022 lb/sec-ft and the maximum mass transfer rate that was realized experimentally was 0.154 lb/sec-ft, a difference of 700 percent. Almost all of that difference (about 690 percent) was due to the entrainment phenomenon. In fact, it is important to note that with the exception of a very small number of tests (most of which employed water as the film coolant which has a relatively high value of surface tension), the contribution to the total rate of mass transfer that was due to the entrainment of liquid into the gas stream was substantially larger than the contribution that was due to evaporation.

This was true considering both the simple rate of evaporative mass transfer and the evaporative mass trunsfer that was due to the phenomenon of interfacial roughness.

## 3.1.4 The Physical Significance of the Correlation for e

The physical significance of the satisfactory correlation of e<sub>o</sub> in terms of the parameter X<sub>e</sub> is discussed in the present section. The discussion will suggest a procedur that is employed in Subsection 3.1.4 to correlate the roughness parameter r.

Since it is generally accepted that the phenomenon of entrainment is primarily the result of the shearing off of liquid droplets from disturbance waves on the liquid film surface, it is instructive to consider the forces that act on a wave that develops on the surface of the liquid film. If l characterizes the scale of those interfacial waves, then the aerodynamic drag force that is exerted by the gas stream on the wave that acts to destroy the wave is proportional to  $M_0 \frac{12}{5}$ , and the force that is due to the surface tension that acts to keep the wave intact is proportional to  $\sigma$ . The balance between these two forces should be the controlling factor in the breakdown of a disturbance wave on the surface of the liquid film. Therefore, that balance of forces should be important in determining the rate at which liquid is entrained from the liquid film into the gas stream. Thus, if a dimensionless Weber number, denoted by We, is

defined as the ratio of those two forces such that

We = 
$$\frac{M_0 l_s}{\sigma}$$
 (3-38)

it is reasonable to assume that the entrainment parameter  $e_o$  could be correlated as

$$\mathbf{e}_{\mathbf{0}} = \mathbf{e}_{\mathbf{0}}(\mathsf{W}\mathbf{e}) \tag{3-39}$$

The results of the foregoing subsection showed that

$$e_0 = e_0(X_e)$$
 (3-40)

Upon comparing the defining equations for  $X_e$  and  $W_e$ , and ignoring for the sake of argument the term  $(T_g/T_s)^{\frac{1}{4}*}$ , it is seen that the apparent contradiction between Eq. (3-39) and Eq. (3-40) could be reconciled if a dimensional constant "K" existed such that

 $K = 1_{S} M_{O}^{\frac{1}{2}} = constant$  (3-41)

There is, in fact, some physical evidence presented herein that supports the assumption that the dimensional constant K does exist. Recall from the qualitative analysis of the photographic data in Section 2.4 of this report that the

\*The maximum value of this term for the subject investigation was 1.15.

parameter  $M_0$  (and not G or  $u_g$  or any parameter for the liquid phase) was the parameter that most characterized the scale of the waves on the surface of the liquid film. Moreover, the photographs showed that the scale decreased with increasing values for  $M_0$ . Therefore, although the validity of the exponent of  $\frac{1}{2}$  in Eq. (3-41) cannot be argued on the basis of the qualitative photographic data, there is some physical evidence to support the form of the equation.

Another interesting observation can be made with reference to Eq. (3-41) that gives some insight into the phenomena. It is well documented (18, p671) that the shattering (or atomization) of liquid drops that are suspended in a gas stream can be characterized by a dimensionless Weber number, where the characteristic length  $(1_s \text{ in Eq. } (3-38))$ is chosen as the radius of the drop. If that Weber number is greater than about 6, the liquid drop will break down into smaller drops. Now, if an analogy is drawn between the breakdown of such liquid drops and the breakdown of the liquid waves on the surface of the liquid film, then a crude calculation can be made for the constant K in Eq. (3-41). Assuming that the critical value for We for the case where  $l_s$  characterizes the scale of the waves on the surface of the liquid film is still 6, and recalling that the critical value for the parameter X<sub>e</sub> below which no entrainment of liquid occurred was approximately 1000  $lb_{f}^{-\frac{1}{2}}$ , then K can be

approximated by the following calculation. Thus

$$K = \frac{We}{X_e} = \frac{We}{X_{e,c}} \approx \frac{6}{1000} = 0.006 \ 1b_f^{\frac{1}{2}}$$
(3-42)

Continuing this heuristic argument, it is interesting to use the above value for K, together with Eq. (3-41), to estimate the range of values for the scale  $l_s$  for the subject investigation. From Table G.1 of Appendix G, the maximum and the minimum experimental value for the momentum parameter  $M_o$  was  $1087 \ lb_f/ft^2$  and 20  $lb_f/ft^2$ , respectively. Thus, employing the value for K of 0.006  $lb_f^{l_2}$ , the maximum and the minimum value for the scale  $l_s$  for the subject investigation can be estimated from Eq. (3-41). Thus

 $1_{s,max} = 0.00135 \text{ ft}$ 

1

ひょうこう

きょう 二、「「「「「「」」」」」

 $l_{s,min} = 0.00018 \text{ ft}$ 

While these calculations are admittedly very approximate, the values obtained for the scale  $l_s$  are surprisingly similar to the dimensions that characterize the structure of a liquid film that is subjected to relatively large interfacial shear forces (as is the case in the present investigation). For example, Ref. (19) shows that the mean thickness of such a liquid film is of the order of  $10^{-4}$  ft. Moreover, the wave length of the small scale disturbances can be roughly approximated as 10 times the mean thickness of the liquid film (8, 5) or the order of  $10^{-3}$  ft, and the amplitude of those disturbance waves is of the same order as the thickness of the liquid film (5), or 10<sup>-4</sup> ft. The precedin, calculations, therefore, add some additional support to the assumption that the dimensional constant K exists. Moreover, they also suggest that the breakdown of the interfacial waves and the resultant entrainment of liquid into the gas stream may be somewhat analogous to the atomization of liquid drops that are suspended in a gas stream.

To summarize briefly, the above discussion suggests that (a) the dimensional parameter  $X_e$  and the dimensionless Weber number We are physically equivalent parameters, and (b) a constant K =  $l_s M_0^{\frac{1}{2}}$  exists such that the characteristics (really, the gross characteristics) of the interfacial film structure can be characterized in terms of the flow parameters for the gas stream. The second of these two points, which follows directly from the first, is most pertinent to the correlation of the roughness parameter r that is presented in the following section.

#### 3.1.5 The Correlation of the Roughness Parameter r

1997 - 1997 - 1997 - 19

The present section is concerned with the correlation of the roughness parameter r that was defined by Eqs. (3-4) and (3-6). The primary assumption that was employed to develop the correlation was that, at least to a first approximation, the degree to which the effective roughness of the surface of the liquid film intensifies the turbulent transfer of energy from the hot gas stream to the surface of the liquid film is primarily a function of the scale of the small disturbance waves that develop on the surface of the film. That is, it was assumed that the roughness parameter r could be correlated in the form

$$\mathbf{r} = \mathbf{r}(\mathbf{l}_{s}) \tag{3-43}$$

Unfortunately, the utility of Eq. (3-43) is limited because of the problems inherent in trying to measure the scale  $l_s$ . Indeed, there is considerable uncertainty as to what physical dimension for the waves on the surface of the liquid film corresponds to the length  $l_s$ . However, an argument was presented in the foregoing subsection that the scale  $l_s$  could be approximately related to the momentum parameter  $M_0$  by the following expression. Thus

$$l_{s} = \frac{K}{M_{o}^{\frac{1}{2}}}$$
 (3-44)

Therefore, if a roughness group " $X_r$ " is defined by

1:

5. G

Ē

$$X_{r} = M_{0}^{\frac{1}{2}} \left(\frac{T_{g}}{T_{s}}\right)$$
 (3-45)\*

Equation (3-43) can be rewritten in the following equivalent form. Thus

\*Again, as for Eq. (3-3]) in Subsection 3.1.3, the term  $(T_g/T_s)^{\frac{1}{4}}$  is an empirical correction for the dependence of the density of the gas in the boundary layer on temperature.

(3-46)

R

I

$$r = r(X_r)$$

Equation (3-46) is the basic expression that was employed in the subject investigation to correlate the roughness parameter r. The value of the roughness parameter r was estimated for each experimental test as follows:

(a) The average value for the simple rate of mass transfer, m's, was calculated following the procedure outlined in Subsection 3.1.2.

(b) The value of the entrainment parameter  $e_0$  was determined from the average value of the entrainment group  $X_e$  for that test, together with the analytical expression given by Eq. (3-32).

(c) The value of the intercept "A" (see Fig. 19) was estimated by determining which curve having the slope (1-e<sub>0</sub>)
 resulted in the best "eyeball" fit of the experimental data.
 (d) The roughness parameter r was then determined from the expression

 $r = \frac{A}{m_s^{T}} - 1.0$  (3-47)

Figure 22 presents the roughness parameter r-determined by the aforementioned procedure-as a function of the parameter  $X_r$ . The various conditions for the gas stream pressure and temperature are listed in the figure. Although the data



Į

r , roughness parameter

in Fig. 22 do exhibit some scatter, much of the scatter can be attributed to the fact that the value for the intercept A was very sensitive to relatively small errors in the measurement of the net rate of mass transfer from the liquid film to the gas stream. Furthermore, any error in determining the value of the intercept A was compounded in the calculation of the roughness parameter r. To illustrate, assume that the true value for the parameter  $A/m^{\,\prime}_S$  is 1.2 and that the experimentally determined value is 1.4. From Eq. (3-47), the corresponding values for the roughness parameter r are 0.2 and 0.4, respectively. Thus, for this example, an error of approximately 17 percent in the intercept A results in an error of 5. percent in the evaluation of r. In addition to that problem, some of the scatter of the data in Fig. 22 is probably due to the fact that the rather complex phenomenon of interfacial roughness cannot be characterized completely in terms of the relatively simple parameter  $X_r$ . However, considering the wide range of values for the liquid and the gas flow parameters that is represented by the data in Fig. 22, the correlation of r in terms of the group  $X_r$  would appear to have some physical justification.

The solid curve that is shown in Fig. 22 is given by the following analytical expression. Thus

 $= \frac{3.0}{\chi_0.8}$ 

106

(3-48)

a state of

Ś

}{

where  $X_r$  has dimensions of  $lb_f^{\frac{1}{2}}/ft$ . While the experimental data in Fig. 22 are in general agreement with this analytical expression, the final form of Eq. (3-48) was arrived at by considering the following three points, which are listed in their order of importance:

(a) The correlation of all of the experimental data of the present investigation, including those for which the gas stream was accelerated over the wetted test plate;
(b) The correlation of the mass transfer data that were reported by Kinney, Abramson, and Sloop (19) (these data are analyzed in Section 4.3); and

(c) The correlation of the experimental data for mass transfer that are reported by Emmons (20) (these data are analyzed in Section 4.4).

F

It is important to note that Eq. (3-48) and the data in Fig. 22 suggest that the interfacial roughness can contribute substantially to the convective transport of energy from the hot gas stream to the surface of the liquid film. Indeed, Fig. 22 suggests that the simple mass transfer rate  $m'_s$  could be increased by 100 percent (r = 1), or more, due to the interfacial roughness. In comparison, for the case of heat transfer to a dry rough wall, Fig. 7-56 of Ref. (39) shows that the value for r can be as great as 2.5. It would seem, therefore, that a value of r of 1.0 for the problem of liquid-film cooling is not unreasonable. It should be emphasized, however, that in the analysis of the experimental data of the present investigation and in the analysis of the data of the primary references (19) and (20), that the contribution to the net mass transfer rate that was due to the phenomenon of interfacial roughness was, with very few exceptions, substantially less than the contribution that was due to the entrainment phenomenon. Frequently, the mass transfer due to entrainment was 10 to 20 times the evaporative mass transfer that was attributed to the effective roughness of the surface of the liquid film.

# 3.1.6 The comparison of the Predicted and the Measured Rates of Mass Transfer

A comparison is presented in this subsection of the rates of mass transfer from the liquid film to a constant velocity gas stream that were predicted from the correlation developed herein to those that were determined experimentally. The comparison is presented for both the datum points that were obtained for the hot-flow experiments (approximately 500 datum points) and those that were obtained for the coldflow experiments (76 datum points). In addition, to emphasize the utility of the present theory, the experimental data are also compared with the results that were predicted from the simple theory of mass transfer. The latter comparison serves to illustrate graphically the general inadequacy of the simple theory for the subject problem.

The primary equation that was employed to calculate the rates of mass transfer from the liquid film to the gas stream is repeated here for convenience. Thus

$$m' = (1 + r) m'_{s} + e_{0} m_{1}$$
 (3-6)

The simple mass transfer  $m'_{s}$  was calculated following the developments of Subsection 3.1.2. The roughness parameter r was calculated from Eq. (3-48) and the entrainment parameter  $e_{0}$  was determined from Eq. (3-32). The surface tension  $\sigma$  that appears in the entrainment group  $X_{e}$  in Eq. (3-32) was based on the average of the temperature of the liquid at the point of injection and at the point of withdrawal\*. All of the numerical computations were performed on an IBH 7094 digital computer.

Figures 23 through 26 summarize graphically the results of the aforementioned calculations for those experimental data where the gas stream was heated. Each figure represents one of the four liquids that was investigated (Fig. 23: methanol; Fig. 24: butanol; Fig. 25: water; Fig. 26: RP-1). The nominal value, for the gas stream pressure and the gas stream temperature for each set of experimental data are indicated in the figures. Part (a) of each figure is a plot of the simple mass transfer rate,  $m'_s$ , that was predicted from the theory outlined in Subsection 3.1.2 versus the measured rate of mass transfer. Part (b) of each figure compares the measured rate of mass transfer with the rate of mass transfer that was predicted from

!!

\*See Table G.1 of Appendix G for a tabulation of the average inlet and outlet liquid temperatures for each experimental 'test.









and the second

ĺ

]

]

]

7

Ŋ

3

Л

APPENDER CONT

2.0

8.0

0.02

o

matinal flee permeters T<sub>g</sub>(f) p(mita)

0.12

-L 7.

0.16

ŧ

٤

4: \* :

.....

r r 8

3 <u>3</u> 3

0.10

8.9

() \* bearaus sympthy ()phare-is)

9.16

9.14

0.12

0.10

9.0

8.0

0.0

20.0

•

n' , mesured (16/356-ft)



Description of the second second second second second section of the second second second second second second

K.





ļ

}

Į

]

]

8



112

°°°

1 14 10

The second second

0.07

0°.0

0.05

0.0







(b) Present Theory

Constant Velocity Comparison of Simple Theory and Present Theory With Data for Mass Transfer. Constant Velocit RP-1 Film Coolant Hot Gas Stream. F1g. 26.

Eq. (3-6) for the present theory. Each figure graphically demonstrates that, in general, the simple theory for mass transfer is a poor approximation to the complex phenomena that characterize liquid-film cooling. The maximum difference between the experimental result and the result that was predicted from the simple theory is shown in Fig. 23(a) for methanol as the film coolant. For that figure, the maximum value for the measured mass transfer rate is 0.154 lb/sec-ft and the corresponding simple rate of mass transfer is 0.022 lb/sec-ft, a difference of 700 percent. The present theory, however, accounts for the existence of the interfacial phenomena, and the resultant agreement between the predicted and the measured mass transfer rates that is shown in part (b) of each figure is quite satisfactory. The maximum difference between the result that is predicted from the present theory and the experimental result is shown in Fig. 26(b) for RP-1 as the film coolant. For a measured rate of mass transfer of 0.0944 lb/sec-ft in that figure, the corresponding mass transfer rate that is predicted by the present theory is 0.0765 lb/sec-ft, a difference of about 20 percent.

Figure 27 presents the results of the computations that were performed for the case where the gas stream was not heated. Presented in the figure is a plot of the mass transfer rate that was predicted by the present theory versus the measured rate of mass transfer. The experimental data in the figure are for one nominal gas stream pressure of 75 psia and for the liquids methanol, butanol,



ſ

E

Fig. 27. Comparison of Present Theory With Data for Mass Transfer. Constant Velocity Cold Gas Stream

ç

- かっぱい、そうふがいってん みってい 教室にす うなをとれならず。 産業産業が多少産業業

and water\*. It was assumed in the computations that the rate of simple mass transfer, m's, was essentially zero because of the low gas stream temperatures. Thus, the realized rates of mass transfer were assumed to be due to entrainment only. The maximum value that was computed for the entrainment parameter e, was 0.32, as compared to a maximum value of 0.63 for the hot-flow tests. The agreement between the measured and the predicted rates of mass transfer is quite satisfactory, particularly in view of the fact that the pertinent physical properties for both the liquid and the gas phases differed so significantly between the case where the gas stream was heated and the case where it was not. For example, the viscosity for the liquid varied by as much as a factor of 15:1, the surface tension for the liquid by as much as a factor of 1.5:1, and the viscosity for the gas stream by as much as a factor of 2:1 between the hot and cold gas stream flow conditions. It would appear, therefore, that the relatively simple theory that is proposed herein for determining the rate of mass transfer from a thin liquid film to a high velocity gas stream can be applied quite The utility of the present theory is demonstrated generally. further in Section 4 of this report where the correlation of

\*The number of datum points that could be obtained with water as the film coolant was limited because the gas stream was generally cooled sufficiently by the expansion from the storage tank pressure of 1000 to 2500 psia to the test section pressure of 75 psia that a portion of the water that was injected onto the test plate froze before it reached the capture slot. Moreover, no experimental tests were performed with RP-1 as the liquid phase because of the limited supply of this liquid.

the experimental data for liquid-film cooling of the primary references (19, 20, 24) is considered. The gas stream flow conditions that characterized those investigations will be shown to be, in most instances, much more severe than those that characterized the present investigation.

The analyses of the experimental data of the aforementioned primary references requires first the consideration of the problem of how the basic correlation for mass transfer can be extended to the more general case where the film cooled length 1 is arbitrary. That is, the case where 1 does not necessarily equal  $l_0 = 10$  inches. The following section presents an analysis of that problem. The result of the analysis will also suggest a method for correlating the experimental data for mass transfer that were presented in Section 2.6 of the present report for the case where the gas stream was accelerated over the film cooled length  $l_0$ . The correlation of those data is then discussed in Section 3.3.

## 3.2 <u>The Modification of the Basic Correlation</u> <u>for Mass Transfer to Account</u> <u>for an Arbitrary Film Cooled Length</u>

The correlation that was developed for the entrainment parameter  $e_0$  in Subsection 3.1.3 is necessarily dependent on  $l_0$ , the length of the film cooled test section for the subject investigation. That is, if the film cooled length  $l_0$ had been, say, greater than the 10-in. length that was employed, then, for the same flow conditions, there would have been more entrainment of liquid into the gas stream because

of the prolonged exposure of the surface of the liquid film to the gas stream shear forces. In order to be able to predict the mass transfer that occurs for an arbitrary film cooled length 1, the basic correlation for  $e_0$  has to be modified. A rather simple model and analysis is developed in Subsection 3.2.1 to solve that problem. Subsection 3.2.2 then suggests an empirical approximation to the result of that analysis that is generally more convenient to employ.

#### 3.2.1 Model and Analysis

Figure 28 presents the simple model that is employed in the analysis. The pertinent notation for the figure is defined as follows:

X '	2	$x - x_1$
<sup>m</sup> 1	=	the rate of liquid injection at x' = o
1	E	the liquid-film cooled length
۱ <sub>0</sub>	=	the length of each film element*
n	2	1/1 <sub>0</sub> (assumed to be an integer)
m <b>k</b>	r	the net rate of mass transfer from the k <u>th</u> element of the liquid film, k = 1, 2, 3,n
<sup>m</sup> 's,k	E	the net rate of simple mass transfer for the k <u>th</u> film element, k = 1, 2, 3,n

5

ing a

\*For the purposes of the present analysis, it is not necessary to specify that 0 = 10 inches. However to "tilize the result of the analysis to perform any numerical calculations, it will be necessary to make that specification.



ł

(

[

[]

0

{

ì

0

R

F1g. 28.

اليوايد ومنتقلة الألاسة لده

ļ

Sarah.

1

Ŋ

P

Ē

8

The basic assumptions employed in the analysis are as follows:

- (a) The flow parameters for the gas stream are independent of x<sup>1</sup>.
- (b) The roughness parameter r is independent of x'.
- (c) The local rate of simple mass transfer varies with the coordinate x' as

 $m_{s}^{"} = c x^{p}$  (3-50)

where c and p are appropriate constants.

- (d) The primary equation (3-6) applies to each of the elements of the liquid film.
- (e) The liquid film terminates when l = n l<sub>0</sub>, where n is an integer.

Thus, for the first element of the liquid film, the net rate of mass transfer is given by

$$m'_{1} = (1 + r) m'_{s,1} + e_{0}(m_{1})$$
 (3-51)

Similarily, the mass transfer for the second film element is given by

 $m_2' = (1 + r) m_{s,2}' + e_0(m_1 - m_1')$  (3-52)

In general, for the kth element of the liquid film, the net

rate of mass transfer  $m'_k$  is expressed as

l

(

E

ſ

ß

$$m'_{k} = (1 + r) m'_{s,k} + e_{o}(m_{1} - m'_{1} - m'_{2} - \dots - m'_{k-1})$$
 (3-53)

Substitution of Eq. (3-51) into Eq. (3-52) gives the result

$$m'_{2} = (1 + r) m'_{s,2} + e_0[m_1 - (1 + r) m'_{s,1} - e_0m_1] (3-54)$$

~~ の治しの、夢、読を生ま 夢らのます

Substitution of Eqs. (3-51) and (3-54) into Eq. (3-53) with k = 3 yields

$$m_{3}^{i} = (1 + r)m_{s,3}^{i} e_{0} \{m_{1} - (1 + r) m_{s,1}^{i} - e_{0}m_{1} - (1 + r) m_{s,2}^{i} - e_{0} [m_{1} - (1 + r) m_{s,1}^{i} - e_{0}m_{1}]\}$$

$$(3-55)$$

This substitution procedure can be continued in a stepwise fashion so that for the general  $k\underline{th}$  term  $m'_k$ , the terms  $m'_1$ ,  $m'_2$ ,  $\dots$   $m'_{k-1}$  can be eliminated from the resultant expression. Moreover, after considerable algebra and a fortuitous combining of terms, it is possible to show that

$$\sum_{k=1}^{n} m_{k}^{i} = m_{1} - m_{1}(1-e_{0})^{n} + (1+r) \sum_{k=1}^{n} m_{s,k}^{i} (1-e_{0})^{n-k}$$
(3-56)

But, if it is assumed that the liquid in the film is consumed at a point where  $l = n l_0$ , then it follows from the conservation of the liquid that

$$\sum_{k=1}^{n} m_{k}' = m_{1}$$
 (3-57)

Equation (3-56), therefore, can be rewritten in the form

$$m_1 = (1 + r) \sum_{k=1}^{n} m'_{s,k} (1 - e_0)^{-k}$$
 (3-58)

From assumption (c) of the model it follows that the rate of simple mass transfer from the k<u>th</u> film element is given by

$$m_{s,k}^{k} = \int_{0}^{k} m_{s}^{u} dx' = c \frac{\{(k_{0})^{p+1} - [(k-1)_{0}]^{p+1}\}}{(p+1)}$$
(3-59)

Moreover, the total rate of simple mass transfer from the liquid film is equal to

$$m'_{s} = \int_{0}^{n} m''_{s} dx' = c \frac{(nl_{o})^{p+1}}{(p+1)}$$
(3-60)

R

Ì

K

Ŋ

Finally, substituting Eq. (3-59) into Eq. (3-58) and dividing through both sides of the resultant expression by Eq. (3-60)yields the following non-dimensional equation. Thus

$$\frac{m_1}{m_s^{\prime}} = (1 + r) \sum_{k=1}^{n} \left[ \left(\frac{k}{n}\right)^{p+1} - \left(\frac{k-1}{n}\right)^{p+1} \right] (1 - e_0)^{-k} \quad (3 - 61)^*$$

Equation (3-61) is the desired analytical result. For given gas stream and liquid flow conditions (and, therefore, given values for the parameters p, r and  $e_0$ ), Eq. (3-61) can be utilized to calculate the net rate of mass transfer from the liquid film into a constant velocity gas stream if the film cooled length 1 is equal to n  $l_0$ , where n is an integer. The obvious disadvantage of Eq. (3-61) is the fact that it can be utilized for only those special cases where  $l = n l_0$ . Therefore, it is convenient to replace that analytical result by an approximate empirical expression that is a continuous function of 1, where 1 can be either greater than or less than  $l_0$ . Such an empirical expression is developed in the following subsection.

ſ

\*Note that for the special case where the interfacial phenomena are not important, i.e.,  $r = e_0 = 0$ , then the RHS of Eq. (3-61) reduces to unity as required.

### 3.2.2 An Empirical Approximation to the Analytical Result

An empirical approximation to the analytical equation (3-61) can be derived as follows. First note that for the special case where  $l = l_0$ , Eq. (3-61) reduces to the simple expression

$$\frac{m_1}{m_s'} = (1 + r) (1 - e_0)^{-1}$$
(3-62)

1

<u>}</u>

Now assume that Eq. (3-62) can be modified as follows for the case where the film cooled length is arbitrary. Thus

$$\frac{m_1}{m_s^{\prime}} = (1 + r) (1 - e_0)^{-(1/1_0)^{\rm b}}$$
(3-63)

The parameter b in Eq. (3-63) is determined by first specifying a value for the parameter p that appears in the analytical expression (3-61), and then "matching" the empirical expression (3-63) and the analytical expression (3-61) at those points where  $(1/1_0) =$  an integer.

To illustrate the aforementioned procedure for the evaluation of the parameter b, consider the classical case of the turbulent transport of heat and mass from a flat plate where the parameter p in Eq. (3-50) is equal to -1/5. Figure 29 presents a plot of the quantity  $(m_1/m_s^2)/(1 + r)$  as a function of the ratio  $(1/l_0)$  with the entrainment parameter  $e_0$  as a parameter. The solid curves shown in Fig. 29



attribution of the state of the state and the state of th



 $m_{s}^{\mu} = c x^{-1/5}$ 

ſ

I

Kirk and

E

U

uraat.

were calculated from Eq. (3-63) with a value for b of 0.6, and the circles are the "match" points that were calculated from the analytical expression (3-61).

Figure 30 presents another illustrative example for the case where the parameter p in Eq. (3-50) is equal to 1.0; that is, the case where the rate of simple mass transfer increases linearly with the coordinate x'\*. The solid curves were calculated from Eq. (3-63) where the value of 0.9 for the parameter b was employed, and the "match" points were calculated from Eq. (3-61).

Figures 29 and 30 show that the analytical result of Subsection 3.2.1 can be approximated with sufficient accuracy by the relatively simple empirical formula of the present section. Moreover, these figures show that a substantial change in the parameter p in Eq. (3-61) requires a relatively small change in the parameter b in the empirical expression (3-63) in order to obtain an adequate matching of the analytical and empirical results.

The significance of the foregoing developments can be further illustrated by writing Eq. (3-63) in a more explicit form. Substituting the analytical expression (3-48) for the roughness parameter r and the analytical expression (3-32) for the entrainment parameter  $e_0$ , Eq. (3-63) can be

ł

\*This example is presented solely for the purposes of illustration. It may or may not have any physical significance.


the formation of the second second

Fig. 30. Matching of Empirical Equation to Analytical Result  $m''_s = c x$ 

. .

ſ

Ĩ

I

I

I

E

I

written in the form

$$\frac{m_1}{m_s^*} = (1 + \frac{3.0}{X_r^{0.8}}) \exp \left[5 \times 10^{-5} (X_e - 1000) (1/1_o)^b\right]$$
(3-64)\*

Thus, if the value for the parameter b is approximated from the aforementioned "matching" procedure, Eq. (3-64) gives an explicit expression that can be employed to calculate the net rate of mass transfer from a liquid film of arbitrary length 1 to a constant velocity gas stream.

The validity of the foregoing analysis (and, therefore, Eq. (3-64)) will be investigated in Section 4 of the present report. Considered in that section is the correlation of the experimental data for mass transfer that were reported in the primary references (19, 20). The range of values for the film cooled length 1 that is represented by those data is 3 inches to 36 inches long.

Equation (3-64) also serves to suggest a method for the correlation of the experimental data for mass cransfer that were obtained for the case where the gas stream was accelerated

\*So that the reader is not confused, the substitution of Eq. (3-32) for the entrainment parameter  $e_0$  requires that the length  $l_0$  in Eq. (3-64) be taken as 10 inches. Moreover, the numerical constants in Eq. (3-64) require that the roughness group  $X_r$  be substituted with the units  $(lb_f^{\frac{1}{2}}/ft)$  and the entrainment group  $X_e$  be substituted with the units  $(lb_f^{-\frac{1}{2}})$ .

over the film cooled length 1<sub>0</sub>. Those data were presented in Section 2.6 of the present report. The correlation of the subject data is considered in the following section.

### 3.3 <u>Correlation of the Data for Mass Transfer</u> <u>for an Accelerating Gas Stream</u>

The purpose of the present section is to correlate the experimental data for mass transfer that were obtained for the case where the gas stream was accelerated over the length of the film cooled test plate. These data were shown graphically in Fig. 17 of Section 2.6. Subsection 3.3.1, which follows, shows how the simple rate of mass transfer was computed for the subject case. Subsection 3.3.2 then demonstrates how that simple rate of mass transfer was employed, together with a logical extension of the developments that were presented in Sections 3.1 and 3.2, to correlate the experimental data for an accelerating gas stream. The comparison between the experimental and the predicted net mass transfer rates is presented in Subsection 3.3.3.

้าประกรรษณ์ ส่วนสะครรรณ (การระบุกรรณ) การระบุกรรณ รู้สูงสามารรณ์ การระบุกรรณ

3.3.1 The Evaluation of the Simple Rate of Mass Transfer for an Accelerating Gas Flow

Kays (36) suggests that the following expression can be employed to evaluate the local rate of simple mass transfer for the case where the gas stream is accelerating. Thus

$$m_{s}^{"} = G B_{h}F_{T}F_{C}F_{m}^{"}F_{Pr} [0.0295 Re_{x}^{-.08} Re_{x}^{+-0.12}]$$
  
\* (3-65)\*

where

$$Re_{x}^{\prime} = \frac{1}{\mu_{g}} \int_{0}^{x} G dx$$
 (3-66)

$$\operatorname{Re}_{x_{t}}^{\prime} = \frac{1}{\mu_{g}} \int_{x_{t}}^{x} G dx$$
 (3-67)

The "F" correction factors in Eq. (3-65) are given by Eqs. (3-18) - (3-21) of Subsection 3.1.2. For the purposes of the present calculations, the energy balance parameter  $\phi$ that enters into the definition of the parameter  $B_h^{**}$  was approximated from Eq. (3-13) of Subsection 3.1.2. Moreover, because of the limited Mach number range that was investigated, the variation in the gas stream temperature,  $T_g$ , over the film cooled length  $l_o$  was so small that the average of

\* Kays (36) introduces an additional empirical correction factor into Eq. (3-64) to correct for acceleration effects that are not predicted analytically. However, preliminary calculation of that correction factor showed that it was of the order of 0.98 for the range of flow conditions for the subject investigation. Therefore, that correction factor was not considered in the present calculations for  $m_s^{*}$ .

h

\*\*See Eq. (3-22).

the values of  $T_{q}$  at  $x = x_{1}$  and at  $x = x_{2}$  was employed to calculate the parameters  $\mu_{\textbf{q}},\ B_{\textbf{h}},\ \textbf{and}\ F_{\textbf{T}}$  in Eq. (3-65). To complete the solution for  $m_s^{"}$ , it is necessary to determine the local values for the static pressure p, the gas stream Mach number  ${\rm M}_{\rm a}$  , the gas stream velocity  ${\rm u}_{\rm g}$  , and the mass velocity for the gas stream G. The procedure employed to evaluate those parameters is outlined as follows. The static pressure was measured at the location x = 20 inches in the approach section and at 5 axial locations (equispaced at 2 inches) in the test section by wall taps that were located in the roof of the experimental tunnel. The stagnation pressure for the gas flow was calculated from the static pressure that was measured in the approach section, together with the value that was computed for the gas stream velocity at the entrance to the test section\*. The local value for the Mach number within the experimental test section was computed by forming the ratio of the measured static pressure and the calculated stagnation pressure, and then employing the following relationship for isentropic flow with a specific heat ratio  $\gamma = 1.385$ . Thus

3

3

ŝ

7

$$\frac{P}{P^{0}} = \{1 + \frac{\gamma - 1}{2} N_{a}^{2}\}^{\frac{\gamma}{\gamma - 1}}$$
(3-68)

\*See Appendix F.2 for the procedure that was employed to calculate the gas stream velocity at the entrance to the test section.

The complete profile for the Mach number throughout the test section was obtained by fitting a third-order polynominal to the local values for the Mach number that were computed from the experimental measurements for the static pressure. The resultant form or the approximate expression for the Mach number as a function of x', the distance from the point of liquid injection, was

$$M_{2} = 0.145 + 0.02x' - 0.0014x'^{2} + 0.00028x'^{3} \quad (3-69)$$

The local values for the gas stream velocity,  $u_g$ , were computed from Eq. (3-69) and the equation for the acoustic speed in a perfect gas:

$$u_g = M_a (\gamma R_g T_g)^{0.5}$$
 (3-70)

Finally, the local values for the gas mass velocity, G, were calculated from the defining equation

 $G = \rho_{q} u_{q} \qquad (2-1)$ 

where the gas stream density,  $\rho_{g}$ , was evaluated from the equation of state for a perfect gas:

T<sub>g</sub>

, {

(3-71)

Figure 31 shows graphically the values that were computed from the foregoing procedure for the parameters  $p/p^0$ ,  $M_a$ , and  $u_g$  as a function of x'. The solid curves that are presented in the figure are based on the analytical expression (3-69) for the Mach number. The circles presented in Fig. 31 represent the static-stagnation pressure ratios that were realized experimentally. Each circle presented in the figure represents the approximate range of values for  $(p/p^0)$ that was obtained at that particular location for the entire spectrum of gas flow conditions that was investigated.

The foregoing equations were programmed for an IBM 7094 digital computer and the integral expressions for the terms  $Re'_x$  and  $Re'_{xt}$  were evaluated numerically.

### 3.3.2 The Model and Analysis

Figure 32 illustrates the model that was employed in the analysis of the subject data. The figure shows that the iiquid film was subdivided into a number of incremental elements of length  $\delta x$ . The pertinent notation that applies to the general kth film element is as follows:

- (a) x<sub>k-1</sub> = (k-1)δx = the distance from the point of liquid injection to the beginning of the k<u>th</u> film element, k=1,2,3,...n.
- (b) u<sub>k</sub>,p<sub>k</sub>,...

l

- the value for the indicated flow parameter for the kth film element evaluated at x<sub>k</sub> - ½0x, k=1,2,3,...n.
- (c)  $X_{e,k}$ ,  $X_{r,k}$  = the value for the indicated dimensional group for the kth film element evaluated at  $x_k \frac{1}{2}\delta x$ , k=1,2,3,...n.



134

I

J

]

]

-

[]

 $\left[ \right]$ 

8



{

[]

1

Super se

a chanalan a mag d

fre print

こう ちょうちょうちょう しんていないないできる しましたいちょうちょうちょう

(d)	, <sup>m</sup> k	=	the rate of liquid flow into the kth film element, k=1,2,3,n.
(e)	<sup>m</sup> k	2	the net rate of mass transfer from the k <u>th</u> film element, k=1,2,3,n.
(f)	<sup>m</sup> s,k		the rate of simple mass transfer from the kth film element, k=1,2,3,n.

The primary assumption that was employed in the analysis is that the net rate of mass transfer from the k<u>th</u> film element,  $m'_k$ , is given by the expression

$$m'_{k} = (1 + r_{k}) m'_{s,k} + e_{k} m_{k}$$
 (3-72)\*

where

$$r_{k} = \frac{3.0}{\chi_{r,k}^{0.8}}$$
(3-73)

and

$$e_k = 1.0 - exp[-5 \times 10^{-5} (X_{e,k} - 1000) (\frac{\delta x}{l_0})^d]$$
  
(3-74)\*\*

The value for the parameter d in Eq. (3-74) was determined from the experimental data.

The procedure that was employed to calculate the net rate of mass transfer from the liquid film to the accelerating gas stream for each datum point can be outlined as

×	Note	that	Eq.	(3-72)	is	analogous	to	Eq.	(3-6).	
* 7	*Note Subse	that ection	Eq. n 3.2	(3-74)	is	motivated	by	the	developments	of

A

Π

follows.

(a) The pertinent gas stream flow parameters were calculated for each location  $x_k$  following the procedure outlined in Subsection 3.3.1.

(b) The entrainment group,  $X_{e,k}$ , and the roughness group,  $X_{r,k}$ , were calculated for each location  $x_k$  from the flow parameters that were determined in step (a).

(c) The rate of simple mass transfer,  $m'_{s,k}$ , for each film element was calculated from the equation

$$\mathbf{m}'_{\mathbf{s},\mathbf{k}} = \mathbf{m}''_{\mathbf{s},\mathbf{k}} \quad \delta \mathbf{x} \quad (3-75)$$

where  $m_{s,k}^{"}$  was computed at the location  $x_k - \frac{1}{2}\delta x$  from the procedure that was outlined in Subsection 3.3.1. (d) A value for the parameter d in Eq. (3-74) was selected. (e) The net rate of mass transfer for the first film element (i.e., k = 1) was calculated from

$$m'_1 = (1 + r_1) m'_{s,1} + e_1 m_1$$
 (3-76)

where  $m_1$  was the experimentally measured rate of liquid injection.

(f) The net rate of mass transfer from the second film element (i.e., k = 2) was calculated from

$$m_2' = (1 + r_2) m_{s,2}' + e_2 m_2$$
 (3-77)

where

$$m_2 = m_1 - m_1'$$
 (3-78)

1.38

(g) Steps (e) and (f) were continued in a stepwise fashion so that for the general k<u>th</u> film element the net rate of mass transfer was calculated from

$$m'_{k} = (1 + r_{k}) m'_{s,k} + e_{k} m_{k}$$
 (3-79)

where

$$m_k = m_1 - m_1' - m_2' - \dots - m_{k-1}'$$
 (3-80)

(h) When the net rate of mass transfer for each film element had been calculated, the net rate of mass transfer from the entire liquid film was calculated from

$$m' = \sum_{k=1}^{n} m_{k}'$$
 (3-81)

where

$$n = \frac{1}{\delta x}$$
(3-82)

(i) The result of step (h) was compared with the experimental result for m' ( =  $n_1 - m_2$ ). If those two results

were not in reasonable agreement, then a new value for the parameter d was selected and steps (e) through (h) were repeated until a satisfactory agreement of those results was obtained.

The aforementioned procedure was employed to analyze each of the 50 datum points that was shown in Fig. 17 of Section 2.6. The computations were performed for values of  $\delta x$  of 0.1, 0.2, and 0.4 inches to see if that parameter significantly influenced the results. The calculations showed that the simple value of 1.0 for the parameter p in Eq. (3-74) resulted in a good correlation of all of the datum points for each of the aforementioned values for the parameter  $\delta x$ .

### 3.3.3 The Comparison of the Predicted and the Measured Rates of Mass Transfer

Figure 33 presents the comparison of the experimental data for the net rate of mass transfer from the liquid film to the accelerating gas stream with the results that were predicted from both (a) the simple theory for mass transfer that was outlined in Subsection 3.3.1, and (b) the present theory for mass transfer that was developed in Subsection 3.3.2. The stagnation pressure,  $p^0$ , and the film coolant that is represented by each set of experimental data is noted in the figure.

Figure 33(a) again graphically illustrates that the simple theory for mass transfer is, in general, a very poor





140

-

the office of

11 14

I

]

approximation to the complex phenomena that characterize liquid-film cooling. The maximum difference between the measured and the analytical result that is shown in the figure occurs for methanol as the liquid phase at a measured rate of mass transfer of 0.1668 lb/sec-ft and a corresponding predicted value of 0.0309 lb/sec-ft, a difference of more than 500 percent.

Figure 33(b) shows that the present theory, with a value of 1.0 for the parameter d in Eq. (3-74), resulted in a satisfactory correlation of the experimental data. The maximum difference between the measured and the predicted result occurred for water as the film coolant at an experimental value for the net mass transfer rate of 0.0462 lb/sec-ft and a corresponding predicted value of 0.0620 lb/sec-ft, a difference of less than 40 percent.

The utility of the present analysis is investigated in further detail in Section 4. Considered in that section is the correlation of the data that were reported in the primary reference (24) for the liquid-film cooling of a Mach number 1.92, axi-symmetric nozzle. Also presented in that section is an illustrative example of the coolant requirements that are predicted by the present theory to cool an exhaust nozzle for which the gas stream has a stagnation temperature of 4000R and a stagnation pressure of 500 psia. These stagnation conditions were chosen as being typical of those that characterize the gas flow in the exhaust nozzle

of a high-energy propulsion device.

### 3.4 <u>Correlation of the Experimental Data</u> for the Maximum Liquid Temperature

The correlation of the experimental data that were reported in Section 2.7 for the maximum liquid temperature,  $T_{1,m}$ , is considered in this section. Subsection 3.4.1 presents a simple theory for calculating that liquid temperature. The comparison of the theoretical and the experimental results is shown in Subsection 3.4.2.

### 3.4.1 A Method for Predicting the Maximum Liquid Film Temperature\*

The maximum liquid-film temperature,  $T_{l,m}$ , can be related to the bulk flow parameters for the gas stream by specifying (a) the appropriate energy balance across the gas-liquid interface, and (b) an analogy between the rates of convective\*\* heat and mass transfer. The discussion of Section 2.7 showed that the maximum liquid temperature is obtained when all of the energy transfer to the liquid film results in vaporization of liquid in the film, with none of the energy transfer resulting in an increase in the sensible enthalpy for the liquid in the film. It follows from the discussion of the interfacial energy balance in Subsection 3.1.2 that the maximum liquid temperature is obtained when

\* This method is essentially the same as that previously presented by the author in Refs. (25) and (26).
\*\*Mass transfer due to entrainment is not considered.

the energy balance parameter  $\boldsymbol{\varphi}$  at the surface of the liquid film reduces to

$$\phi = H_{V,m} \tag{3-83}$$

where  $H_{v,m}$  is the heat of vaporization for the liquid evaluated at the maximum liquid temperature  $T_{l,m}$ .

The simplest analogy between convective heat and mass transfer is that given by (36)

$$St_s = St'_s$$
 (3-84)

where  $St'_s$  is the Stanton number for simple mass transfer defined by

$$St'_{s} = \frac{m''_{s}}{G} - \frac{p_{v,m}}{p_{v,m}} \frac{M_{g}}{M_{v}}$$
 (3-85)\*

The term  $p_{v,m}$  in Eq. (3-85) is the partial pressure for the vapor at the surface of the liquid film evaluated at the temperature  $T_{1,m}$ .

It is important to note that this analogy, although very simple, should be approximately correct for quite general flow conditions, including the case of accelerating

<sup>\*</sup>Equation (3-85) can also be written in terms of the con-'centration of the vapor at the surface of the liquid film, C<sub>s</sub>. Thus



(3-86)

gas flow and the case where considerable entrainment of liquid into the gus stream is occurring. In other words, the acceleration of the gas stream and/or the entrainment of liquid should have a similar influence on the development of both the thermal and the vapor-concentration boundary layers over the surface of the liquid film.

Combining Eq. (3-15) with Eqs. (3-83) - (3-85) yields

$$\frac{p - p_{v,s}}{p_{v,s}} \frac{M_g}{M_v} = \frac{\phi}{c_{pg}(T_g - T_s)}$$
(3-87a)\*

$$= \frac{H_{v,m}}{c_{pg}(T_g - T_{1,m})}$$
(3-87b)

(3-26)

Equation (3-87b) is the desired analytical expression. Since a value for the liquid temperature  $T_{1,m}$  uniquely determines the value for both the vapor pressure  $p_{V,m}$  and the heat of vaporization  $H_{V,m}$ , it follows from Eq. (3-87b) that the maximum liquid temperature is expressed solely in

\*Note that if Eqs. (3-85) and (3-86) are combined to eliminate the left-hand side of Eq. (3-87a), and if the parameter B<sub>h</sub> is introduced, then the resultant expression can be written in the form

$$C_{s} = \frac{B_{h}}{1 + B_{h}}$$

Equation (3-26) was introduced in Subsection 3.1.2 without derivation.

terms of the flow parameters p,  $T_{\mbox{g}}$  ,  $c_{\mbox{pg}}$  , and  $M_{\mbox{g}}$  for the gas stream.

Figure 34 presents, for purposes of illustration, calculated values of the maximum liquid temperature  $T_{1,m}$  for an air-water system\*. Shown in the figure is a plot of  $T_{1,m}$ as a function of the static pressure p with the gas stream temperature  $T_g$  as a parameter. The saturation curve for water that is presented in the figure illustrates the approximate error that is introduced by the assumption that the maximum film temperature is equal to the saturation temperature for the liquid coolant at the local static pressure, p. That assumption was employed, for example, by Refs. (20, 30, 23, 42). References (25), (26) and (28) suggest that the calculated results shown in Fig. 34 are in good agreement with the reported experimental data for the film temperature  $T_{1,m}$  for an air-water system.

3.4.2 Comparison of the Experimental and the Predicted Results for the Maximum Film Temperature

1

F

Section 2.7 previously discussed the nature of the experimental measurements for the maximum liquid temperature T1.m<sup>\*\*</sup>. Presented in Fig. 18 of that section was a comparison

- \* The calculations were performed by employing Eq. (3-87b) in conjunction with the tabulated data of Refs. (40) and (41). A sample calculation is presented in Appendix F.
- \*\*Recall from Section 2.7 that the maximum liquid temperature  $T_{1,m}$  was determined experimentally by measuring the temperature for the liquid at the point of liquid withdrawal,  $T_{1,2}$ .





146

41.6

Í

Ĩ

۲ ب

B

[]

between the temperature  $T_{1,m}$  and the boiling temperature for the various liquids\* of the subject investigation at the prevailing gas stream pressure. Figure 18 illustrated that the difference between those two temperatures was generally significant, with the maximum difference for the subject investigation being equal to approximately 350F. A similar comparison is made in this section between the experimental values for the maximum liquid temperature and those that are predicted by the theory outlined in the foregoing subsection.

The primary equation (3-87b) was employed, together with the thermodynamic data presented in Appendix D of the present report, to calculate the values for  $T_{l,m}$  for the gas stream pressure and temperature corresponding to the experimental tests conducted. Figure 35 illustrates the results of those computations. Presented in the figure is a plot of the measured value for the liquid temperature versus the predicted value for the temperature  $T_{l,m}$ . The cross-hatched areas in Fig. 35 represent the range of the experimental values for the maximum liquid temperature that were realized for each liquid at a particular value for the gas stream temperature and pressure\*\*. The agreement between the experimental and analytical results shown in Fig. 35 is generally quite satisfactor". The only substantial

ţ

. 3

73

غ.

\* water, methanol, butanol, and RP-1
\*\*Several hundred datum points are represented by the cross-hatched areas in Fig. 35.



ł

ţ

.

ţ,

-

East.

. 1

1

0



ĩ

mail stall a series

difference between the measured and the predicted values for  $T_{1,m}$  occurs for some of the experimental data that were obtained with RP-1 as the film coolant. However, part of that difference may be due to the fact that RP-1 was the liquid for which the maximum temperature rise occurred after it was introduced onto the test plate. Therefore, the liquid film may not have always achieved its maximum temperature by the time the film reached the point of withdrawal. However, even if that were not the case, the results of Fig. 35 suggest that the simple theory presented herein for the prediction of the maximum film temperature is sufficiently accurate for the purposes of most engineering-type calculations.

Ľ

E

### 4. CORRELATION OF THE DATA OF THE PRIMARY REFERENCES

### 4.1 Introduction

The experimental data for mass transfer that are reported in the subject report are limited in two fundamental respects: (a) the liquid-film cooled length was held constant at 10 inches, and (b) the gas stream temperature was limited to 600F and the gas stream pressure was limited to 150 psia so that the experimental problems would be minimized. The purpose of the present section is to analyze those data that have been reported in the literature for the problem of liquid-film cooling that are not so restricted. The analysis of those data will give some insight into the general usefulness of the correlation for mass transfer that is presented in the subject report.

The experimental data that are reported by the primary references (19) (20) and (24) are analyzed. The data reported by Kinney, et al. (19) and those reported by Emmons (20) were obtained under essentially constant gas siteam velocity conditions. The basic experimental configuration for each of those two investigations was the flow of a hot gas through a horizontally mounted tube with the liquid film introduced around the circumference of the tube.

The experimental data reported by Hermann, et al. (24) are for the case of a strongly accelerating gas flow. They investigated the problem of liquid-film cooling an axisymmetric convergent-divergent nozzle. For that investigation, the gas stream was accelerated from a Mach number of approximately 0.07 at the plane of liquid injection to a Mach number of 1.92 at the exit of the nozzle.

Table 3 presents a comparison of the nominal flow parameters for the experimental investigation of Kinney, et al. and of Emmons with those for that portion of the present investigation where the gas stream velocity was essentially constant over the length of the wetted test plate. Table 3 shows that the experimental parameters for the subject references represent a substantial extension over those for the experimental program reported herein. For example, the maximum gas stream temperature, T<sub>g</sub>, investigated in both references is significantly greater than the maximum temperature of 600F for the present investigation. Moreover, the gas stream pressure that was employed by Emmons is more than three times that employed in the subject investigation; the maximum gas stream velocity that was investigated by Kinney is nearly four times the maximum that is reported herein; and the maximum film cooled length that is reported by Kinney is more than three times longer than the 10-inch film cooled length that was employed in the present investigation.

ო
Ð
<b>م</b>
g
$\vdash$

Comparison of the Nominal Flow Parameters for the Primary References to Those for the Present Investigation--Constant Velocity Gas Stream

Parameters	Present Investigation	Kinney, et al. (19)	Emmons (20)
T <sub>d</sub> , Gas S∵ream Temperature, F	40-600	800-2000	3640
y , Gas S ream Pressure, psia	75-150	23-37	500
u <sub>d</sub> , Gas Stream Velocity, fps	34-430	1000-1700	38-127
, Gas Mass Velocity, lb/ft <sup>2</sup> -sec	12-93	39-82	10-33
M <sub>o</sub> , Momentum Parameter, 1b <sub>f</sub> /ft <sup>2</sup>	20-1087	700-3000	12-130
X <mark>e</mark> . Entrainment Group, lb <sup>-</sup> <sup>4</sup>	1570-19,790	6700-13,700	1600-7300
X <sub>r</sub> . Roughness Group, lb <sub>f</sub> <sup>}</sup> /ft	5-38	36-70	5-18
l , Film Cooled Length, in.	10	9-36	3-8
Liquid Film Coolants	Water	Water	Water
	Methanol		Ethanol
	Butanol		Ammonia
	RP-1		Freon-113

•

152

Ĩ

J

1

Į

]

]

]

]

1

l

Table 4 presents a similar comparison of the nominal flow parameters for the experimental investigation of Hermann, et al. with those for that portion of the subject investigation where the gas stream was accelerated over the length of the film cooled test section. Again, the experimental parameters for the referenced investigation represent a considerable extension over those for the experimental program reported herein, particularly with reference to the Mach number,  $M_a$ , the gas stream velocity,  $u_g$ , and the staticstagnation pressure ratio,  $p/p^0$ .

Obviously, any evaluation of the present theory in terms of the referenced experimental data is only as conclusive as is the accuracy of those experimental data. In that regard, it should be mentioned that the wetted surface area (and therefore the net rate of mass transfer from the liquid film to the gas stream) was determined in each of the subject references by external wall temperature measurements. Moreover, each reference employed the assumption that the liquid film terminated at that point where the wall temperature just exceeded the boiling temperature for the liquid at the prevailing local static pressure. Section 3.4 herein showed, however, that that assumption is never exactly correct, with the error in the assumption increasing with decreasing gas stream temperature and/or decreasing gas stream pressure. Thus, there is some limitation in accuracy introduced by the experimental technique employed

## 154

### Table 4

Comparison of the Nominal Flow Parameters of the Primary Reference (24) to Those of the Present Investigation--Accelerating Gas Stream

	Parameter	Present Investigation	Reference (24)
•			
Τ <sup>ο</sup> g,	, Gas Stream Stagnation Temperature, R	860	1460-1940
р <b>о</b> ,	, Gas Stream Stagnation Pressure, psia	50-150	69-82
p/p <sup>0</sup> ,	Static-Stagnation Pressure Ratio	0.84-0.98	0.14-0.99
<b>T</b> ∕T <sup>0</sup> ,	Static-Stagnation Temperature Ratio	0.95-0.99	0.58-0.99
M <sub>a</sub> ,	. Gas Stream Mach Number	0.15-0.51	0.07-1.92
u <sub>g</sub> ,	Gas Stream Velocity, fp	os 200-660	150-2750
Liqui	d Film Coolants	Water Methanol	Water

j

in the aforementioned investigations.

.

E

E

E

į.

Furthermore, Kinney, et al. (19) comment that the experimental data for mass transfer that they reported could be reproduced in only a few cases to within 5 percent, and more generally to within only 10 or 15 percent. They also state that the accuracy of the experimental data for the short film cooled lengths--less than, say, 12 inches--was poorer than that for the longer film cooled lengths that were investigated. The problem that they experienced for the shorter wetted lengths was that the liquid film seldom terminated uniformly around the circumference of the wetted tube. Thus, it was difficult to interpret the external wall temperature measurements and to accurately determine the actual wetted surface area. Of the three references considered in the present section, the research program reported by Kinney, et al. is considered the most comprehensive and was conducted under the most well controlled experimental conditions. Thus the confidence in the accuracy of the experimental data for Refs. (20) and (24) is somewhat less than that for the experimental data of Ref. (19) because of the short film cooled lengths that were employed in those two investigations.

One additional problem occurred in the evaluation of the theory of the present report from the data that were presented in the subject references. Namely, how to properly determine the value for the surface tension  $\sigma$  that

enters into the entrainment group  $X_e$ . In each of the three subject references, the liquid coolant was injected at essentially the ambient temperature, so that a substantial change occurred in the value for the temperature of the liquid (and therefore the value for  $\sigma$ ) over the length of the wetted test section. This problem was particularly acute for the investigation due to Emmons because the final equilibrium film temperature at the end of the liquid .lm was generally very high. For example, for the film cooling experiments reported by Emmons with water as the liquid phase, the temperature for the water increased from approximately 70F at the point of liquid injection to something in excess of 400F at the terminus of the liquid film. The corresponding decrease in the surface tension for the water coolant was roughly 50 percent. Obviously, the change in the temperature of the liquid coolant over the length of the wetted test section creates some uncertainty about how the parameter  $X_e$  should be evaluated. It is also obvious, however, that such a variation in the liquid film temperature would characterize the general application of liquidfilm cooling. Because of the genera' importance of the problem of determining a proper reference temperature for the calculation of the surface tension  $\sigma$ , this problem is discussed in greater detail in Section 4.2 that follows.

Sections 4.3, 4.4, and 4.5 consider the analysis of the data reported in Refs. (19), (20), and (24), respectively.

ł.

ł

(i)

l

The basic assumptions that were employed to analyze those experimental data are listed in each section, and a comparison is made in each case between the reported rates of mass transfer and those that are predicted from the present theory.

# **4.2** An Approximate Method for Evaluating the Surface Tension

An approximate method for the evaluation of a reference liquid temperature, which will be denoted by  $T_{l,r}$ , that can be employed to estimate the value for the surface tension in the group  $X_e$  is presented in this section. The method strictly applies to only the case where the wetted surface is adiabatic\*, but it could be readily extended to the more general case where the film cooled region is also cooled externally.

For the general case of liquid-film cooling, the bulk temperature for the liquid increases from a minimum value at the point of injection,  $T_{1,1}$ , to a maximum value at (or near) the end of the liquid film. That maximum film temperature, denoted by  $T_{1,m}$ , can be estimated from the theory outlined previously in Subsection 3.4.1. The basic problem, therefore, is to determine how the minimum and the maximum film temperatures should be "weighted" so as to determine the reference temperature  $T_{1,r}$ .

First it is convenient to introduce a new parameter  $\overline{\phi}$ \*This case includes the subject references (19), (20), (24).

which is defined as the average value for the energy balance parameter  $\phi$  over the length of the liquid film. The parameter  $\overline{\phi}$  can be reasonably approximated by the following expression. Thus\*

 $\overline{\phi} = H_{v,m} + h_{l,m} - h_{l,l}$  (4-1)

where

The rate at which the bulk temperature for the liquid film approaches the maximum film temperature,  $T_{l,m}$ , once that it is injected onto the test plate is a function of how much of the total energy transfer to the surface of the liquid film results in an increase in the sensible enthalpy of the liquid and how much results in the evaporation of liquid. That is, it is dependent on the relative magnitudes of the terms  $H_{v,m}$  and  $(h_{l,m} - h_{l,l})$  on the right of Eq. (4-1). Thus, a weighting parameter, denoted by "w", can be

\*It can be shown that in most cases, the group of terms  $(H_V + h_1 - h_{1,1})$ , where  $H_V$  and  $h_1$  are evaluated at any temperature between  $T_{1,1}$  and  $T_{1,m}$  is essentially a constant over that range of liquid temperatures (22). Thus the evaluation of  $H_V$  and  $h_1$  at  $T_{1,m}$  is mostly a matter of convenience.

defined as follows:

;

1-6 2.4

見いり

1

.

ちょうちょう

B

$$w = H_{v,m} / \overline{\phi}$$
 (4-2)

To illustrate: if the value for w is large (i.e., near unity) so that most of the energy transfer to the liquid film results in evaporation of liquid, then the reference temperature  $T_{1,r}$  should be chosen as nearily equal to the maximum film temperature  $T_{1,m}$ . However, if w is very small, so that most of the energy transfer to the liquid film results in an increase in the sensible enthalpy of the liquid, then the average temperature for the liquid over the length of the wetted test surface will be much less than  $T_{1,m}$ , and the reference temperature  $T_{1,r}$  should be weighted more in favor of the minimum liquid temperature  $T_{1,1}$ . Therefore, in view of the lack of any definitive data for the problem, it is suggested herein that the reference temperature  $T_{1,m}$  be approximated by the following expression. Thus

 $T_{1,r} = w T_{1,m} + (1-w) T_{1,1}$  (4-3)

Equation (4-3) was employed to estimate the surface tension for the analysis of the data of the primary references (19) and (20). The values that were computed for  $T_{l,r}$ are given in the appropriate subsection.

### 4.3 The Data Due to Kinney, Abramson, and Sloop\*

### 4.3.1 Introductory Remarks

Kinney, et al. (19) present experimental data for the liquid-film cooling of a 2-inch and two 4-inch inside diameter ducts. The 2-inch diameter duct and one of the 4-inch diameter ducts were seamless tubes having a honed inner surface. These two ducts are termed herein the smooth-surface ducts. The second 4-inch diameter duct was a 4-inch inside diameter rolled tube with a longitudinal weld. Although the inside surface of the tube was machined after welding, some surface roughness and waviness remained. This tube is termed herein the rough-surface tube.

Subsection 4.3.2, which follows, describes the basic experimental apparatus for Ref. (19). The primary assumptions that were employed in order to obtain a comparison of the subject data with the theory that is developed in the present report are outlined in Subsection 4.3.3. Subsection 4.3.4 then presents the comparison between the reported experimental data and the calculations from the present theory.

#### 4.3.2 The Basic Apparatus

ě

Citing State

Figure 36 illustrates schematically the basic experimental apparatus employed in the investigation reported in Ref. (19). The basic apparatus consisted of a horizontally

\*Refer to Table 3 for the nominal flow parameters for these data.

4



Schematic Diagram of Apparatus of Kinney, Abramson, and Sloop Fig. 36.

μ., ...',

mounted Inconel approach section that was 40 inches long and a film cooled test section that was 48 inches long. The walls of the experimental duct work were essentially adiabatic. Mixing baffles and a calming section were installed upstream of the approach section in order to obtain a uniform gas stream having a relatively low level of turbulence. A gasoline-air combustor was employed to produce the hot gas stream.

The liquid injector that was employed was fabricated from a porous wire cloth (36 mesh) that was mounted flush with the internal flow passage. The porous injector element was two inches long in the direction of the gas flow. The data reported were for a water film.

Ì

Ĵ.

i

)

1

Ĵ

### 4.3.3 The Assumptions

The basic assumptions that were employed to analyze the experimental data for the subject reference are as follows: (a) The gas flow was treated as boundary-layer type flow over a flat plate. The velocity boundary layer was assumed to be fully turbulent and to develop from the leading edge of the 40-inch long approach section.

wan de Loner & Uzbringer

1

(b) The approach section was assumed to be adiabatic so that a thermal entrance length  $x_t = 40$  inches existed. (c) The thermodynamic properties for the gas stream were assumed to be the same as those for dry air at the prevailing gas stream temperature. The value for the gas stream
Prandtl number, Pr, was taken to be 0.7 for all computations.(d) Radiation heat transfer was assumed negligible.

(e) The reference temperature  $T_{1,r}$ , for the evaluation of the surface tension  $\sigma$  in the entrainment group  $X_e$ , was estimated from the procedure outlined in Section 4.2 to be 165F for the lowest gas stream temperature that was investigated (800F), and 210F for the highest gas stream temperature that was investigated (1600F). The values for the reference temperature for intermediate values of the gas stream temperature were determined by a linear interpolation between those limiting values.

(f) The value for the energy balance parameter  $\phi$  was calculated from Eq. (4-1).

## 4.3.4 Comparison of the Present Theory With the Experimental Data

j.

The primary equation of the present theory is repeated here for convenience. Thus

$$\frac{m_1}{m_s^{T}} = (1.0 + \frac{3.0}{\chi_r^{0.8}}) \exp[5.0 \times 10^{-5} (\chi_e - 1000) (1/1_0)^{b}]$$
(3-64)

Equation (3-64) gives the desired relationship between the rate at which liquid is injected,  $m_1$ , and the resultant film cooled length, 1. The simple rate of mass transfer,  $m'_s$ , was computed from the procedure outlined in Subsection 3.1.2, together with the aforementioned assumptions. Figure 29 of

Subsection 3.2.2 predicts a value of approximately 0.6 for the parameter b in Eq. (3-64) for the subject flow situation. The present subsection will investigate the validity of that prediction (and, therefore, the validity of the model that was employed in the analysis of Subsection 3.2.1).

Figure 37 presents a comparison of the experimental data for mass transfer in terms of the film cooled length versus the injected liquid flowrate for the 4-inch diameter, smooth-surface tube with the results that were computed for both (a) the simple theory for mass transfer (Fig. 37(a)), and (b) the present theory for mass transfer (Fig. 37(b)). Figure 38 presents a similar comparison for the 2-inch diameter, smooth-surface tube. The solid curves presented in these two figures represent the data that are presented in Fig. 6 of Ref. (19). Approximately 25 datum points are represented by each of the solid curves. The values for the gas stream temperature and the gas stream Reynolds number (based on the tube diameter D) are indicated in the figures for each set of experimental data.

「おおお」となったので、「ころのできた」を見ていて、「「「「ある」」を見ていたかできる。

Figures 37(a) and 38(a) again illustrate how inadequate the simple theory is for the problem of liquid-film cooling. For example: Figure 37(a) shows that at a gas stream Reynolds number of 4.7 x  $10^5$  and a gas stream temperature of 1600F, the value for m<sub>1</sub> required to establish a film cooled length of 2.2 ft is more than three times the value for m<sub>1</sub> that is predicted from the simple theory for mass transfer.













1.100

เมษณีประสานสร้างสารที่จะ ะร่างกระมาสามาร . จะมีระสารระการสะการที่มาหลายแกะ - - - -



1.41, 11

------

1 × ++++



÷

ļ

7

Ī

]

Ţ

]

**, ]** 

.]**j** 

]

Ŋ

ß

J

Comparison of Simple Theory and Present Theory with Data for Mass Transfer Due to Kinney, Abramson, and Sloop. 2-inch Diameter Smooth-Surface Tube. Water Film Coolant F1g. 38.

The possible catastrophic consequences of designing a liquidfilm cooled system on the basis of the simple theory are evident.

Figures 37(b) and 38(b) present the resultant computations for the present theory of mass transfer. Because the values for the gas stream pressure were not indicated explicitly for the data that were reported in Ref. (19), a band is shown for the present theory for each set of experimental da-For the two-inch diameter duct, p was assumed to vary ta. from 30 to 35 psia. And f the 4-inch diameter duct, p was assumed to vary from 25 to 30 psia. The larger values for  $m_1$ at any particular value of 1 correspond to the lower values for pressure. (For the same value of gas stream Reynolds number, or gas mass velocity G, the value for the momentum parameter M<sub>n</sub> increases with decreasing pressure. And the amount of entrained liquid increases with increasing  $M_{o}$ .) A value of 0.5 for the parameter b in Eq. (3-64) was employed to compute the results that are shown in Figs. 37(b) and 38(b) for the smooth-surface ducts. Recall that the simple theory presented in Subsection 3.2.2 for the extension of the results of the present investigation to an arbitrary film cooled length showed that for the classical case of turbulent boundary layer flow over a flat plate, a value of approximately 0.6 for the parameter b was predicted. That value for b resulted in predicted mass transfer rates that were generally 10 percent greater than the calculated

「青龍」をいるため、「「「

.

1、牛按 "卢南大不也"

**\***.

results that are shown in Figs. 37(b) and 38(b). However, as discussed earlier in Section 4.1, that agreement between the predicted and the experimental results is probably within the accuracy of the experimental data.

Comparison of Figs. 37(b) and 38(b) shows that the agreement between the experimental results and the results calculated from the present theory is somewhat better for the 4-inch diameter, smooth-surface duct than it is for the 2-inch diameter duct. However, at least two reasons can be suggested as to why the experimental data for the 4-inch diameter duct should be weighted more heavily in an evaluation of the present theory: (a) the gas flow in the 4-inch diameter duct should more closely correspond to the case of boundary layer flow over a flat plate\*, and more importantly, (b) it was stated in the subject reference that the agreement between the data for any two experiments conducted at the same flow conditions was generally much poorer for the experiments with the 2-inch diameter duct than it was for the 4-inch diameter ducts.

Figure 39 presents a similar comparison between the calculated results and the experimental results for the 4-inch diameter, rough-surface tube. The solid curves shown in the figure represent the data that are presented in Fig. 6(c) of Ref. (19). Approximately 15 datum points are represented by each of the solid curves.

3

\*See assumption (a) in Subsection 4.3.3.







Comparison of Simple Theory and Present Theory With Datd for Mass Transfer Due to Kinney, Abramson, and Sloop. 4-inch Diameter Rough-Surface Tube. Water Film Coolant Fig. 39.

169

Ren x 10<sup>-5</sup>

6.8 5.3 6.7

t<mark>c (f)</mark> 1400 1400 1400

0.24

0.20

0.16

ļ

. .

•

ſ

I

I

Figure 39(a) again shows the poor agreement between the experimental results and the results that were computed from the simple theory for mass transfer. The mass transfer rates that were realized experimentally are shown in Fig. 39(a) to be as much as 400 percent greater than those predicted by the simple theory.

The computed results shown in Fig. 39(b) for the present theory were calculated from Eq. (3-64) with a value of 0.6 for the parameter b. The band that is presented for each set of experimental data represents a variation in the static pressure of 25 to 30 psia. In general, the agreement between the experimental data and the results that are calculated from the present theory is satisfactory. The greatest difference occurs for the experimental data at the larger rates of liquid injection,  $m_1$ . However, even for those data, the present theory is in agreement to within the reported accuracy of the experimental data.

The following conclusion is drawn from the foregoing comparison of the computed results of the present theory with the experimental data of Ref. (19) for both the smoothsurface ducts and the rough-surface duct: to perform conservative engineering calculations so as to allow for unknown factors such as the physical condition of the wetted surface, the value of 0.6 for the parameter b in Eq. (3-64) ļ

Ţ

Ĩ

1

should be employed\*. That conclusion is in good agreement with the result that was predicted by the analysis of Subsection 3.2.1 and Subsection 3.2.2.

4.3.5 A Comment on the Presentation of Data for Liquid-Film Cooling in Terms of Dimensionless Parameters

Consider the presentation of the experimental data of Ref. (19) in terms of a Nusselt number-Reynolds number type of plot. The effective\*\* average Nusselt number, denoted by  $Nu_e$ , can be calculated from the following expression. Thus

$$Nu_{e} = \frac{m_{1}/l}{G} \frac{Re_{D}}{B_{h}}$$
(4.4)

Figure 12 of Ref. (19) presents plots of Nu<sub>e</sub> as a function of Re<sub>D</sub> showing some of the experimental data for both the smooth-surface tubes and the rough-surface tube. Figures 40 and 41 of the present report show a comparison between those data and the results that are calculated from the

\* Note that this conclusion applies to only that case where the gas stream velocity is essentially constant over the length of the liquid film.

ことないまというで、大ちというとなどになるないというであるというよう

E

E

\*\*The term "effective" is introduced because Nu includes the mass transfer due to entrainment in addition to that due to evaporation.



こころになっている。 ここれないないないないないないないないない

Fig. 40. Nusselt-Reynolds Number Plot of Data Due to Kinney, Abramson, and Sloop. Smooth-Surface Tubes

3

I

Ŋ



Fig. 41. Nusselt-Reynolds Number Plot of Data Due to Kinney, Abramson, and Sloop. Rough-Surface Tube

Ţ

E

F

E

E

ſ

ß

こう 記留まれい

present theory\*. Figure 40 shows the results for the smooth-surface ducts, and Figure 41 presents the results for the rough-surface duct. Note that each set of experimental data is characterized by a certain value for the rate of liquid injection,  $m_1$ ; the values for  $m_1$  are listed in the figure. The interrupted curve shown in both Fig. 40 and Fig. 41 corresponds to the case of simple mass transfer, where the various "F" correction factors (see Subsection 3.1.2) were employed in the evaluation of the average Because these correction factors are de-Nusselt number. pendent on parameters other than the gas stream Reynolds number, the curves for simple-mass transfer are labeled "approximate". In evaluating the net mass transfer rate,  $m_1$ , a value of 0.5 for the parameter b in Eq. (3-64) was utilized in the calculations for the smooth-surface ducts and a value of 0.6 was employed for the rough-surface duct.

\*In the calculation of the Nusselt number in Pef. (19), the thermal conductivity for the gas stream,  $k_g$ , was introduced. However, the values for  $k_g$  that were employed in those calculations were generally 10 percent to 15 percent greater than those listed, for example, in Ref. (40). To illustrate, Appendix C of reference (19) gives a value for  $k_g = 15.6 \times 10^{-6}$  (Btu/sec-ft-F) for a gas stream temperature T = 2000F. But Ref. (40) gives a value for  $k_g = 13.1 \times 10^{-6}$  (Btu/sec-ft-F) at the same temperature. Therefore, to make some correction for this fact, the values for Nu<sub>e</sub> presented in Figs. 40 and 41 are greater than those presented in the original reference by a factor of 1.10. The bands shown in Fig. 40 and Fig. 41 for the theory that was developed in the present report correspond to the gas stream pressure variation indicated in the figures. Figures 40 and 41 demonstrate again the poor agreement of the simple mass theory and the much more satisfactory agreement of the present theory with the experimental results. However, a more important point should be noted from the comparisons illustrated in the figures. Investigators of the problem of liquid-film cooling have commonly presented their experimental data in the form of such a Nusselt-Reynolds number type of plot (19, 20, 24). The experimental data of the present investigation, however, together with the data due to Kinney, et al. (19), show that this is generally a very poor way to present such experimental data because it does not account for the considerable importance that the parameters  $M_0$ ,  $\sigma$ , and  $m_1$  can have on the net rate of mass transfer from a thin liquid film to a high velocity gas stream. This type of data presentation can be very misleading for the practicing engineer who is concerned with the problem of designing a liquid-film cooled system, but who is not cognizant of the importance of the interfacial phenomena, particularly the entrainment phenomenon. It is hoped that the results of the experimental investigation reported herein clearly demonstrate the general importance of the interfacial phenomena in the calculation of the net rate of mass transfer from a thin liquid film to a hot gas stream.

#### 4.4 The Data Due to Emmons\*

4.4.1 Introductory Remarks

Emmons (20) presents experimental data for liquid-film cooling of a cylindrical rocket motor combustion chamber. Experimental determinations were made of the effects of changes in the gas stream temperature, pressure, and Reynolds number upon the required flow rate of a single liquid (water) needed to establish a range of Im cooled lengths (1 = 4 to 8 inches). Experiments were also conducted with other coolants\*\*, at one gas stream flow condition, to study the influence of the physical property values for the liquid on the rate of mass transfer from the liquid film to the gas stream. Most of the data presented in Ref. (20) were obtained for a gas stream temperature of 4100 R and a chamber pressure of 500 psia. Only those data are reviewed herein.

Subsection 4.4.2, which follows, describes the basic experimental apparatus that was employed by Emmons. The primary assumptions employed in the calculation of the mass transfer rates from the theory of the present report are discussed in Subsection 4.4.3, and the comparison is shown in Subsection 4.4.4 between the measured and the predicted results.

\* See table 3 for the nominal flow parameters for these data.
 \*\*anhydrous ammonia, ethyl alcohol, and Freon-113

## 4.4.2 The Basic Apparatus

Figure 42 illustrates schematically the basic experimental apparatus employed by Emmons in his study of liquidfilm cooling. The experimental rocket motor burned gaseous hydrogen and air as the propellants. The inside diameter of the chamber was 3.14 inches. Two convectively cooled cylindrical sections were installed upstream of the adiabatic film cooled test section. The first convectively cooled chamber was 9.42 inches long and the second convectively cooled chamber was 1.48 inches long. Emmons utilized the 1.48-inch long section to calorimetrically estimate the hea'. transfer to the dry wall immediately upstream of the film cooled test section. An observation will be made in Subsection 4.4.5 regarding the measurements that were obtained for that 1.48-inch long section.

The liquid coolant was injected through 72 V-shaped grooves that were milled into the downstream face of the film coolant injector. The liquid was introduced onto the test surface with a component of velocity in the tangential, or circumferential, direction.

#### 4.4.3 The Assumptions

The primary assumptions employed to analyze the experimental data of Ref. (20) are summarized as follows: (a) The gas flow was treated as boundary-layer type flow over a flat plate. The boundary layer was assumed to be

手にという歌には

. .

> نړ د



•

ţ

.

A -----

1

-

ļ

fully turbulent and to develop from the leading edge of the 9.42-inch long convectively cooled section.

(b) The temperature for the liquid film at the point of injection was much less than the wall temperature immediately upstream so that a thermal entrance length  $x_t = 9.42 + 1.48 = 10.90$  inches existed.

(c) No chemical reactions occurred between the vapor for the film coolant and the main stream gas within the boundary layer that developed over the liquid film.

(d) The value for the energy balance parameter  $\phi$  was calculated from Eq. (4-1). The computed values for  $\phi$  are shown in Table 5 below.

(e) Radiation heat transfer was neglected.

(f) The properties of the gas stream were estimated from the results presented by Rannie in Ref. (43)\*. For a gas stream temperature of 4100 R and a chamber pressure of 500 psia, the pertinent thermody...amic properties that were estimated from Ref. (43) are as follows:

 $M_{g} = 23.0 \ lb/lb-mole$   $c_{pg} = 0.43 \ Btu/lb-R$   $\gamma = 1.255$   $\mu_{g} = 4.75 \ x \ 10^{-5} \ lb/ft-sec$ 

\*Emmons (20) docs not list the values that he assumed for the thermodynamic properties of the hot gas stream. (g) The reference temperature for the evaluation of the surface tension  $\sigma$  was estimated from the procedure that was suggested in Section 4.2. Table 5 presents the values for the pertinent parameters that entered into that evaluation, together with the computed values for  $T_{1,r}$  and  $\sigma$ . Also listed in Table 5, for completeness, are the values for the heat transfer parameter  $B_h$  and the interfacial vapor concentration  $C_s$ .  $B_h$  was calculated from the right-hand side of Eq. (3-87b), and  $C_s$  was calculated from Eq. (3-26)\*.

Of the aforementioned assumptions, Assumption (3) that the heat transfer due to radiation is negligible is probably one of the least valid. The products of combuction of air and gaseous hydrogen contain a substantial quantity of water vapor which is a strong radiator, particularly at the temperature level of 4100 F. Perhaps 25 percent or more of the net heat flux to the surface of the liquid film was due to radiation heat transfer. An attempt was not made, however, to estimate that radiation heat transfer. Significant conclusions can be drawn about the usefulness of the theory developed in the present report for the prediction of the rate of mass transfer from a liquid film to a hot, high velocity gas stream without trying to solve that complex problem.

\*Appendix D.2 presents  $H_v$ ,  $p_{v,s}$ , and  $\sigma$  as a function of the liquid temperature for the liquids that were investigated by Emmons.

Table 5

1

Į

[

ĺ

Į,

Į

I

. . . . . . .

Estimation of the Surface Tension for the Analysis of the Data Due to Emmons

Assumption		4100 R,	p = 500	psia						
Liquid Coolant	1,1 F	T1,2 F	Hγ,m Btu ib	φ Btu 1bu	a B	> 0 x x	<b>*</b> и С	3	ی ۳۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰۰	d dynes cm
Water	70	430	295	11.65	1.75	0.75	0.64	0.68	315	46.0
Ammonia	70	145	425	515	3.50	0.74	0.78	0.82	130	13.5
Ethanol	70	360	245	480	6.00	2.0	0.86	0.51	216	15.0
Freon-113	70	400	20**	011	70.0	8.2	0.58	0.128	128	16.0
* Emmons pressur (500 ps	tssumed at the a).	C <sub>S</sub> = 1.0 surface	for all of the ]	cases. Iiquid	He i film v	incorre Ias equ	ectly a al to	ssumed t the chan	chat the nber pres	vapor ssure
**Emmons a	ssumed	H. = 0.0	for Free	on-113.	See	the ab	ove fo	otnote f	or expla	unation.

٠

cplanation. >

•4 1.4.5

. ,

## 4.4.4 Comparison of the Present Theory With the Experimental Data

The aforementioned assumptions were employed in conjunction with the primary equation (3-64) to analyze the data of Ref. (20). The results of those computations are summarized in the following paragraphs.

Figure 43 shows the comparison of the computed results and the experimental data for the case where the gas stream Reynolds number was a variable and water was the film coolant. The gas stream Reynolds numbers are listed on the figure. Part (a) of the figure presents a plot of the mass transfer rate that was predicted from the simple theory for mass transfer versus the measured mass transfer rate. Part (b) of Fig. 43 then shows the comparison between the experimental data and the present theory. A value of 0.6 for the parameter b in Eq. (3-64) was employed to compute the results for the present theory.

Figure 43(b) shows that the present theory for mass transfer predicts a mass transfer rate that is consistently less than that realized experimentally. However, Fig. 43 also shows that the present theory is a substantial improvement over the simple-theory for mass transfer. Moreover, if the radiation heat transfer was accounted for, the agreement between the present theory and the experimental data would be improved. Indeed, if one also accounts for the likelihood of considerable experimental error in the



F

ł

I

ſ

ſ

E

- ういうしょうとうないないであるとないとうないないであるとないないないであるとうないないです。

F

I

I

I

A WAR Decker of the

.

reported data\*, it could probably be argued that the agreement between the present theory and the reported experimental data is quite satisfactory.

Figure 44 presents a similar comparison for the data that were obtained for the film coolants water, ammonia, ethanol, and Freon-113 at a gas stream Reynolds number of  $0.68 \times 10^5$ . Again, part (a) of the figure compares the experimental data for mass transfer with the mass transfer that was predicted from the simple theory; and part (b) of Fig. 44 shows the comparison between the theory that was developed herein and the reported data. Figure 44 demonstrates further that the proposed correlation for mass transfer is an improvement over the simple theory. If the neglect of the radiation heat transfer and the probable experimental error is taken into consideration, the agreement between the present theory and the experimental data shown in Fig. 44 is guite satisfactory.

1

X

R

#### 4.4.5 An Additional Comment of the Research Due to Emmons

It was indicated in Subsection 4.4.2 that the 1.48-inch long convectively cooled section\*\* (just upstream of the film cooled test section) was employed to determine the rate of heat transfer to a dry wall as a function of the pertinent \* See the discussion of Section 4.1.

\*\*Refer to Fig. 42.



185

Į

Ţ

Į

Į

E

E

[]

ľ

ſ

[

E

ſ

C

E

ことでである。「日本のなたいなたちをあるのなたいなたいとう」

gas stream flow conditions. Emmons compared those dry-wall measurements to the measurements he obtained for the liquidfilm cooled wall region and concluded that the rate of heat transfer to the liquid film was less than the rate of heat transfer to the dry wall for the same gas stream conditions. In essence, he concluded that the analogy between simplemass-transfer cooling and liquid-film cooling was valid. However, the experimental data reported herein, together with those reported by Kinney, et al. (19), show rather conclusively that the analogy between the simple-mass transfer process and the process of liquid-film cooling is generally very poor. In fact, the computed results shown in Fig. 43 and Fig. 44 suggest that the experimental data reported by Emmons do not support the analogy. This apparent discrepancy is discussed in the following paragraph.

Emmons found that the dry-wall heat transfer data could be correlated by a Nusselt-Reynolds number type of correlation that was suggested by Humble, et al. (44) for fullydeveloped turbulent tube flow. That particular correlation, however, specifies that the properties for the gas that enter into both the Nusselt number and the Reynolds number be evaluated at the wall temperature. The evaluation of the properties at the wall temperature is a common procedure employed to correlate heat transfer data for the case where heat is transferred from the wall to the gas stream (36) (i.e., the wall is at a higher temperature than the gas

186

8

0

1

Ŋ

gas stream flow conditions. Emmons compared those dry-wall measurements to the measurements he obtained for the liquidfilm cooled wall region and concluded that the rate of heat transfer to the liquid film was less than the rate of heat transfer to the dry wall for the same gas stream conditions. In essence, he concluded that the analogy between simplemass-transfer cooling and liquid-film cooling was valid. However, the experimental data reported herein, together with those reported by Kinney, et al. (19), show rather conclusively that the analogy between the simple-mass transfer process and the process of liquid-film cooling is generally very poor. In fact, the computed results shown in Fig. 43 and Fig. 44 suggest that the experimental data reported by Emmons do not support the analogy. This apparent discrepancy is discussed in the following paragraph.

R

Emmons found that the dry-wall heat transfer data could be correlated by a Nusselt-Reynolds number type of correlation that was suggested by Humble, et al. (44) for fullydeveloped turbulent tube flow. That particular correlation, however, specifies that the properties for the gas that enter into both the Nusselt number and the Reynolds number be evaluated at the wall temperature. The evaluation ~? the properties at the wall temperature is a common procedure employed to correlate heat transfer data for the case where heat is transferred from the wall to the gas stream (36) (i.e., the wall is at a higher temperature than the gas

186

C.

stream). That was the case that was investigated by Humble, et al. (44). However, for the case where the heat transfer is from the gas stream to the wall, as is the case for the data reported by Emmors, the standard procedure is to evaluate the physical properties for the gas at some intermediate temperature between that for the gas stream and that for the wall. The choice of the reference temperature for the evaluation of the properties for the gas can be very important. To illustrate, Kays (36) suggests that the Nusselt number for the case of turbulent heat transfer from the wall to the gas stream is greater than the Nusselt number for the case of turbulent heat transfer from the gas stream to the wall, all other conditions being equal, by the factor  $(I_g/T_s)^{0.25}$ . For the dry-wall experiments of Emmons, a characteristic value of  $(T_q/T_s)$  would be 4 with the resulting value for that factor being 1.41. It would appear, therefore, that the heat transfer rates that were realized for the dry-wall test section were approximately 40 percent greater than those that would be predicted by a standard correlation for fully developed, turbulent gas flow with the direction of heat transfer being from the gas to the wall of the tube. Part of that excess heat transfer could probably be attributed to radiation heat transfer. More significantly, however, it can be argued that part (or perhaps most) of that excess heat transfer was probably

F,

E

the rine and the second second

due to the fact that the assumption of fully-developed tube flow was a poor approximation to the flow over the 1.48-inch convectively cooled test section, particularly as regards the thermal flow field. Reference to the schematic illustration of the experimental apparatus presented in Fig. 42 shows that for the 9.42-inch long convectively cooled section, the external coolant (water) was introduced at the upstream end. It follows, therefore, that because of the heat transfer from the combustion chamber to the external coolant, both the temperature for the wall and for the coolant would increase in the flow direction until the coolant was withdrawn at a point just upstream of the 1.48-inch long convectively cooled test section. Assuming that the temperature of the external coolant was the same at the inlet to the 9.42-inch long section and at the inlet to the 1.48-inch long section, it then can be argued that there existed a decrease in wall temperature (that could have been substantial) in going from the downstream end of the 9.42-inch section to the 1.48-inch section. Such a change in wall temperature would cause the rate of heat transfer to the 1.48-inch section to be significantly higher, and it makes the assumption of a fully-developed thermal flow field much poorer, than if the wall temperature was everywhere isothermal. Emmons did not attempt to determine what the pertinent thermal boundary conditions were. Thus the validity of the above argument cannot be investigated further.

The argument is presented here solely as a possible explanation for the apparent high rates of heat transfer that were measured for the dry-wall test section. It is unfortunate, however, that Emmons arrived at such an important conclusion that the liquid-film cooling process and the process of simple-mass transfer were analogous without having considered the interpretation of the dry-wall heat transfer data in greater detail. Also it will be shown in Appendix A.4 of the present report that the major conclusion that Emmons drew with reference to his experimental data for the gas-vapor cooled region immediately downstream of the terminus of the liquid film is open to some question Lecause of the possible incomplete interpretation of the experimental data for the dry-wall heat transfer experiments.

تا ياند ال

## 4.5 The Data Due to Hermann, Leitinger, and Melnik\*

11

#### 4.5.1 Introductory Remarks

Hermann, et al. (24) experimentally investigated the feasibility of liquid-film cooling the nozzle of a hypersonic wind tunnel. For the preliminary experiments reported in the subject reference, the film cooling of a Mach number 1.92 axi-symmetric Laval nozzle was considered. They were not primarily interested in obtaining quantitative data on the rate of mass transfer from a liquid film to an accelerating gas stream. In fact, only three datum points are

\*Refer to Table 4 for the nominal flow parameters for these data.

presented that can be employed to evaluate the present Moreover, the accuracy of these data points is theorv. questionable, primarily due to the fact that the instrumentation was limited. Specifically, only 15 thermocouples that were welded to the external surface of the film cooled wall were utilized to determine the average film cooled length. The thermocouples were distributed both axially along and circumferentially around the nozzle. Only 9 of the 15 thermocouples were installed downstream of the throat of the nozzle where the liquid film usually terminated. Thus, due to the non-uniform way in which the liquid film usually terminates, it would be impossible, in general, to accurately determine the actual wetted area (or average film cooled length) from only 9 thermocouple readings. However. in spite of the foregoing limitations, it is still of some interest to compare the reported data for mass transfer with the mass transfer that is predicted by the present theory for the case of an accelerated flow.

Subsection 4.5.2 presents a discussion of the basic apparatus, and Subsection 4.5.3 outlines the primary assumptions that were employed to analyze the experimental data. Subsection 4.5.4 presents a comparison between the predicted and the experimental results.

To obtain some idea of the possible film coolant requirements for the cooling of the same axi-symmetric nozzle if it were installed on a high-energy propulsion device, ĮĮ.

Subsection 4.5.5 presents an illustrative calculation based on the present theory for the case where the stagnation temperature  $T_g^0$  = 4000 R and the stagnation pressure  $p^0$  = 500 psia.

#### 4.5.2 The Basic Apparatus

Figure 45 illustrates schematically the experimental apparatus of Ref. (24). The figure presents a Mach number 1.92, axi-symmetric Laval nozzle having a 3-inch long subsonic section and a 5-inch long supersonic section. The diameter at the inlet to the nozzle was 8 inches; the diameter at the throat of the nozzle was approximately 2.7 inches; and the diameter at the exit of the nozzle was approximately 3.5 inches. The walls of the nozzle were essentially adiabatic.

A J-47 jet engine combustor was employed to produce the hot gas stream. The maximum stagnation temperature that was investigated was 1940 R.

The film coolant (water) was introduced at the inlet to the subsonic portion of the nozzle. The coolant was injected onto the walls of the nozzle through 48 holes that were evenly spaced around the circumference of the nozzle. For most of the investigation, the diameter of each hole was 0.011 inches.

1

1

[]

I





ŧ

- 30 Mar

.

-----

:)

]

#### 4.5.3 The Assumptions

The primary assumptions employed to analyze the experimental data of Ref. (24) are summarized as follows. (a) The theory outlined in Subsection 3.3.1 for predicting the rate of simple mass transfer from a liquid film to an accelerating gas stream was employed to determine the simple mass transfer rates for the subject case. The boundary layer was assumed to be fully turbulent and to develop from the leading edge of the 3-inch long subsonic portion of the nozzle. In addition, the adiabatic-wall recovery temperature for the gas stream was assumed to be equal to the stagnation temperature  $T_q^0$ .

(b) Except for the thin boundary layer region next to the nozzle wall, the gas flow was assumed to be isentropic with a specific heat ratio of 1.33. With that assumption, the values for p,  $T_g$ , and  $u_g$  were computed as a function of the position along the axis of the nozzle. No corrections were made in the calculations for displacement effects.

(c) The thermodynamic properties for the gas stream were assumed to the same as those for dry air at the same temperature.

- うまい、そうしょう、という言葉を、いたい、おからして、かんかいの言葉を思めたななななななななななななない。 つます いまい

Ì

E

(d) The average temperature for the liquid film throughout the nozzle was estimated from Fig. 28 of Pef. (24) to be 220F for the case of the three datum points that are analyzed herein.

(e) A value of 1000 Btu/lb for the energy balance parameter

 $\phi$  was estimated from Eq. (4-1) with the substitution of the pertinent thermodynamic properties for water.

(f) Radiation heat transfer was neglected.

The foregoing assumptions were employed, together with the solution technique that was outlined in Subsection 3.3.2 for the present theory, to analyze the subject experimental data. The value for the parameter d in Eq. (3-74) that was employed in the calculations was 1.0. That value for d was found to correlate the data for the present investigation for the case where the gas stream was accelerated over the wetted wall region.

## 4.5.4 Comparison of the Present Theory With the Experimental Data

Reference (24) presents, for three different gas stream conditions, the approximate film coolant flow rate required to wet the first 4 to 5 inches of the nozzle. The comparison between the experimental results and the results predicted from both the simple theory for mass transfer and from the present theory (assuming d = 1.0) is as follows: Point 1 :  $T_{\alpha}^{0} = 1460R$ ,  $p^{0} = 69$  psia.

experimental:	m <sub>1</sub> = 0.037 lb/sec
simple theory:	m <sub>1</sub> = 0.017 to 0.021 lb/sec
present theory:	$m_1 = 0.023$ to 0.028 lb/sec

Point 2 : $T_g^0 = 1470R$ ,	p <sup>0</sup> = 82 psia.
experimental:	m <sub>1</sub> = 0.037 lb/sec
simple theory:	m <sub>1</sub> = 0.019 to 0.23 lb/sec
present theory:	$m_1 = 0.028$ to 0.033 lb/sec
Point 3 : $T_g^0 = 1940R$ ,	p <sup>0</sup> = 82 psia.
experimental:	m <sub>l</sub> = 0.097 lb/sec
simple theory:	m, = 0.029 to 0.036 lb/sec

present theory:

The foregoing results show that the agreement between the experimental and the predicted results is satisfactory for Point 2, is less satisfactory for Point 1, and is rather poor for Point 3. However, for each point, the result predicted by the present theory is better than that predicted by the simple theory for mass transfer.

 $m_1 = 0.043$  to 0.053 lb/sec

It is unfortunate that additional experimental data were not presented in the subject reference in such a form that the present analysis could be further evaluated. Moreover, it is unfortunate that the accuracy of the three datum points that are reported is uncertain. One question that remains to be answered is to what degree (if any) compressibility effects for Mach numbers greater than, say, 0.5 alter the form of the correlation proposed herein for the rate of mass transfer from a thin liquid film to an accelerating, hot gas stream.

1

Ē

F

# 4.5.5 An Illustrative Example for $T_a^0$ = 4000 R and $p^0$ = 500 psia

To illustrate further the present theory, an example is presented in this section to show the film coolant requirements (predicted) for the cooling of the above axi-symmetric nozzle for the case where the gas flow has a stagnation temperature of 4000 R and a stagnation pressure of 500 psia. These stagnation conditions are characteristic of the flow conditions that might be encountered in the application of film cooling to thermally protect an exhaust nozzle of a high-energy propulsion device. This example serves to indicate the type of deviation from the simple theory for mass transfer that can be expected in the liquid-film cooling of such an exhaust nozzle.

The assumptions employed in the calculations are the same as those listed in Subsection 4.5.3. Again, the value of 1.0 for the parameter d in Eq. (3-74) was employed in the computations.

Figure 46 presents a plot of the net rate of coolant injection as a function of the film cooled length of the exhaust nozzle (measured along the axis of the nozzle). The pertinent parameters that were employed in the calculations are summarized in the rigure. The interrupted curve shown in the figure represents the result that is predicted from the simple theory for mass transfer. The two solid curves represent the result that is predicted from the present

X

R

l



£

•
theory for two different values of surface tension  $\sigma$ . The larger value for  $\sigma$  (50 dynes/cm) represents roughly the case where water is employed as the film coolant. Most of the other film coolants that might be of interest (e.g., alcohols, hydrocarbon fuels) are approximately represented by the smaller value for  $\sigma$  (25 dynes/cm). Figure 46 shows that the present theory predicts, for the case where  $\sigma = 50$ dynes/cm, a coolant flow rate to cool the subsonic portion of the nozzle that is approximately 50 percent greater than that predicted from the simple theory. And, for the case where  $\sigma = 25$  dynes/cm, the present theory predicts an increase in the coolant requirements to cool the subsonic portion of the nozzle of 100 percent. More interestingly, however, Fig. 46 shows that to cool both the subsonic and the supersonic portions of the nozzle, by injecting the coolant at only the leading edge of the subsonic portion, the coolant requirements predicted from the present theory become substantial. For the case where  $\sigma = 50$  dynes/cm, about 6 times more coolant is required to cool the entire nozzle than is predicted by the simple theory. For the case where  $\sigma$  = 25 dynes/cm, the amount of coolant required to cool the entire nozzle is roughly 30 to 40 times that predicted by the simple theory. Although these predicted coolant flow rates are considerably greater than those predicted by the simple theory, the three datum points discussed in Subsection 4.5.4 suggest that the estimates from

i

 $\left\{ \left| \right. \right\} \right\}$ 

Ĥ

the present theory may even be low.

It is worthy of note that almost all of the difference between the results predicted by the simple theory and those predicted by the present theory is due to the entrainment phenomenon. That is, except for the very leading edge of the subsonic portion of the nozzle where the gas stream velocity is relatively low, the roughness parameter r is small, less than, say, 0.10.

It also is interesting to note the local rates of mass transfer from the liquid film to the gas stream that are predicted by the present theory for the subject illustrative case. Figure 47 presents a plot of the predicted local rate of mass transfer from the liquid film into the gas stream as a function of the axial location throughout the nozzle. The pertinent parameters employed in the calculations are listed in the figure. Only the one value for surface tension of 50 dynes/cm was employed in the computations. The result that is predicted by the simple theory is indicated in Fig. 47. In addition, three curves are presented in the figure to illustrate the result that is predicted by the present theory for three values of the net rate of coolant injection,  $m_1$ . The coolant flow rates shown are 0.5, 1.2, and 3.0 lb/sec. From Fig. 46, those coolant flow rates correspond to predicted film cooled lengths of approximately 4, 6, and 8 inches, respectively. Figure 47 illustrates that for all cases, the maximum value for the local rate of



The second se

.....

2 + 6800 FC

;

e .....

ł

-

ł

1

Support and the second

]

9

P

mass transfer occurs at (or very near) the throat of the nozzle. The figure also shows that the local rate of mass transfer predicted by the present theory can be substantially greater than that predicted by the simple theory. Of course, if the liquid coolant has a surface tension less than the 50 dynes/cm that is represented by the calculated results in Fig. 47, then the local rates of mass transfer would be substantially greater than those shown.

In summary, the illustrative example presented in this subsection suggests that the simple theory for mass transfer can lead to a catastrophic design of a liquid-film cooled exhaust nozzle. In most cases that are of practical importance, the entrainment phenomenon would completely dominate the mass transfer process.

It also follows from the results of the foregoing calculations that it is not practical, in general, to introduce the film coolant at only one location. The net rate at which liquid is entrained into the gas stream can be reduced substantially if the coolant is introduced at several axial locations. It would seem that this particular problem needs further experimental investigation.

#### 5. CONCLUSIONS

The following principal conclusions are drawn from the results of the subject investigation.

1. The scale of the small disturbance waves on the surface of the liquid film is principally characterized by the momentum parameter  $M_0$  for the gas stream, with the scale decreasing with increasing values of  $M_0$ .

2. The surface tension apparently affects the film surface characteristics because with water as the liquid phase (relatively high values of  $\sigma$ ), a second type of disturbance wave was frequently observed (two dimensional, long wavelength) that was never observed with either methanol, butanol, or RP-1 as the liquid phase (relatively low values of surface tension).

3. The basic interfacial structure is not altered appreciably by substantial variations in the liquid viscosity. Also it is not noticeably influenced by a significant variation in the rate of liquid flow in the liquid tilm so long as the gas stream velocity is greater than about 50 fps.

まったいっていたい

4. The rate of mass transfer from a thin liquid film into a high velocity, hot gas stream can be substantially greater (typically 5 to 10 times greater) than that predicted by the simple theory for mass transfer, with the phenomenon of entrainment normally accounting for the bulk of the mass transfer.

5. The mass transfer due to entrainment is a function of the rate of liquid injection  $m_{\perp}$ , the momentum parameter  $M_{o}$ , and the surface tension  $\sigma$ , with the amount of liquid entrained into the gas stream increasing with increasing values for  $m_{\perp}$  and  $M_{o}$  and decreasing values for  $\sigma$ . 6. The liquid viscosity was not found to influence the rate of entrainment, thereby contradicting the proposal by many observers that the liquid-film Reynolds number is a primary parameter characterizing the phenomenon of entrainment. 7. The "effective" roughness of the liquid film can result in an increase in the evaporative transfer of mass of as much as 80 percent or more over that predicted by the simple theory for mass transfer.

8. The data for mass transfer with a constant velocity gas stream was satisfactorily correlated in terms of two dimensional parameters, one characterizing the entrainment phenomenon and one characterizing the phenomenon of film surface roughness. The comparison of the experimental data with the present theory showed that the agreement was generally to within 5 to 10 percent.

Ę

9. There is some physical evidence that the dimensional entrainment group X<sub>e</sub> is equivalent to a dimensionless Weber number if the characteristic length for the latter is taken as the scale of the small disturbance waves on the surface of the liquid film. Moreover, there is some evidence that the phenomenon of entrainment of liquid from the surface of a thin liquid film is analogous to the phenomenon of atomization of liquid drops suspended in a gas stream. 10. The data for mass transfer with an accelerating gas stream was satisfactorily correlated by locally applying the correlation that was developed for the mass transfer data with a constant velocity gas stream and then numerically integrating over the length of the liquid film to obtain the total mass transfer.

The analysis that was developed to extend the results 11. of the present investigation to the general case of an arbitrary film cooled length was shown to agree favorably with the experimental data of Refs. (19) and (20). This favorable comparison also suggests that the relatively simple mass transfer correlation developed herein has considerable utility because the flow parameters that characterized these two investigations were substantially different from those that characterized the subject investigation. 12. The liquid coolant requirements for cooling the nozzle of a high energy propulsion device can be reduced substantially by injecting the liquid coolant at several axial locations, and by selecting coolants which have a high surface tension.

13. Correlation of the data for the maximum liquid temperature showed that the simple theory developed herein predicted that temperature with sufficient accuracy for most engineering-type calculations.

. . . . .

#### BIBLIOGRAPHY

i

ţ

ţ.

Coulbert, C. D., "Selecting Cooling Techniques for 1. Liquid Rockets for Spacecraft," Journal Spacecraft and Rockets, March-April 1964. Kramer, E. L., and Gronich, S., "A Transpiration-Cooled Nozzle at Applied to a Gas-Core Nuclear Propulsion 2. System," Journal Spacecraft and Rockets, May-June 1965. Benser, W. A., and Graham, R. W., "Hydrogen Convective Cooling of Rocket Nozzles," ASME Paper 62-AV-22 (1962). 3. Kelley, J. B., and L'Ecuyer, M. R., "Transpiration cooling, its theory and application," Jet Propulsion 4. Center, Purdue University, Report No. TM-66-5, June 1966. 5. Charvonia, D. A., "A study of the mean thickness of the liquid film and the characteristics of the interfacial surface in annular, two-phase flow in a vertical pipe," PhD Thesis, Purdue University, Lafayette, Indiana, May 1959. Schneiter, G. R., "Mass transfer in annular two-phase flow," PhD Thesis, Purdue University, Lafayette, 6. Indiana, June 1966. Craik, A. D. D., "Wind-generated waves in thin liquid films," J. Fluid Mech., <u>26</u>, 369, Part 2, 1966. 7. 8. Knuth, E. L., "The mechanics of film cooling-Part 1," Jet Propulsion, Nov.-Dec. 1954, p. 359. Kinney, G. R., and Abramson, A. E., "Investigation of Annular Liquid Flow with Cocurrent Air Flow in 9. Horizontal Tubes," NACA RM E5 C 3 (1951). 10. Tate, K., "Liquid-film cooling, a cold flow study of film instability and entrainment," MSME Thesis, Purdue University, Lafayette, Indiana, June 1967.

- 11. Levy, S., "Prediction of two-phase annular flow with liquid entrainment," Int. J. Heat Mass Transfer, <u>9</u>, 171, (1966).
- 12. Collier, J. G., and Hewitt, G. F., Trans. Inst. Chem. Engrs., <u>39</u>, 127 (1961).
- 13. Murgatroyd, W., "The role of shear and form forces in the stability of a dry patch in two-phase film flow," Int. J. Heat Mass Transfer, 8, 297, (1965).
- 14. Hartley, D. E., and Murgatroyd, W., "Criteria for the bread-up of thin liquid layers flowing isothermally over solid surfaces," Int. J. Heat Mass Transfer, <u>7</u>, 1003, (1964).
- 15. Hewitt, G. F., and Lacey, P. M. C., "The breakdown of the liquid film in annular two-phase flow," Int. J. Heat Mass Transfer, 8, 781, (1965).
- 16. Wicks, M., and Duker, "Entrainment and Pressure Drop in Co-current Gas-Liquid Flow, Part 1," AIChE Journal, 6, 463-468, (1960).
- 17. Martinelli, R. C., Boelter, L. M. K., Taylor, T. H. M., Thomsen, E. G., and Morrin, E. H., "Isothermal Pressure Drop for Two-Phase Two-Component Flow in a Horizontal Pipe," Trans. ASME, <u>66</u>, 139, (1944).
- 18. Collier, J. G., and Wallis, G. B., "Two Phase Flow and Heat Transfer," 4 Volumes, Notes for a Summer Course, July 24 - August 4, 1967, Department of Mechanical Engineering, Stanford University, Stanford, California.
- 19. Kinney, G. R., Abramson, A. E., and Sloop, J. L., "Internal-Liquid-Film-Cooling Experiments with Air-Stream Temperatures to 2000° F in 2- and 4-Inch Diameter Horizontal Tubes," NACA Report 1087 (1952).

E

U

- 20. Emmons, D. L., "Effects of Selected Gas Stream Flow Parameters and Coolant Physical Properties on Film Cooling of Rocket Motors," PhD Thesis, Purdue University, Lafayette, Indiana, August 1962.
- 21. Knuth, E. L., "The Mechanics of Film Cooling--Part 2," Jet Propulsion, January 1955.
- 22. Graham, A. R., "Film Cooling of Rocket Motors, Part II. Film Cooling, Its Theory and Application," PhD Thesis, Jet Propulsion Center, Purdue University, Lafayette, Indiana, January 1958.

- 23. Sellers, J. P., "Experimental and Theoretical Study of the Application of Film-Cooling to a Cylindrical Rocket Thrust Chamber," PhD Thesis, Purdue University, Lafayette, Indiana, June 1958.
- 24. Hermann, R., Leitinger, H., and Melnik, W. L., "Research on the Design of Hypersonic Nozzles at High Stagnation Temperatures-Part 1," WADC TN 55-507, Rosemount Aeronautical Laboratories, Research Report No. 127, March 1955.
- 25. Gater, R. A., "Liquid-film cooling, its physical nature and theoretical analysis," MSME Thesis, Purdue University, Lafayette, Indiana, January 1966.
- 26. Gater, R. A., L'Ecuyer, M. R., Warner, C. F., "Liquidfilm cooling, its physical nature and theoretical analysis," Jet Propulsion Center, Purdue University, Report No. TM-65-6, October 1965.
- 27. Luikov, A. V., "Heat and Mass Transfer in Evaporation," International Chemical Engineering, April 1963.
- 28. Crabtree, D. L., "An experimental investigation of the velocity and temperature profiles in the boundary layer above an unstable evaporating film of liquid under the influence of a moving stream cf hot gas," PhD Thesis, Purdue University, Lafayette, Indiana, January 1967.
- 29. Graham, A. R., and Zucrow, M. J., "Some Considerations of Film Cooling for Rocket Motors," Jet Propulsion, June 1957.

しょうかいろいろ いい かったか な茶の手を着せるなますの

- 30. Graham, A. R., "Film Cooling of Rocket Motors, Part I. An Experimental and Theoretical Investigation of Film Cooling of Rocket Motors," PhD Thesis, Jet Propulsion Center, Purdue University, Lafayette, Indiana, January 1958.
- 31. Hall Taylor, N., Hewitt, G. F., and Lacey, P. M. C., "The Motion and Frequency of Large Disturbance Waves in Annular Two-Phase Flow of Air-Water Mixtures," Chemical Engineering Science, <u>18</u>, 537-552, (1963).
- 32. Chien, S., and Ibele, M., "Pressure Drop and Liquid Film Thickness of Two-Phase Annular and Annular-Mist Flows," Trans. ASME, Ser. C.; J. Heat Transfer <u>86</u>, 88-96, (1964).

ł١

- 33. Cohen, L. S., and Hanratty, T. J., "Generation of Waves in the Concurrent Flow of Air and a Liquid," AIChE Journal, <u>11</u>, 138, (1965).
- 34. Craik, A. D. D., "Wind Generated Waves in Liquid Films," PhD Dissertation, University of Cambridge, 1965.
- 35. Hanratty, T. J., and Engen, J. M., "Interaction Between a Turbulent Air Stream and a Moving Water Surface," AIChE Journal, September 1957.
- 36. Kays, W. M., <u>Convective Heat and Mass Transfer</u>, McGraw-Hill, Inc., New York, 1966.
- 37. Steen, D. A., and Wallis, G. B., "The Transition from Annular to Annular-Mist, Co-Current Two-Phase Downflow," Report, No NYO-3114-2, Dartmouth College, 1964.
- 38. Steen, D. A., MS Thesis, Dartmouth College, 1964.
- 39. Hinze, J. O., <u>Turbulence</u>, McGraw-Hill, Inc., New York, 1959.
- 40. Eckert, E. R. G., and Drake, R. M., Jr., <u>Heat and Mass</u> <u>Transfer</u>, McGraw-Hill, Inc., New York, 1959.
- 41. Keenan, J. H., and Keyes, F. G., <u>Thermodynamic</u> <u>Properties of Steam</u>, First Edition, John Wiley and Sons, Inc., New York, 1936.
- 42. Rannie, W. D., "A Simplified Theory of Porous Wall Cooling," Progress Report No. 4-50 Jet Propulsion Laboratory, California Institute of Technology, Pasadena, California, November 1947.
- 43. Rannie, J. B., "The Influence of Selected Gas Stream Parameters on the Film Cooling Requirements of Rocket Motor Combustion Chambers, "MSME Thesis, Purdue University, Lafayette, Indiana, 1962.
- 44. Humble, L. V., Lowdermilk, W. H., and Grele, M. D., "Heat Transfer Coefficients and Friction Factors for Air Flowing in a Tube at High Surface Temperatures," ASME Heat Transfer and Fluid Mechanics Institute, 1949, pp. 165-180.
- 45. Guinn, G. R., "The Influence of Dual Injection Slots on the Film Cooling of Rocket Motor Combustion Chambers," MSME Thesis, Purdue University, Lafayette, Indiana, January 1963.

- 46. Hatch, J. E., and Papell, S. S., "Use of a Theoretical Flow Model to Correlate Data for Film Cooling or Heating an Adiabatic Wall by Tangential Injection of Gases of Different Fluid Properties," NASA TN D-130, November 1959.
- 47. Persh, J., and Lee, R., "A Method for Calculating Turbulent Boundary Layer Development in Supersonic and Hypersonic Nozzles Including the Effects of Heat Transfer," U. S. Naval Ordnance Lab. Report 4200, 1956.
- 48. Schlichting, H., <u>Boundary Layer Theory</u>, Fourth Edition, McGraw-Hill, Inc., New York, 1960.
- 49. Bartz, D. R., "An Approximate Solution of Compressible Turbulent Boundary-Layer Development and Convective Heat Transfer in Convergent-Divergent Nozzles," ASME Trans., November 1955.
- 50. Elliot, D. G., Bartz, D. R., and Silver, S., "Calculation of Turbulent Boundary-Layer Growth and Heat Transfer in Axi-Symmetric Nozzles," Jet Propulsion Lab. TR 32-387, February 1963.
- 51. Librizzi and Cresci, R. J., "Transpiration Cooling of a Turbulent Boundary Layer in an Axisymmetric Nozzle," AIAA Journal, April 1964.
- 52. Libby, P. A., and Cresci, R. J., "Experimental Investigation of the Downstream Influences of Stagnation Point Mass Transfer," Journal Acrospace Sciences, January 1961.

enteres... subservices and the

- 53. Seban, R. A., "Heat Transfer and Effectiveness for a Turbulent Boundary Layer with Tangential Fluid Injection," ASME Paper No. 59-A-177 (1960).
- 54. Seban, R. A., and Back, L. H., "Effectiveness and Heat Transfer for a Turbulent Boundary Layer with Tangential Injection and Variable Free-Stream Velocity," Journal Heat Transfer, August 1962.
- 55. Hartnett, J. P., Birkebak, R. C., and Eckert, E. R. G., "Velocity Distributions, Temperature Distributions. Effectiveness and Heat Transfer for Air Injected Through a Tangential Slot Into a Turbulent Boundary Layer," Journal Heat Transfer, August 1961.
- 56. McAdams, W. H., <u>Heat Transmission</u>, McGraw-Hill, Inc., New York, 1954.

- 57. Maxwell, J. B., <u>Data Book on Hydrocarbons</u>, D. Van Nostrand Co., 1950.
- 58. <u>Handbook of Chemistry and Physics</u>, 41st Edition, 1959-1960, Chemical Rubber Publishing Co.
- 59. Reid, R. C., and Sherwood, T. K., <u>The Properties</u> of Gases and Liquids, McGraw-Hill, Inc., New York, 1958.
- 60. <u>Liquid Propellants Handbook</u>, Battelle Memorial Institute.
- 61. Freon, Technical Literature File for Contractors and Dealers, Freon Products Division, E. I. Du Pont de Nemours & Co. (Inc.).

-----

١.

1

-

D

1

[]

# NOTATION

a	acoustic speed (Eq. (3-69))
Α	constant (see Eq. (3-1))
b	constant (Eq. (3-63))
B	constant (see Eq. (3-2))
B <sub>h</sub>	heat-mass transfer parameter (Eq. (3-22))
С	mass concentration of vapor
° <sub>f</sub>	skin friction coefficient (Eq. (3-20))
с <sub>р</sub>	specific heat at constant pressure
c <sub>v</sub>	specific heat at constant volume
d	constant (Eq. (3-73))
D	tube diameter
e <sub>o</sub>	entrainment parameter (see Eq. (3-5))
f	denotes a function
F	correction factor (see Eq. (3-16))
G	mass velocity for gas stream (Eq. (2-1))
h	specific enthalpy
h	heat transfer coefficient (Eq. (1-2))
H <sub>v</sub>	latent heat of vaporization
k	thermal conductivity
к	dimensional constant (Eq. (3-41))

1	liquid-film cooled length
10	10-inch long film cooled length
۱ s	measure of scale of waves on surface of liquid film
m	rate of coolant injection or withdrawal
m '	rate of mass transfer from liquid film (see Eqs (3-2) and (3-6))
m "	local rate of mass transfer from liquid film
<b>m</b> "	rate of gas flow through the boundary layer per unit spanwise length (Eq. (5-6))
M	molecular weight
Ma	Mach number
Mo	Momentum parameter (Eq. (2-2))
n	parameter in the power-law velocity profile (Eq. (5-3))
Nu	Nusselt number (Eq. (4-1))
p	static pressure
p <sup>O</sup>	stagnation pressure
р <sub>v</sub>	partial pressure for vapor
Pr	Prandtl number = $\frac{\mu c_p}{k}$
q	energy transfer due to conduction
۹ <sub>r</sub>	energy transfer due to radiation
r	roughness parameter (see Eq. (3-4))
R	perfect gas constant
ReD	Reynolds number based on D
Rex	Reynolds number based on x
Ret	see Eq. (3-65)

1

ļ

1

1:

**·**[]

F

l

[

Ŧ

Re' Xt	see Eq. (3-66)
Re <sub>0</sub>	Reynolds number based on 0 (Eq. (5-21))
St <sub>s</sub>	corrected Stanton number for simple heat transfer (Eq. (3-15))
St <sub>so</sub>	uncorrected Stanton number for simple heat transfer (Eq. (3-23))
St's	Stanton number for simple mass transfer (Eqs. (3-85) and (3-86))
т	static temperature
то	stagnation temperature
u	streamwise component of velocity
W	weighting parameter (Eq. (4-2))
We	Weber number (Eq. (3-38))
x	distance measured from effective starting point of velocity boundary layer
x'	distance measured from point of liquid injection
×t	thermal entrance length
Х <sub>е</sub>	entrainment group (Eq. (3-31) with a = ½)
× <sub>r</sub>	roughness group (Eq. (3-45))
У	distance normal to the wall or the liquid film surface
Greek	
Y	specific heat ratio
δ	dynamic boundary layer thickness
ót	thermal boundary layer thickness
δ*	displacement thickness (Eq. (5-14))

- θ momentum thickness (Eq. (5-15))
- μ dynamic viscosity

214

**Manufacture** 

Ţ.

]

]

a nation

HE MANADA -

π	dimensionless entrainment parameter (Eq. (3-34))
ρ	density
σ	surface tension
φ	energy balance parameter (Eqs. (3-13) and (3-14))
Subscript	5
1	evaluated at the plane of liquid coolant injection
2	evaluated at the terminus of the liquid-film cooled length
3	evaluated at the terminus of the gas-vapor cooled length
aw	adiabatic wall
c	<b>critical</b>
e	effective
g	main-stream state (y = δ)
g	main-stream gas
gv	gas-vapor cooling
k	summation index
1	liquid
L	evaluated at the gas-liquid interface but in the liquid phase
m	maximum
r	reference
S	evaluated at the gas-liquid interface
S	simple
S	evaluated at the gas-liquid interface but in the gas phase
v	vapor

Ī

I

I

Ī

Y

مورية الإستاسيس يساله ترارا

w wall

Superscripts

o stagnation conditions

#### APPENDIX A

#### AN ANALYSIS FOR THE GAS-VAPOR-COOLED REGION

#### A.1 Introduction

It is difficult, if not impossible, to make a complete mathematical analysis of the boundary layer and heat transfer characteristics of the region immediately downstream from the end of the liquid film. Moreover, an involved analysis would not be generally warranted because of the large number of assumptions that must be made concerning the velocity and temperature profiles, the form factor or factors, the skin-friction law, the heat-transfer law, etc. As a consequence of this fact, the analysis presented herein is based on an idealized and relatively simple flow model. It will be shown that the analysis cannot be completely developed until more definitive experimental data are obtained for the subject problem. However, the limited data that do exist show that the analysis presented herein has considerable promise. These data are discussed in Subsection A.2.4.

Both Emmons (20) and Guinn (45) developed semiempirical analyses for the gas-vapor-cooled region; these

analyses were based on the flow model suggested by Hatch and Papell (46) for gas-film cooling. The analysis presented here differs from those due to Emmons and to Guinn in the following fundamental respects:

(a) boundary-layer type flow rather than fully-developedpipe low is considered;

(b) the analysis is generalized so that it can be applied to a compressible gas flow in the presence of a streamwise pressure gradient; and

(c) no direct similarity between gas-vapor and gas-film cooling is assumed.

As discussed in Subsection 2.1.2, an analysis for the gas-vapor-cooled region requires the consideration of two cases: (a) the case where the wall is adiabatic; and (b) the case where the wall is externally cooled. These two cases are considered in Sections A.2 and A.3, respectively.

A.2 The Case Where the Wall is Adiabatic

A.2.1 The Flow Model

The flow model employed for the analysis of the gasvapor-cooled region is defined by the following conditions. (1) The wall is adiabatic.

(2) The boundary layer flow is fully turbulent.

(3) The thermal, the diffusional, and the dynamic boundary layer thicknesses are equal; the latter is denoted by  $y = \delta$ .

(4) The stagnation temperature for the main stream,  $T_g^0$ , is a constant in the streamwise direction.

(5) The specific heat for the gas,  $c_p$ , is a constant.

(6) Radiation heat transfer and chemical reactions within the boundary layer are not important.

(7) The velocity profile and the stagnation-temperature profile for the boundary layer are of the function forms:

$$\frac{u}{u_g} = f_1(\frac{y}{\delta}) \tag{A-1}$$

and

$$\frac{T^{0} - T_{aw}}{T_{g}^{0} - T_{aw}} = f_{2}(\frac{y}{\delta})$$
 (A-2)

where  $T_{aw}$  is the temperature of the fluid adjacent to the adiabatic wall. To illustrate,  $f_1$  could be assumed of the power-law form:

$$f_1 = \left(\frac{y}{\delta}\right)^{1/n} \tag{A-3}$$

and  $f_2$  could be assumed of the form

$$f_2 = A + B(\frac{y}{\delta}) + C(\frac{y}{\delta})^2 + D(\frac{y}{\delta})^3 \qquad (A-4)$$

1000

The constants in Eq. (A-4) can be evaluated from the

following boundary conditions that apply to f<sub>2</sub>. Thus

 $f_{2}(0) = 0$   $f_{2}(1) = 1$   $\frac{\partial f_{2}}{\partial y}(0) = 0$   $\frac{\partial f_{2}}{\partial y}(1) = 0$ 

The resultant expression for f<sub>2</sub> is

$$f_2 = 3\left(\frac{y}{\delta}\right)^2 - 2\left(\frac{y}{\delta}\right)^3$$
 (A-5)

For the present analysis, however, the actual forms of  $f_1$ and  $f_2$  <u>need not be assumed</u>, only the assumption that  $f_1$  and  $f_2$  exist for the gas-vapor-cooled region is required.

## A.2.2 The Energy Balance

Figure A.1 presents, schematically, a control volume for the gas-vapor-cooled region; the control surface of the control volume is defined by  $x = x_2$ , x = x,  $y = \delta$ , and y = o. The following energy fluxes, written in accordance with the flow model defined in the foregoing section, are presented in the figure:

(a)  $\int_{0}^{\delta_{2}} \rho u c_{p} T^{0} dy =$  the rate at which thermal and kinetic energy is transferred into the control volume per unit spanwise length at the station x = x<sub>2</sub>.

.



-

(b) 
$$\int_{0}^{\delta} \rho u c_{p} T^{0} dy =$$
 the rate at which thermal and kinetic energy is transferred out of the control volume per unit spanwise length at the station  $x = x$ .  
(c)  $(m'' - m_{2}'') c_{p} T_{g}^{0} =$  the rate at which thermal and kinetic energy is transferred from the main stream into the control volume per unit spanwise length.

The parameter m" in term (c) is defined as follows:

$$m'' = \int_{0}^{\delta} \rho u \, dy = \text{ the rate at which mass is trans-} \\ \text{ferred through the boundary layer} \\ \text{per unit spanwise length at the} \\ \text{station } x = x$$

(A-6)

-

I

The energy balance for the control volume in Fig. A.1 can be written as follows. Thus

1

$$\int_{0}^{\delta_{2}} \rho u c_{p} T^{0} dy + (m^{"} - m_{2}^{"}) c_{p} T_{g}^{0} = \int_{0}^{\delta} \rho u c_{p} T^{0} dy$$
(A-7)

Employing Eq. (A-6), and dividing through by  $c_p$ , Eq. (A-7) can be written in the form

$$m_{2}^{\sigma} = \int_{0}^{\delta_{2}} \rho u T^{0} dy$$

$$m_{2}^{\sigma} = \int_{0}^{\delta_{2}} \rho u dy$$

$$\int_{0}^{\delta_{2}} \rho u dy$$

$$(A-10)$$

In order to realize the desired simplicity of the present analysis, the following approximation is introduced concerning the integral terms in Eq. (A-10). Thus\*

$$\int_{0}^{\delta} \rho u T^{0} dy \qquad \int_{2}^{\delta} u T^{0} dy$$
$$\int_{0}^{0} \rho u dy \qquad \int_{0}^{\delta} u dy$$

1

1

E

Thus, Eq. (A-10) can be written in the following form by introducing the above approximation and assumption (7) of the flow model:

\*Note that this approximation is not equivalent to and is less restrictive than the assumption of constant density.



The latter expression can be rewritten in the form

$$m_{2}^{"}(T_{g}^{0} - T_{aw,2}) - m_{2}^{"}(T_{g}^{0} - T_{aw,2}) \frac{\int_{0}^{1} f_{1} f_{2} d(\frac{y}{\delta})}{\int_{0}^{1} f_{1} d(\frac{y}{\delta})} =$$

$$m^{"}(T_{g}^{0} - T_{aw}) - m^{"}(T_{g}^{0} - T_{aw}) \frac{\int_{0}^{1} f_{1} f_{2} d(\frac{y}{\delta})}{\int_{0}^{1} f_{1} d(\frac{y}{\delta})}$$

(A-11)

1

Finally, the terms in Eq. (A-11) can be combined to yield the following energy balance for the control volume presented in Fig. A.1. Thus

$$\frac{T_{g}^{0} - T_{aw}}{T_{g}^{0} - T_{aw}, 2} = \frac{m_{z}^{"}}{m"}$$
(A-12)

The relatively simple form of Eq. (A-12) is due principally to assumption (7) of the flow model.

Equation (A-12) enables the determination of the adiabatic-wall temperature  $T_{aw}$  as a function of x once  $T_{aw,2}$ and  $T_g^0$  are specified and m" as a function of x is known. The parameter  $T_g^0$  is normally known from the statement of the problem, and the parameter  $T_{aw,2}$  can be estimated from the developments presented in Subsection 3.4.1. The evaluation of m" as a function of x is considered in Subsection A.2.3.

A.2.3 The Evaluation of m"

Equation (A-6), which defines the parameter m", can be rewritten in the form

$$m^{*} = (\rho u)_{g} \left(\frac{\delta}{\theta} - \frac{\delta^{*}}{\theta}\right) \theta \qquad (A-13)$$

where

I

New York

ľ

 $\delta^* = \int_{0}^{\delta} (1 - \frac{\rho u}{(\rho u)_g}) dy = \text{the displacement thickness}$ 

(A-14)

$$\theta = \int_{0}^{\infty} \frac{2u}{(\rho u)_g} (1 - \frac{u}{u_g}) dy = \text{the momentum thickness}$$

(A-15)

Equation (A-13) shows that m" can be determined as a function of x if  $\frac{\delta}{\theta}$ ,  $\frac{\delta}{\theta}$ , and 0 are known as functions of x. The evaluation of these quantities is considered in the following paragraphs.

<u>The Evaluation of  $\delta/\theta$  and  $\delta^*/\theta$ </u>. Persh and Lee (47) have tabulated the values of  $\delta/\theta$  and  $\delta^*/\theta$  as functions of the main-stream Mach number and the wall-to-main-stream temperature ratio. It is suggested, therefore, that these tabulated data be employed for evaluating  $\delta/\theta$  and  $\delta^*/\theta$  when the variation in the density for the gas stream across the boundary layer is significant. However, for the more general case where that density variation is not significant, it is suggested that the subject parameters be calculated by applying the power-law velocity profile (see Eq. (A-5)). The resultant expressions for  $\delta/\theta$  and  $\delta^*/\theta$  are (48, p. 536)

$$\frac{\delta}{\theta} = \frac{(1+n)(2+n)}{n}$$
 (A-16)

and

$$\frac{\delta^{*}}{\Theta} = \frac{(2+n)}{n}$$
 (A-17)

<u>The Evaluation of  $\theta$ </u>. To determine the value of the momentum thickness  $\theta$  as a function of x, one must solve the momentum integral equation. That equation can be written in the form (49)

$$\frac{c_f}{2} = \frac{d\theta}{dx} + \theta \left[ \left( 2 + \frac{\delta}{\theta}^* \right) \frac{1}{u_g} \frac{du_g}{dx} + \frac{1}{\rho_g} \frac{d\rho_g}{dx} \right]$$
 (A-18)

Or in the form (50)

$$\frac{c_{f}}{2} = \frac{d\theta}{dx} + \theta \left[\frac{1}{M_{a}} \frac{dM_{a}}{dx} + \frac{\left\{\left(\delta^{*}/\theta\right) + 2 - M_{a}^{2}\right\}}{\left\{1 + \left(\frac{\gamma - 1}{2}\right) M_{a}^{2}\right\}}\right]$$
(A-19)

where

$$\frac{c_f}{2} = \frac{\tau_w}{\rho_g u_g^2} = \text{the skin-friction coefficient} \qquad (A-20)$$

1.1.1.1.1.1.1

To solve\* either Eq. (A-18) or Eq. (A-19) for  $\theta$ , one must specify the form of the skin-friction law; the latter is denoted functionally by

$$\frac{c_f}{2} = \frac{c_f}{2} (Re_{\theta})$$

\*A numerical solution is generally required for either Eq. (A-18) or (A-19). where

$$Re_{\theta} = \frac{\rho_g u_g \theta}{\mu_g} = the Reynolds number for the mainstream based on the characteristiclength  $\theta$$$

Several empirical skin-friction laws are presented in Chapter 22 of Ref. (48) for the case where the gas flow is turbulent, incompressible, and over a flat plate. In the analysis of gas flows with favorable pressure gradients, it is normally assumed that the skin-friction law is the same as that established empirically for the flatplate case (49, 50, 48). Furthermore, the case where heat transfer effects and/or compressibility effects are important can be analyzed by evaluating the thermodynamic properties for the gas stream at an appropriate reference temperature. For example, the reference temperature, denoted by  $T_r$ , can be determined from (36):

$$T_r = 0.5(T_g + T_w) + (Pr)^{1/3} \frac{u_g^2}{2c_{pq}}$$
 (A-21)

It is suggested that in the evaluation of the physical properties that the composition of the gas be assumed uniform throughout the boundary layer and equal to that for the main stream; i.e., assume that a reference temperature is sufficient to define the reference state.

To complete the solution for  $\theta$ , the boundary condition

on  $\theta$  at the station x = x<sub>2</sub>, denoted by  $\theta_2$ , must be specified. An approximate technique for the evaluation of that boundary condition was proposed in Ref. (25). However, subsequent analysis of the available experimental data suggests that that proposal is not satisfactory. In order to illustrate that point, an analysis of the data that are presented in Ref. (19) is discussed in Subsection A.2.5. However, to facilitate that discussion, it is convenient to first consider a special case for which the foregoing developments are simplified substantially.

### A.2.4 A Special Case

Consider the special case that is defined by the following conditions.

Į

F.

- The gas flow is incompressible, constant density, and over an adiabatic flat plate.
- The skin-friction law is that due to Prandtl (48, Eq. 22.7). Thus

$$\frac{c_f}{2} = 0.0128 \text{ Re}_{\theta}^{-\frac{1}{4}}$$
 (A-22)

 The velocity profile for the boundary layer is given by Eq. (A-3). Thus

$$\frac{u}{u_q} = \left(\frac{y}{\delta}\right)^{1/n}$$
 (A-3)

For this special case, Eq. (A-12) reduces to the form

$$\frac{T_{g} - T_{aw}}{T_{g} - T_{aw}, 2} = \frac{m_{Z}^{"}}{m_{T}^{"}}$$
(A-23)

Equation (A-13) reduces to (refer to Eqs. (A-16) and (A-17))

$$m'' = (r - n) \rho u_g \theta \qquad (A-24)$$

Equation (A-18) simplifies to the form

$$\frac{c_{f}}{2} = \frac{d\theta}{dx} = \frac{dRe_{\theta}}{dRe_{x}}$$
(A-25)

where

$$Re_{\chi} = \frac{\rho u_{g} x}{\mu} = the Reynolds number for the mainstream based on the characteristiclength x$$

(A-26)

\*\*\*\*

Note that due to assumption (1), the distance x is measured from the "effective" starting point for the growth of the turbulent boundary layer over the gas-vapor cooled region. It follows from the discussion presented below that the specification of that starting point is equivalent to the specification of the parameter  $\theta_2$ . Upon combining Eqs. (A-23) and (A-24), one obtains

$$\frac{T_{g} - T_{aw}}{T_{g} - T_{aw}, 2} = \frac{\theta_{2}}{\theta}$$
(A-27)

In addition, substitution of Eq. (A-22) into Eq. (A-25) yields

I

I

E

2

$$0.0128 \text{ dRe}_{x} = \text{Re}_{\theta}^{\frac{1}{4}} \text{ dRe}_{\theta}$$
 (A-28)

Integrating Eq. (A-28) and substituting the limit  $\text{Re}_{\theta} = \text{Re}_{\theta}_2$ at x = x<sub>2</sub>, one can readily show that

$$\frac{\operatorname{Re}_{\theta}}{\operatorname{Re}_{\theta}} = \frac{\theta}{\theta} = \left(\frac{x_2}{x}\right)^{4/5}$$
(A-29)

Substituting Eq. (A-29) into Eq. (A-27) results in

$$\frac{T_{g} - T_{aw}}{T_{g} - T_{aw}, 2} = \left(\frac{x_{2}}{x}\right)^{4/5}$$
(A-30)

Equation (A-30) presents, for the subject special case, a simple expression for the calculation of the adiabatic-wall temperature  $T_{aw}$  as a function of the distance from the end of the liquid film (x -  $x_2$ ). The problem, therefore, is to determine  $x_2$ . The data of Ref. (19) give some insight into the problem of the evaluation of that parameter. Those data are considered in the following subsection.

# A.2.5 Experimental Data for the Adiabatic-Wall Temperature From Reference (19)

Kinney, Abramson, and Sloop (19) present some data for the temperature for an adiabatic wall immediately downstream of a liquid-film cooled wall region\*. To the author's knowledge, these are the only such data that appear in the published literature. The experimental apparatus that was employed in the subject investigation was discussed previously in Subsection 4.3.2. The reported data were obtained with the 4-inch diameter smooth surface tube at a nominal gas stream temperature of 1400F and a nominal gas stream Reynolds number of 5.3 x  $10^5$ . The length of the liquid-film cooled region was approximately 19.0 inches.

To obtain an evaluation of the present analysis, it is assumed that Eq. (A-30) that was developed for the aforementioned special case is applicable to the subject data. The determination of the value for  $x_2$  (and thus the effective starting point for the growth of the boundary layer over the gas-vapor cooled region) for these data gives some insight into the phenomena involved.

Figure A.2 presents a plot of the left-hand side of Eq. (A-30) as a function of  $(x - x_2)$ , the distance measured from the end of the liquid film. The circles shown in the figure were calculated from the data presented in Fig. 4 of Ref. (19). The solid curves shown in Fig. A.2 were calculated \*See Figure 4 of Ref. (19).



Ĩ

ſ

ſ

E

ľ

F

ľ

ĺ

いたたいという

111



. 233
from Eq. (A-30) with values for  $x_2$  of 0.5, 0.8, and 1.0 inches. For the wall region immediately downstream of the liquid film (say,  $x - x_2 < 10.0$  inches), Fig. A.2 shows that a value of 0.8 inches for  $x_2$  results in an excellent correlation of the experimental data. The small value of 0.8 inches for  $x_2$ , however, is difficult to interpret physically. This small value suggests that the boundary layer phenomena are complex, with perhaps a considerable change in the boundary layer structure occurring at (or very near) the terminus of the liquid film. The basic analysis, however, shows considerable promise. Even the complex case where a compressible gas is strongly accelerated over the gas-vapor cooled region can be reduced to the problem of determining the "effective starting point (x = 0) for the growth of the boundary layer over the subject wall region. The task that remains, therefore, is to obtain experimental data on the adiabatic wall temperature for a wide range of flow conditions so that the effective starting point can be correlated in terms of the pertinent variables. Unfortunately, since such data are not now available, the analysis proposed herein for the gas-vapor cooled region cannot be completely formulated.

A.3 The Case Where the Wall is Cooled Externally

For the subject case, the basic analytical problem is that of relating the local values for the wall temperature and the wall heat flux to the main-stream conditions. It can be inferred from several mass transfer cooling (and heating) studies that are reported in the literature (e.g., Refs. 51-55) that heat-transfer data for the gas-vapor cooled region should correlate satisfactorily in the following manner. Thus

$$St_{av} = St_{o}$$

where

C +			۹ <sub>W</sub>		
Stgv	-	G	c <sub>pg</sub> (T <sub>aw</sub>	- T <sub>w</sub> )	-

(A-32)

(A-31)

- Stgv = the Stanton number for heat transfer for the
  gas-vapor cooled region
- St = the Stanton number for heat transfer in the absence of gas-vapor cooling, assuming that the heat transfer begins at the leading edge of the gas-vapor cooled region
- T<sub>aw</sub> = the temperature the wall would obtain if it was adiabatic
- T<sub>w</sub> = the temperature the wall obtains due to the fact that it is cooled externally
- $q_W =$ the heat flux from the hot gas stream to the external coolant

Equation (A-31) reduces the present problem to that of determining the Stanton number St<sub>o</sub> as a function of the distance measured Jownstream from the end of the liquid film. The theoretical methods that were outlined in Subsection 3.1.2 and in Subsection 3.3.1 can be employed to calculate that Stanton number\*. In that calculation, however, it is important to note that the following boundary condition applies (51-55)\*\*:

 $St_0 = \infty at x = x_2$ 

The task remains to obtain experimental data for the rate of heat transfer to an externally cooled surface that is located immediately downstream of the terminus of the liquid film so that the validity of Eq. (A-31) can be determined. Although Emmons (20) and Guinn (45) did obtain some experimental data on that rate of heat transfer, the validity of Eq. (A-31) cannot be investigated with those data because neither Emmons nor Guinn determined the adiabatic wall temperature  $T_{aw}$  that enters into the equation. There is, however, an important observation that can be made with reference to the experimental results due to Emmons. That observation is discussed in the following subsection.

\* With  $F_{C} = 1.0$  and  $F_{m''} = 1.0$ 

\*\*Guinn (45) developed a semi-empirical analysis for the gas-vapor cooled region. Guinn, in essence, utilized Eq. (A-31) in that analysis. However, he did not recognize that the above boundary condition applied in the evaluation of the Stanton number St<sub>0</sub>. As a consequence, his analysis is not valid at the leading edge of the gas-vapor couled region.

## A.4 <u>An Observation on the Data of Emmons</u> for Gas-Vapor Cooling

A STATE

THE REAL

Emmons (20) investigated experimentally the rate of heat transfer to an externally cooled test section that was located immediately downstream of the liquid-film cooled test section. Those heat transfer rates, which were determined calormetrically at several locations downstream of the film cooled region, were compared with the heat transfer rates that were realized for the aforementioned 1.48-inch convectively cooled test section that was located immediately upstream of the liquid film\*. As a result of that comparison, Emmons drew the following two conclusions (p. 67, Ref. (20)):

"A large portion of the film coolant vapor forms a mass of cool gas which remains adjacent to the wall for a considerable distance downstream from the liquid film. As a result the heat flux to the wall protected by the vapor is significantly reduced;" and

"Heat transfer rates measured approximately 9.0 in. downstream from the liquid film were reduced by 30 to 70 percent from the corresponding values measured upstream from the liquid film. Ammonia, with the lowest boiling temperature (160F) effected the largest heat transfer reductions for the coolants investigated."

These two conclusions imply that the rate of mixing of \*See Fig. 42.

3 (A 🚛

t.

the relatively cool boundary layer gases with the mainstream gas is slow. However, the data of Ref. (19) that were discussed in Subsection A.2.5 for the adiabatic wall temperature suggest that the rate of mixing is quite rapid. That is implied by the small value for  $x_2$  (or  $\theta_2$ ) that was found to result in a satisfactory correlation of the experimental That small value for  $x_2$  suggests that the rate of data. entrainment of the hot mainstream gas into the boundary layer (or the rate of growth of the boundary layer) is very rapid for the subject wall region. The data presented in Fig. A.2 show that at a point approximately 10 inches downstream of the liquid film, the cool boundary layer gas and the hot mainstream gas are essentially completely mixed. There appears to exist, therefore, some inconsistency in the aforementioned conclusions that were drawn by Emmons and the experimental data that were reported in Ref. (19). However, an explanation for that inconsistency can be suggested: It was argued in Subsection 4.4.5 that the measured rates of heat transfer to the 1.48-inch convectively cooled test section were apparently about 40 percent higher than those that would be predicted for fully-developed tube flow at the same bulk flow conditions. Moreover, it was argued that part (or perhaps all) of that difference could have been due to a discontinuous thermal boundary condition at the leading edge of the 1.48-inch test section. For the convectively cooled test section immediately downstream of the end of the

liquid film, however, there would be a continuous transition in the wall temperature from the end of the film cooled test section to the gas-vapor conled region. Therefore, because of the fundamental difference between the thermal boundary conditions for those two test sections, it would appear that the aforementioned conclusions that were drawn from the direct comparison of the heat transfer rates for the two test sections may not have been completely valid. If, in fact, an isothermal-wall boundary condition for the 1.48inch long test section would have resulted in heat transfer rates that were 40 percent lower than those that were actually realized, then the comparison of the experimental data for the two convectively cooled test sections would have showed that the heat transfer rates measured approximately 9.0 downstream of the liquid film would have been reduced by 0 to 20 percent from the corresponding values measured up tream from the liquid film, as compared to the 30 to 70 percent reduction that was quoted above from Emmons (20). That particular result would appear to be in agreement with the data that were discussed in Subsection A.2.5. However. it cannot be argued conclusively that the significant conclusions that were drawn by Emmons with respect to the subject wall region are in error. That question cannot be completely answered until considerable more experimental data are obtained under well controlled experimental conditions for the gas-vapor cooled wall region. Perhaps the

Ì

E

:

foregoing discussion will serve as an impetus for that research.

•

Marcure.

1

ţ

J

J

#### APPENDIX B

#### DESCRIPTION OF APPARATUS

#### **B.1** General Description

The gas source employed in the subject investigation was an air blowdown wind tunnel in which a portion of the air flow was heated by combustion with a liquid fuel. The principal features of the apparatus were that it had a wide range of operating conditions, i.e., flow rate, gas temperature and pressure, reproducibility of operating conditions from run to run, low drift about the operating point over run times up to one hour, and was safe and reliable enough to work with in the test cell during operation of the apparatus.

ž,

Figure B.1 presents a photograph of the basic configuration of the experimental apparatus for that phase of the investigation where the gas stream velocity was essentially constant over the length of the wetted test plate. Figure B.2 presents a similar photograph of the basic configuration that was employed in that phase of the investigation where the gas stream was accelerated over the wetted test plate. The various components of the apparatus are described as follows. The numbers refer to either Fig. B.1



l-control valves 2-primary combustor 3-secondary mixer 4-mixing section 5-approach section 6-test section 7-nozzle 8-camera 9-strobe 12-timer 16-liquid flow meter 17-air flow meter 19-milliampmeter

I

I

Ĩ

Q

Û

Ū

B

U

B

I

Ţ

J

242

Fig. B.l. Photograph of Experimental Apparatus. Configuration for Tests With Constant Velocity Gas Stream



Configuration Photograph of Experimental Apparatus. for Tests With Accelerating Gas Stream Fig. 8.2.

243

Ì

ļ

I

I

J

Ĩ

I

I

I

1

Į

Į

€:

or Fig. B.2.

、おいいのなやいない

(1) The air and fuel flow rate control valves. A separate valve was utilized to control (a) the air flow rate to the primary combustor, (b) the air flow rate to the secondary mixer, (c) the fuel flow rate to the primary combustor for ignition, and (d) the fuel flow rate to the primary combustor during full-scale operation.

(2) The primary combustor wherein a high-temperature combustion zone was established.

(3. The secondary mixer wherein diluent air was added to the hot gases from the primary combustor so as to obtain the desired gas stream temperature in the experimental tunnel.

(4) The 10-inch long mixing section which contained one 10-mesh screen and one 30-mesh screen that were spaced 10 inches apart, and which reduced the non-homogeneities of the gas stream flow substantially.

(5) The 36-inch long adiabatic approach section.

(6) The experimental test section which contained the film cooled test plate.

(7) The variable area nozzle which controlled the gas stream pressure for a given gas stream Reynolds number.
(8) The 35mm Nikkorex camera that was employed to photograph the surface characteristics of the liquid film. The camera lens was a Micro-Nikkar f55mm lens with a close-copy adapter.

245

(9) The stroboscopic light source that was employed in conjunction with the camera (8). The strobe had a flash duration of 3 microseconds.

(10) The cyclone separator that was employed to separate the liquid and the gas phases that were withdrawn at the capture slot. The details of the separator design is given in Appendix B.4.

(11) The vent values that controlled the rate at which the gas phase was vented to the surroundings (and, therefore, the rate at which the gas phase was withdrawn at the capture slot).

(12) The electronic timer that was employed to measure the sample time for the mass transfer measurements.

(13) The 500 psia air solenoid valve that served as the on-off control for the capture slot. The activation of the electric solenoid valve resulted in the flow of a high pressure jet of air in the withdrawal passage in the reverse direction through the capture slot into the test section, thereby stopping the capture of the liquid film on the test plate. This solenoid valve and the electric timer (12) were controlled simultaneously so that the length of time that the gas-liquid mixture was withdrawn from the test section through the capture slot was accurately controlled. (14) The heat exchanger that was employed to preheat the liquid phase before it was introduced onto the test plate.

The heat exchanger consisted of two 3/4-in. O.D. stainless

こうそう かんかく しん 小学校学校 かった かっかい はまった 御道 ファイン しんかい ちょうせい かいがった いっかい ふん マンド・マンド・マンド・マンド・マンド・マンド・マンド・マンド・マンド

steel tubes of length 36 inches with a calrod heater located along the centerline of each. Each calrod had a maximum capacity of 1000 watts, and the electrical energy that was supplied to each was controlled by a variable reostat. (15) The mercury manometers that were employed to measure the streamwise static pressure profile for those experimental tests in which the gas flow was accelerated over the wetted test plate.

(16) The film coolant flowmeter t was employed to measure the rate at which the coolant was introduced onto the test plate. The flowmeter was a 1/8-inch Potermeter. The frequency output of the flowmeter was displayed to the operator by the milliampmeter (19) and recorded on a stripchart recorder.

(17) The air flow meter. The total rate of air flow through the experimental apparatus was measured with a Brooks turbine type flow meter (model HP-321-6). The frequency output of the flow meter was displayed on a milliampmeter in the control room and was simultaneously recorded on a stripchart recorder.

(18) The precision needle valve that was employed to control the rate of film coolant flow through the heat exchanger and then onto the test plate.

(19) Milliampmeter utilized to monitor the coolant flow rate.

.

Figure B.3 shows the control room. The primary components of the control room are as follows. The numbers refer to Fig. B.3.

(1) The automatic sequencer that was utilized to start the hot gas generator. The operation of the sequencer is discussed in greater detail in Appendix B.2.

(2) The 3 small volume Nullmatic pressure regulators that were employed to control the flow rates through the three Annin Domotor control valves: (a) the primary air control valve, (b) the secondary diluent air control valve, and (c) the primary fuel control valve for full-scale operation.

ł

き ひとす

あるとので

(3) The stripchart recorder that was employed to continuously monitor the gas stream temperature.

(4) The Bourdon tube pressure gage that was employed to continuously monitor the gas stream pressure.

(5) The variable reostat that was utilized to control the temperature for the liquid coolant at the point of injection.

(6) The milliampmeters that continuously monitored the pertinent system flow rates and system pressures. These various parameters were also recorded on stripchart recorders located in an adjoining room.

#### B.2 The Gas Generatur

Figure B.4 presents a simplified sectional drawing of the gas generator portion of the experimensal apparatus.



Ç

Fig. B.3. Photograph of Control Room



Fig. B.4. Schematic Diagram of Gas Generator

います。 まんしょう いいち いろんだいしていたまだけ 多次 いっし い

. 249

Į

Table B.1 lists the components of the gas generator and describes the function of each.

#### Table B.1

Components of the Gas Generator

Component

#### Function

Primary Injector Ic inject the fuel and air to be burned into the primary

Flameholder

ŧ

;

Primary Combustor

Secondary Injector

Perforated cone and turbulence grids

combustor. To mix the fuel and air after

injection and provide turbulence to hold the flame. To provide a contained volume

for combustion.

To inject diluent air to reduce temperature of working fluid to the desired value.

To thoroughly mix the diluent air with the combustion products from the primary combustor in order to establish a uniform temperature throughout the gas flow.

251

Table 9.2 presents the output characteristics of the gas generator for the overall allowable range of operation.

## Table B.2

Output Characteristics of the Gas Generator

Parameter	<u>Overall Range</u>
Gas Stream Pressure, psia	
Cold flow	up to 400
.Hot flow	50 - 350
Gas Stream Temperature, R	
Cold flow	500
Hot flow	750 - 1100
Gas Flow Rate, lb/sec	
Cold flow	up to 15
Hot flow	0.75 - 10.0
Fuel Flow Rate, lb/sec	0.02 - 0.25

Ŀ

E

E

E

E

ļ

I

The air flow entering the gas generator was divided such that approximately 30 percent of the total flow entered the primary combustor to be burned with the fuel while the remaining 70 percent of the air flow was injected into the secondary mixing chamber. Approximately five percent of the fuel was injected from the ignition fuel injector located in the center of the primary injector while the remaining fuel was injected fro: the main portion of the primary injector. The fuel and the primary air (representing approximately 200 percent excess air) were burned in the primary combustor at a temperature of 2500F with a thermal efficiency of approximately 50 percent. Smooth stable combustion rather than high thermal efficiency was desired. The low thermal efficency was the result of insufficient volume in the primary combustor to permit complete combustion of the fuel.

1

T - LA MANUNA

-----

1

~ = = {

-]

The primary injector comprised three principal parts, as shown in Fig. B.4, namely, the face plate, the hat section air manifold and the cap plate. Ninety-five percent of the methyl alcohol fuel was injected through two concentric rings having a total of 30 separate fuel injector buds located on the face plate (see detail in Fig. B.2). The face plate was fabricated from a circular piece of stainless steel plate ½ in. thick On each side of the fuel injectors were two 0.100-in. diameter injector orifices for air directed at 45° to the direction of the fuel flow, the combination forming a triplet. The primary air was introduced

into the injector manifold by a single 1.0-in. tube fitting. The fuel was manifolded by 4-in. O.D. stainless steel tube to six clusters of heavy wall tubing, each with tubes welded into the back of the face plate; these clusters fed fuel to the 30 fuel injectors. The ignition fuel which was required as part of the start sequence was manifolded directly to the center of the face plate and was injected through four small orifices against a copper splash ring. Air for the combustion of the ignition fuel was supplied through a ring of small orifices concentric to the ignition fuel injector orifices.

The flameholder consisted of a central perforated disc of stainless steel supported by eighteen ½-in. tubes of the same material which ran radially to an outside ring. Cooling water was introduced through a 1.0-in. tube fitting located 180° from the output fitting. The flameholder was of all welded construction. In the assembled gas generator the flameholder was sandwiched between the primary injector and the primary combustor.

The primary combustor and secondary mixer chambers were of similar design and construction, differing only in length. The former was 10 in. long and the latter was 20 in. long. The inner wall was a section of 13-in. I.D. mild steel pipe having a wall thickness of ½ in. Welded at each end of the pipe section were one-inch thick mild steel flanges with twenty-four 1-in. nominal bolt holes equispaced on a bolt

> na fatien. Anti-

circle of 20 in. The coolant passages were formed from extruded lead stripping with a 4-in. by 3/4-in. cross section which was wrapped in a spiral around and soldered to the pipe. The outer jacket of the coolant passages was formed from half-sections of 1/8-in. wall, 16-in. I.D. stainless steel tube with an expansion joint located in the middle. The jacket was sealed all around by weldments. Inlet and outlet fittings for the coolant, which was water, were welded over holes in the outer jacket and located at either end of the lead spiral. All cooled parts were incorporated in a single pass coolant path.

÷

3

]

Renver

}

The secondary air injector was fabricated from seven parallel 3/4-in. O.D. stainless steel tubes set in a 13-in. I.D. outer ring which served as a distribution manifold. Each tube was drilled with a 0.125-in. O.D. hole every 3/8 in. along its length in such a way that two rows of orifices were formed directed downstream 45° off from the direction of gas flow from the primary combustor (see detail in Figure B.2). This arrangement enhanced turbulent mixing immediately downstream from the injector tubes. The injector tubes were fed from the outer ring which in turn was fed secondary air through a 2-in. pipe fitting located on the outermost part of the ring.

Downstream of the secondary air injector was located a stainless steel cone perforated with ½-in. diameter holes. The cone pointed upstream and was clamped between the end plate and the aft flange of the secondary mixer chamber. Turbulence grids were mounted just ahead of the 2-in. x 4-in. exit port located in the center of the end plate. The perforated cone and turbulence grids induced mixing of the secondary and primary gases. The end plate was fabricated from 1-in. thick mild steel plate reinforced by gussets welded on the outside.

The flanged sections of the gas generator were bolted together with twenty-four 1-8 studbolts. The gaskets between the separate sub-assemblies were either 1/32-in. thick aluminum or 1/16-in. thick composition asbestos. Concentric serrations to aid sealing at the gaskets were machined on the injector face plate, the end plate and the flanges; no serrations were present on the flameholder or secondary air injector.

ſ

5.1

and desired and events and the second states of the second states and

<u>Operation</u>. Figure B.5 presents a simplified schematic diagram of the flow system for the experimental apparatus. The air used in the experiments was drawn from the high pressure supply tanks which comprise the blowdown air supply for the Jet Propulsion Center. The tanks were charged with a maximum pressure of 2400 psig. The compressed air was delivered to an isolating valve located on the roof of the test cell by a pipeline from the supply tanks. The isolating valve on the roof was a 2-in. 1500 ASA Annin pneumatically operated open/closed valve. It was present as a safety feature to isolate the high pressure air from the



and the second s

Fig. B.5. Schematic Flow Diagram

Ī

Ì

į

.

]

]

]

test cell housing the apparatus in the event of a severe mechanical malfunction of the gas generator.

The air supply was passed through a Fluid Dynamics high pressure air filter. The filter element removed particles from the gas stream that were larger than 10 microns. The air was then reduced from the supply line pressure by a 2-in. Grove Powereactor dome regulator located immediately downstream from the air filter. The Grove regulator was loaded by means of a small volume Grove pressure regulator. The volume flow rate of air was measured with a 2-inch Brook's turbine flowmeter located after a straight section of vipe downstream of the regulator. The weight flow rate of air was determined from the frequency output of the flowmeter and the static pressure and temperature of the air measured immediately ahead of the flowmeter. The static pressure was measured with a Wiancko static pressure transducer and the temperature was measured with an iron-constantan ther-The three aforementioned parameters were remocouple. corded during an experiment on Brown self-balancing stripchart recorders.

The air flow to the gas generator was controlled by two Annin Domotor valves, a 2-in. valve for the secondary air and a 3/4-in. valve for the primary air. Control air for these pneumatic valves as well as for other pneumatic functions in the system was supplied at pressures of 400, 100, 45, and 30 psig from either Grove Powereactor

regulators or Nullmatic small volume regulators.

The methyl alcohol fuel was stored in 55-gallon shipping drums and transferred to two 6-cu. ft. run tanks located outside the building housing the test cell. The fuel was pressurized with air by means of a small volume Grove pressure regulator. The fuel tanks were isolated from the building by a pneumatically operated Annin valve located beneath the fuel tanks. The flow rate of fuel was measured with a 1/8-in. Potter turbine flowmeter, the frequency output of which was recorded on a Brown recorder. The fuel flow rate was controlled by means of a modified 3/4-in. Annin Domotor valve which was rebuilt to control with precision the relatively small flow rate of fuel.

Before the gas generator was started in hot flow, the air pressure and fuel tank pressure were set to the desired value. Coolant water was supplied to the gas generator by a Worthington turbine pump driven by an electric motor. The three Annin Domotor valves were pre-set using Nullmatic control regulators (see Fig. B.3) to give the proper start flow rates of primary air, secondary air and fuel.

The ignition source for the gas generator was a small combustor (see Fig. B.4) which burned gaseous methane and air ignited by durl spark plugs. 1

Ŋ

P

The ignition sequence was controlled by an automatic sequencer (see Fig. B.3). The ignition sequence was as follows:

- 1. The methane-air ignitor was started.
- 2. When the ignitor reached the desired temperature, the automatic sequence was initiated.
- 3. The ignition fuel and primary air valves opened simultaneously and a pilot flame was established in the center of the primary injector off of the ignition fuel injector.
- 4. Two seconds later, the primary fuel and secondary air valves opened simultaneously.
- 5. The ignitor combustor was shut down and a second or two later the mainstage ignition was arcomplished by the pilot ignition flame.

The whole automatic start sequence required approximately six or seven seconds. Once mainstage ignition was accomplished (usually at reduced flow rates), the operator brought the gas generator up to the desired run point using the controls for the Annin Domotor valves.

During operation of the gas generator, monitoring of the various pressures associated with the system was accomplished either by Bourdon tube pressure gages or Wiancko pressure transducers. The outputs of the latter were displayed to the operator by milliampmeters\* and were also recorded on Brown stripchart recorders. Flow rates of fuel and air were also displayed to the operator by milliampmeters\* and recorded on stripchart recorders. The temperature of the gas leaving the gas generator was measured by two chromel-alumel thermocouples located immediately downstream of the end plate at the entrance to the approach section. The output of one of the thermocouples was \*Shown in Fig. B.3.

displayed to the operator on a direct reading milliampmeter while the other output was recorded on a stripchart recorder. The pressure of the gas in the lead section was the value employed to set the pressure of the run point. The gas stream pressure was measured, displayed and recorded in the manner as previously described.

After the desired operating point had been achieved, approximately ten minutes were required for the test section to achieve thermal equilibrium. Only then were the experiments begun. Luring run point operation the gas generator exhibited a random drift about the run point of  $\pm$  20F of gas stream temperature and  $\pm$  2 psi of gas stream pressure.

#### B.3 The Test Section and Test Plate

Figures 8.6 and 8.7 present sectional drawings of the experimental test section. The basic internal dimensions of the test section were 2 in. by 5.5. in. cross section by 18 in. long. The test section consisted of 4 side wal's and the wetted test plate. The side walls were made from 9/16in. thick stainless steel plate and the upper and lower walls were made from 3/4-in. thick stainless steel plate. The test plate was ½-in. thick stainless steel plate. All of the internal wall surfaces were surface ground. A single Pyrex window of dimensions 11/16 x 1-5/16 x 9-7/8 in. long was mounted in the top wall of the test section, and 6 circular 1-in. 0.D. Pyrex windows 9/16-in. thick were equispaced-at 2 inches in each side wall. All of the Pyrex

Ì

ふい、御室を読みまで

\$





Į

I

[

Ĩ

l

ſ

Ŀ

[

[]

[

ſ



and the second second second

.



Ī

J

]

]

]

]

]

]

windows were sealed by Viton-B o-rings, as indicated in the drawings.

The dashed line shown in Fig. B.6 represents the profile for the aluminum wedge for that phase of the investigation where the gas stream was accelerated over the wetted test plate.

J

ſ

Ľ

Figures B.6 and B.7 show that the test plate to which the liquid film was confined dropped into a recess that was machined into the bottom wall of the test section and "floated" on an o-ring seal. The liquid injection and liquid withdrawal manifolds are shown bolted directly to the test plate in Fig. B.6. The passage for the high-pressure air jet that was employed for on-off control of the knifeedge capture slot is shown in the withdrawal manifold. The locations of the two chromel-alumel thermocouples that were employed to measure the inlet and outlet liquid temperatures are indicated in the figure. Figure B.6 shows the porous stainless steel injector element that was constructed from a 1/8-in. thick by 2-in. long piece of "regimesh". The cross-sectional view of the knife-edge capture slot is also shown in Fig. B.6. Figures B.6 and B.7 show the 2-in. wide by 0.060-in. deep trough that was machined into the test plate to which the liquid film was confined The 2-in. by 10-in. wetted test surface was hand polished. Figure B.7 also shows the two 0.060-in. wide insulating grooves that were cut into the underside of the wetted test plate so as

to substantially reduce the lateral conduction of heat through the test plate to the liquid film.

The four side walls of the test section were bolted together with 3/8-24 stainless steel bolts. Viton-B cord was employed to seal the test section along each of the four edges. Two flange pieces were welded to each of the four test section walls so that a continuous flange surface existed at each end of the test section when the test section was assembled. The upstream approach section and the transition section between the test section and the exhaust nozzle were bolted to those flanged ends of the test section with ½-20 bolts. Extreme care was taken to align the internal flow surfaces for the test section with those for the approach section.

Figure B.8 presents two photographs of the assembled test section. Figure B.8(a) shows the configuration for that part of the investigation where the gas stream velocity was constant. The figure shows the location of the 35 mm camera that was employed to photograph the surface of the liquid film. Figure B.8(b) shows the test section configuration for the studies with an accelerated gas stream. For those experimental tests, the Pyrex window in the upper wall of the test section was removed so as to have access to the manometer leads employed to measure the static pressure along the aluminum wedge. The tubing for one of the 5 static pressure taps is shown in the figure.

1

1)

いまましい さき おうしいたい いんあい

11, 14

~ 140

I



I

8.

# (a) Constant-Velocity Gas Flow



(b) Accelerated Gas Flow

Fig. B.8. Photographs of Test Section

### B.4 <u>The Separator</u>

Figure B.9 shows a sectional drawing of the cyclone separator that was employed to separate the liquid and gas phases. The gas and liquid mixture that was withdrawn from the test section through the capture slot entered the separator at top of the main chamber in a tangential direction through a ½-in. O.D. stainless steel tube. The centrifugal forces on the liquid drops caused them to form a film on the outer wall of the separator and to eventually drain to the lower chamber. The gas exhausted to the surroundings through the center chamber. The rate at which gas was vented to the surroundings (and, therefore, the rate at which it was withdrawn at the capture slot) was controlled by two precision Hoke needle valves that were mounted on the center chamber.

í

The second second the second second second

المالية المراجعة المراجعة المراجع والمحالية والمحموظ المحالية المحالية المالية المحالية

1

Figure B.10 shows a photograph of the separator and the support stand. Also shown in the figure is the graduated beaker employed to collect the liquid sample from the separator and the electric timer that measured the sample time. The on-off control for the capture slot (an electric solenoid with a 500 psia air supply) is shown located under the test section. The aforementioned timer and the on-off control were simultancously regulated by one of the switches that is shown on the front panel of the support stand for the separator. The procedure for operating the cyclone separator was as follows. First the high-pressure air solenoid

> 1 27 a Tamar sa



Fig. B.9. Sectional Drawing of Separator

.

267

•

Constance: 4

ĺ

ſ

ſ

[

ſ

[

[

ľ

l

S



Fig. B.10. Photograph of Separator

was activated so that no gas or liquid entered the capture slot. Then the bottom drain valve on the separator was opened so that any liquid that remained in the separator was removed. The drain valve was closed. Then the timer was started and the high-pressure air jet was shut off simultaneously, thereby beginning a mass transfer measurement. After a certain length of time, normally one minute, the switch was thrown to both stop the electric timer and to reactivate the electric solenoid for the high-pressure air jet. The drain valve on the bottom of the separator was then opened and the liquid in the lower chamber of the separator was collected in the graduated beaker. The size of the liquid sample that was collected ranged from approximately 100 to 800 ml/min, depending on the rate at which the liquid was injected onto the test plate. The temperature of the liquid in the beaker was frequently measured with a mercury thermometer so as to determine the specific gravity of the liquid at the time of collection. Knowing the specific gravity for the liquid and the volume of the liquid that was collected, the average rate of liquid flow out of the test section during the sample time could be determined.
## APPENDIX C

#### EXPERIMENTAL PROCEDURE

#### C.1 Pre-Run Preparation

The pre-run preparation of the experimental apparatus encompassed three major items, namely, (1) the preparation of the gas generator and instrumentation systems, (2) the calibration of the instruments, and (3) the establishment of the start-up configuration for the entire apparatus prior to the start of an experimental run.

The air storage tanks having a total volume of 2500 cu. ft. were charged with air to a pressure between 1500 and 2400 psi which was sufficient for operation of the gas generator for a period of from 2000 to 4000 seconds. The fuel tanks were filled with methanol. Also the film coolant tank was filled. Several hours prior to an experiment the instrumentation was turned on to allow for warm-up and stabilization. The camera, if employed, was loaded with film and mounted on the camera support. The strobe unit was checked to ensure proper functioning and was patched into the synchronization system of the camera. The combined system was then checked again to ensure correct functioning.

The pressure transducers monitoring the operation of the gas generator were calibrated in 5 psi increments around their normally recorded values employing a precision Bourden tube pressure gage as a standard. The precision gage was periodically calibrated employing an Ashcroft dead weight tester. The calibration of the turbine flowmeters monitoring the air, fuel, and liquid flow rates was accomplished by means of a 100 cps reference signal available at the Potter flowmeter electronic tachometers. The tachometers supplied a d.c. analog output proportional to the rotative speed of the flowmeter elements. All of the aforementioned pre-run calibrations were recorded on the same stripchart recorders which recorded the parameters during an experiment. The Pottermeter that measured the rate of liquid flow onto the test plate was normally calibrated before each run by the time-weight technique.

The establishment of the start-up configuration for the entire apparatus was performed just prior to ignition. The stripchart recorders, which were all set to run at 4 inches per minute, were synchronized. The propellant air and fuel and liquid tank pressures were set to the desired values. The propellant air and fuel were then bled up to the main control valves. The flow rates desired at ignition were established by opening one valve at a time and setting the desired flow rate of either fuel, primary air or secondary air. Five seconds prior to ignition all of the stripchart

recorders were started and ran for the duration of the experiment.

The start-up of the gas generator was accomplished with the aforementioned automatic start sequencer.

#### C.2 In-Run Procedure

After mainstage ignition was secured, the operator brought the gas generator up to the nominal operating point. A period of time from five to ten minutes was required for the apparatus to reach thermal equilibrium after the operating point had been reached. During the warm-up period the film-coolant flow rate was started and the heat exchanger for preheating the liquid was turned on. The film of liquid on the test plate was visually inspected to ensure proper coverage. After thermal equilibrium was established and the desired operating point was set, the experimental measurements were begun. The mass transfer measurements followed the procedure outlined in Section 2.5 and Appendix **B.4**. The following tasks were performed during each mass transfer measurement: (a) three photographs of the liquid film were taken (if the camera was employed during the test), (b) the static pressure in the approach section to the test section (that was displayed on the precision Bourdon tube gage in the control room) was recorded, (c) the readings on the 5 mercury manometers for the static pressure taps in the aluminum wedge were recorded (if applicable), and (d) the inlet and outlet temperatures for

the liquid, that were displayed on their respective stripchart recorders, were noted and the reostat for controlling the inlet temperature was adjusted accordingly.

## C.3 Post-Run Procedure

Upon the completion of an experimental test the gas generator was shut down. An afterbleed of propellant air was allowed to flow through the gas generator and test section to reduce the temperature of the equipment. The electric heat exchanger for preheating the film coolant was turned off. The liquid continued to flow onto the test plate during the cool-down period.

## APPENDIX D

## PHYSICAL PROPERTIES OF FILM COOLANTS

# D.1 Subject Investigation

Figures D.1-D.4 present the surface tension  $\sigma$ , the liquid viscosity  $\mu_1$ , the heat of vaporization  $H_v$ , and the vapor pressure  $p_v$  as a function of the liquid temperature  $T_1$ . A separate plot is given for each of the film coolants of the subject investigation, namely, methanol, butanol, water, and RP-1. Because the film coolant RP-1 is a mixture of complex hydrocarbons, the properties for this coolant can be specified only approximately. The property values that are given herein for RP-1 assume, where necessary, that none of the volatiles have been removed. The circle that is presented for each curve in each of the figures represents the maximum withdrawal liquid temperature that was measured for that coolant during the course of the investigation. Each of the curves that is presented in Figs. D.1-D.4 was obtained from one or more of the references (56-60).

The molecular weights of the subject film coolants (employed in the calculation of the maximum liquid temperature  $T_{1,m}$ -see Subsection 3.4.1) are as follows: methanol : 32; butanol : 74; water : 18; RP-1 : 172.



•

· · · · 4

j.

U

Fig. D.1. Surface Tension as a Function of Liquid Temperature for Liquids of Present Investigation



ţ

1

1444

-

The second second

T

]

Fig. D.2. Liquid Viscosity as a Function of Liquid Temperature for Liquids of Present Investigation



T<sub>1</sub> (F)

Fig. D.3. Heat of Vaporization as a Function of Liquid Temperature for Liquids of Present Investigation

これを記載がないでき、たいないないないないないないないない。

ž

ſ

**f** 

ſ



Í

]

{

Ì

]

- Witness

I

Fig. D.4. Vapor Pressure as a Function of Liquid Temperature for Liquids of Present Investigation

### D.2 Investigation Due to Emmons

Figures D.5-D.7 present the surface tension  $\sigma$ , the heat of vaporization H<sub>v</sub>, and the vapor pressure p<sub>v</sub> as a function of the liquid temperature T<sub>1</sub> for each of the film coolants that was employed by Emmons (20), namely, ethanol, water, ammonia, and Freon-113. The values for the calculated maximum liquid temperature, T<sub>1,m</sub>, and for the liquid reference temperature, T<sub>1,r</sub>, are indicated for the coolants in each of the figures. Figure D.5 shows considerable extrapolation of the data for the surface tension  $\sigma$ . The extrapolation of the data for  $\sigma$  was guided by the critical temperatures for the coolants; i.e., at the critical temperature the surface tension for the liquid is zero. The values for the critical temperature for the various coolants is given in Fig. D.5. Each of the solid curves that is presented in Figs. D.5-D.7 was obtained from one or more of the references (56, 58-61).



Fig. D.5. Surface Tension as a Function of Liquid Temperature for the Liquids Employed by Emmons

ų,

•



ſ

ſ

Fig. D.6. Heat of Vaporization as a Function of Liquid Temperature for the Liquids Employed by Emmons

•



Ę

Fig. D.7. Vapor Pressure as a Function of Liquid Temperature for the Liquids Employed by Emmons

.

.

٠.

**2**82

#### APPENDIX E

A SAMPLE CALCULATION FOR THE MAXIMUM LIQUID TEMPERATURE

To illustrate the manner in which the calculated results presented in Fig. 34 and in Fig. 35 were obtained, a sample calculation for the maximum liquid temperature T<sub>1,m</sub> for an air-water system (Fig. 34) is presented below.

For the flow conditions represented by Fig. 34, Eq. (3-76b) reduces to the form

I

E

G

Į.

ļ

$$\frac{H_{v,m}}{c_{p_{air}}(T_{g}-T_{1,m})} = \frac{M_{g}}{M_{v}} \frac{p - p_{v,s}}{p_{v,s}}$$
$$= 1.61 \frac{p - p_{v,s}}{p_{v,s}}$$
(E-1)

Solving Eq. (E-1) for the partial pressure of the vapor at the interface, one obtains

$$p_{v,s} = \frac{p}{\frac{1}{1.61} \frac{H_{v,m}}{c_{p_{air}} (T_g - T_{1,m})} + 1}}$$
 (E-2)

For purposes of illustration, assume that  $T_g = 2000 R (c_{p_{air}} = 0.2772 Btu/lb-R)$ 

p = 500 psia

For that case, Eq. (E-2) reduces to

$$P_{v,s} = \frac{500}{2.24 H_{v,m}} psia$$
 (E-4)  
 $\frac{1}{(2000 - T_{1,m})} + 1$ 

1

]

THE REAL

To initiate the trial-and-error calculations, assume that  $T_{l,m} = 400F = 860R$  (the accuracy of the first approximation for  $T_{l,m}$  is not critical). For  $T_{l,m} = 960R$ , the data in Ref. 41 gives  $H_{v,m} = 826.0$  Btu/lb and  $p_{v,s} = 247.3$  psia. Substituting those data into Eq. (E-3) one obtains

247.3 psia  $\frac{2}{1.624} + 1$  psia

≠ 190.5 psia

For the second càlculation, assume that  $p_{v,s} = 190.5$  psia. Corresponding to  $p_{v,s} = 190.5$  psia, the data in Ref. 41 gives  $H_{v,m} = 846.7$  Btu/lb and  $T_{l,m} = 837.6R$ . Substituting those data into Eq. (E-3) one obtains

190.5 psia 
$$\frac{2}{1.633 + 1}$$
 psia

🗲 189.9 psia

Continuing this procedure, we assume for the next calculation that  $p_{v,s} = 189.9$  psia. Corresponding to  $p_{v,s} = 189.9$ psia, the data in Ref. 41 gives  $H_{v,m} = 846.9$  Btu/lb and  $T_{1,m} = 377.4F = 837.4R$ . Substituting those data into Eq. (E-3), one obtains

189.9 
$$\frac{?}{1.633 + 1}$$
 psia

= 189.9 psia

F

Thus, for  $T_{l,m}$  = 2000R and p = 500 psia, and for the flow conditions represented by Fig. 34, the calculated value of the maximum liquid temperature is  $T_{l,m}$  = 377.4F = 837.4R.

## APPENDIX F

#### DATA REDUCTION

#### F.1 General Data Reduction

The pressure transducers that were employed to measure the static and total pressures in the gas generator system and the test section were calibrated against a precision pressure gage prior to each experiment. The calibration was in the form of discrete pressure steps both ascending and descending; the maximum value of the largest calibration step was in excess of the value recorded during an experiment. The calibration and the corresponding run data were both recorded on the same Brown stripchart recorder. The numerical value of the particular pressure parameter recorded during an experiment was determined by linear interpolation of the recorded transducer output against the calibration. The estimated error resulting from the transducer hysteresis and data transcription was estimated at ± 2 percent at the nominal run value of pressure.

All of the temperatures monitored during an experiment were recorded either on direct reading Brown recorders or on pre-calibrated Brown recorders. The values of temperature recorded during an experiment were correspondingly read off directly from the recorder chart or determined with aid of a calibration curve. The estimated error in the temperatures read from the Brown recorders was from ± 2 to ± 4 percent at the nominal run value of the temperature.

The volume flow rates of air, fuel and film coolant were measured with turbine flowmeters. The flowmeters which measured the flow rates of the fuel and film coolant were calibrated with the respective liquid in terms of frequency output as a function of weight flow rate. In this manner a calibration constant was established for a particular flowmeter through which the weight flow rate and frequency output were related. The flowmeter employed to meter the flow rate of air to the gas generator was calibrated by the company supplying the instrument as volume rate of flow versus frequency at a static pressure of 600 psig. The flowmeter was subsequently calibrated again at the Jet Propulsion Center, Purdue University, against ASME standard orifices at flow pressures of 400, 600, and 800 psig. The latter calibration and the calibration supplied with the meter agreed within ± 5 percent at the nominal run value. The weight flow rate of air measured during a run was evaluated as

(F-1)

Qρ<sub>a</sub>

1.5

where

- Q = indicated volume flow rate at the frequency output of the flowmeter
- ρ<sub>a</sub> = density of the air determined from the measured static pressure and temperature of the air immediately upstream of the flowmeter. and employing the equation of state for a perfect gas

The output circuit of the flowmeters was calibrated prior to each experiment against known frequencies available in the flowmeter instrumentation. The estimated error in the measurement of the film coolant flow rates was ± 2 percent or less of the nominal run value. The estimated error in the measurement of the air flow rate was taken as the deviation of the two aforementioned calibrations, that is, ± 5 percent.

#### F.2 Gas Stream Parameters

The procedure that was employed to compute the values for the gas stream parameters G,  $u_g$ , and  $M_o$  at the leading edge of the experimental test plate is outlined as follows: (a) A first estimate for the gas mass velocity, G, was obtained by assuming one-dimensional flow in the experimental tunnel, in which case it follows from the conservation of mass that (neglecting the fuel flow rate into the gas generator)

W = G A<sub>cr</sub>

288

(F-2)

where

W = measured air flow rate

 $A_{cr}$  = cross-sectional area of the experimental tunnel (b) The displacement thickness for the turbulent boundary layer on each side wall of the test section at the leading edge of the wetted test plate was computed from the classical expression (36)

$$\delta_1^* = 0.048 x_1 Re_{x_1}^{-0.2}$$
 (F-3)

where

1

$$\operatorname{Re}_{x_1} = \frac{\operatorname{G}_{x_1}}{\mu_g}$$

μg = viscosity for the gas stream evaluated at the measured gas stream temperature

 $x_1 = 40$  inches

(c) A corrected cross-sectional area for the flow at  $x = x_1$ , denoted by  $A'_{cr}$ , was computed from

$$A'_{cr} = A_{cr} - \delta'_{1}$$
 (wetted perimeter of tunnel) (F-4)

(d) The final value for the gas mass velocity was computed by multiplying the result of step (a) by the ratio of the uncorrected to the corrected cross-sectional area for the tunnel at  $x = x_1$ .

(e) The gas stream velocity,  $u_g$ , was computed from the result of step (d) for the mass velocity G together with

the expression

$$u_{g} = \frac{G}{\rho_{g}}$$
 (F-5)

where the gas stream density  $\rho_{\rm g}$  was computed from the equation of state for a perfect gas and the measured values for the gas stream pressure and temperature.

(f) The momentum parameter, M<sub>o</sub>, was computed from

$$M_{o} = \rho_{g} u_{g}^{2} \qquad (F-6)$$

The accuracy of the computed values for the foregoing parameters is estimated as  $\pm$  5 percent for u<sub>g</sub> and G, and  $\pm$  5 to 10 percent for M<sub>o</sub>.

For those tests in which the gas stream was not accelerated over the wetted test plate, the values for the parameters  $u_g$ , G, and  $M_o$  were assumed constant throughout the test section. For those tests where the gas stream was accelerated over the wetted test plate, the local values for the gas stream parameters  $u_g$ , G, and  $M_o$  for  $x > x_1$  were computed from the static pressure measurements that were made down the length of the test section. Those calculations and measurement: were discussed previously in Subsection 3.3.1 and will not be repeated here. The accuracy of the values that were computed for those that were computed for the static as  $\pm 5$  percent for  $u_g$  and G, and

 $\pm$  5 to 10 percent for M<sub>0</sub>.

#### F.3 Mass Transfer Data

The procedure that was employed to measure the net rate of mass transfer from the liquid film to the gas stream has been described in detail in Section 2.5. Moreover, the operation of the separator that was utilized to measure the rate of liquid withdrawal from the wetted test plate has been adequately described in Appendix B.4.

During the course of any one experimental test, a number of consecutive measurements of the rate of liquid withdrawal was normally made for one or more fixed rates of liquid injection in order to check the reproducibility of the data. The measured rates of liquid withdrawal could always be reproduced with an accuracy of 5 percent or better.

A concerted effort was not made to try to reproduce the experimental data of any one experimental test by conducting an independent experimental test at the same flow conditions. It was very difficult to operate the hot gas generator at the same run point for two different experimental tests. The primary problem was that of establishing the same rate of air flow (and, therefore, gas mass velocity or gas stream Reynolds number) on two different occasions. However, a matching of the run conditions was realized for two different sets of experimental tests, and in each case the measurements of mass transfer were found to reproduce to within an accuracy of 10 percent or better.

## APPENDIX G

#### EXPERIMENTAL DATA

Table G.1 lists the measured or computed values for the pertinent flow parameters for the experimental tests with a constant velocity gas stream. The letter that follows the test number indicates the film coolant for that test; i.e., M - methanol; B - butanol; W - water; and R - RP-1. The parameter values listed in the table are the average of the values that were measured for each of the datum points for mass transfer during an experimental test. From 6 to 15 datum points were obtained for each test. Not listed in Table G.1 are those experimental tests that were concerned only with photographic data, those tests which were purely exploratory as regards experimental technique, etc., or those tests for which some failure of the experimental apparatus voided the test.

Table G.2 presents a similar listing of the average values of the pertinent flow parameters for the experimental tests with an accelerating gas stream.

Figures G.1 through G.13 present graphically the mass transfer data for the experimental tests listed in Table G.1. Figure G.14 presents the mass transfer data for the

# experimental tests that are listed in Table G.2.

۱

I

E

I

I

E

Table G.1

Averinge Values of the Pertinent Flow Parameters During the Experimental

Tests With a Constant Velocity Gas Stream

- E S	1b ft-sec	
×		
X <sub>e</sub> ×10 <sup>-4</sup>	٦b <sub>:</sub> <sup>3</sup>	
.T <sub>1,2</sub>	LL.	
.T, , ,	LL_	
× ×	$\frac{1b_r}{ft^2}$	
<b>5</b>	1b ft <sup>2</sup> -sec	
۰ م م	fps	
ف	psia	ts
ţ	L.	ow Tes
TEST.		Hot Fl

37-W	590	78	<b>185</b>	37	210	174	190	3.87	16.4	0.00812
38-W	590	62	239	49	361	173	195	5.07	21.4	0.00987
<b>M-</b> 68	601	80	288	59	523	166	196	6.09	26.0	0.01174
41-W	588	149	75	29	<b>6</b> 6	181	221	2.22	9.2	0.00584
43-W	586	155	202	81	505	163	232	6.09	25.2	0.01300
44-W	594	152	141	55	241	171	233	4.23	17.4	0.00973
45-W	650	78	430	81	1087	154	206	8.87	37.8	0.01682
112-M	424	75	119	27	100	133	129	8.16	11.1	0.01085
113-M	6 j E	73	167	38	199	138	125	11.45	15.5	0.01351
114-M	452	75	247	58	445	134	135	17.24	23.2	0.01843
115-M	352	76	68 <sup>,</sup>	21	59	137	130	6.24	8.4	0.00791
116-M	401	73	196	45	272	144	134	13.59	18.1	0.01512
117-M	404	74	287	66	594	124	138	19.79	26.8	0.02014
<b>M-811</b>	609	75	221	42	286	130	153	14.80	19.4	0.02427
120-M	593	77	108	21	72	129	152	7.36	1.6	0.01340
122-M	597	74	172	33	175	125	144	11.38	15.2	0.01957
125-M	418	150	41	19	24	133	160	4.14	5.4	0.00691

A lower

Table G.1 (Continued)

1

TEST	Тg	٩	6 1	IJ	×°	T1,1	T1,2	X <sub>e</sub> xl0 <sup>-4</sup>	×r	- S E
	<b>ند</b>	psia	fps	1b ft <sup>2</sup> -sec	$\frac{1b}{ft^2}$	LL.	HL.	١Ď <sup>4</sup>	$\frac{1b^3}{f(\cdot)}$	1b ft-sec
126-M	404	150	86	41	109	133	159	8.73	11.4	0.01231
128-M	410	152	150	11	331	134	164	15.32	19.9	0.01965
201-8	402	77	135	32	135	146	190	9.33	12.6	0.01192
202-B	400	74	176	41	224	143	190	11.97	16.2	0.01424
203-8	400	75	52	12	20	131	184	3.50	4.8	0.00526
204-B	398	75	110	26	89	135	184	7.45	10.2	0.00987
205-B	8ú <i>1</i>	76	165	39	201	128	190	11.23	15.4	0.01388
206-B	411	77	233	56	403	131	196	16.04	21.8	0.01825
207-B	400	151	56	26	45	145	210	5.48	7.3	0.00872
210-B	401	151	80	38	94	145	208	7.86	10.4	0.01205
211-8	402	151	106	50	165	159	206	10.54	13.8	0.01593
212-B	599	77	172	34	179	147	215	11.52	15.2	0.02255
213-B	595	75	126	24	94	145	210	8.31	11.0	0.01727
214-8	601	76	251	49	379	149	220	16.86	22.1	0.03054
· 300-R	403	76	153	36	173	169	280	8.46	13.9	0.00851
301-R	383	77	114	28	100	168	270	6.42	10.5	0.00636
302-R	400	77	194	47	285	146	284	10.90	17.9	0.00933
303-R	455	77	231	52	375	166	294	12.62	20.8	<b>U.0148</b> 5
304-R	431	152	89	41	112	172	290	6.84	11.3	0.01073
305-R	425	153	160	75	373	168	300	12.46	20.5	0.01531
306-R	642	78	112	21	75	185	310	5.89	9.6	0.01573
307-R	419	154	53	25	41	184	300	4.12	6.8	0.00615
308-R	554	77	172	35	188	188	338	9.04	14.9	0.01456

TEST	19	٩	<sup>б</sup> л	IJ	° ¥	T1,1	T <sub>1,2</sub>	x <sub>ex10</sub> -4	×r	- S S
	ŧد.	psia	fps	1b ft <sup>2</sup> -sec	$\frac{1b}{ft^2}$	<b>ند</b>	<b>L</b> .	16 <sup>4</sup>	10 <sup>2</sup>	1b ft-sec
Cold F	low Te:	sts								
47-W	20	78	69	30	65	40	37	1.57	8.0	
48-W	10	78	213	63	616	40	32	4.85	24.6	
102-M	42	82	35	16	17	80	40	2.54	4.1	
103-M	38	78	154	66	317	83	34	10.90	17.6	
104-M	40	80	06	39	108	72	37	6.39	10.3	
105-M	40	77	129	53	213	78	34	9.00	14.5	
109-M	26	77	181	77	434	67	22	12.80	20.6	
123-M	14	75	32	13	13	50	28	2.17	с. 5	
124-M	14	, 5 ,	147	63	287	60	24	10.40	16.7	
129-M	9	73	84	36	63	60	32	5.86	9.4	
215-8	10	75	<b>150</b>	65	300	66	16	10.10	17.1	

# Table G.2

Average Values of the Pertinent Flow Parameters During the Experimental Tests With An Accelerating Gas Stream

Test	Tg	р <sup>о</sup>	T <sub>1,1</sub>	T <sub>1,2</sub>	m 's
	F	psia	F	F	<u>lb</u> ft-sec
401-M	442	74	139	142	0.0219
402-M	434	52	140	126	0.0168
403-M	440	103	144	154	0.0267
405-M	447	126	135	173	0.0314
502-W	444	103	118	195	0.0102
503-W	435	54	78	163	0.0062
505-W	405	99	78	188	0.0093
507-W	480	125	71	213	0.0132

and the second second











Í



















- 日日本市の市田市の市内 - 1.

A PARTY STREET, STREET


 $T_g = 400F$ , p = 75 psia. RP-1 Film Coclant

Data for Mass Transfer.

Fig. G.9.





 $T_g = 600F$ , p = 75 psia. RP-1 Film Coolant Data for Mass Transfer. 6.11.

511



.; F





